RESOLUTIONS

(1)

RESOLVED: That upon the recommendation of the RTM Library, Museum and Arts
Committee, in accordance with Section C34-1 of the Town Charter, Jeremy Price and
Andrew Wilk are hereby reappointed to serve as trustees of the Westport Library for a
year term beginning to

(2)

RESOLVED: That upon the request of the Human Services Director, pursuant to CGS 12-630aa et seq., the Westport Country Playhouse and Homes with Hope Inc. are hereby approved as programs eligible for investment by businesses under the provisions of the 2021 Connecticut Neighborhood Assistance Act (NAA) Tax Credit Program.

(3)

RESOLVED: That upon a request of the Finance Director and the Personnel/Human Resources Director, the Retirement Plan for Non-Union Supervisory Employees of the Town of Westport is hereby revised to include current management of the Police and Fire Departments.

5/25/2021

Minutes/Report Library Museum and Arts Committee Meeting 5/25/2021, 4:30 PM
Meeting ID: 880 4035 4222

Committee Members Present:
Amy kaplan, Chair
Dick Lowenstein
Sal Liccione
Harris Falk
Arline Gertzoff

Guest:

Pat Weiser, VP Library Board of Trustees, VP of Governance and Nominating

Once the Committee achieved a quorum of 5 of its 9 members, Chair Amy Kaplan called the meeting to order at 4:32pm. Members had reviewed in advance short bios of Jeremy Price and Andrew Wilk, the two candidates seeking re-appointment to the Library Board of Trustees. Pat Weiser, head of the Library's Gov-Nom Committee, attended as a representative of the Board of Trustees and voiced her full support for both candidates, noting that the Board considers them each valuable and integral members. Jeremy Price is a current VP working closely with Board President Iain Bruce and the Library executive team, and Andrew Wilk has been instrumental in helping to source, connect and produce much of the amazing Library programming that is so well-received by our community. The committee agreed that both candidates are extremely qualified and looks forward to more of their service to the Library and the Town of Westport. Sal Liccione made a motion to recommend that the full RTM approve the re-appointment of both Jeremy Price and Andrew Wilk, each to a new 4-year term as Library Trustees. Arline Gertzoff seconded, and all 5 committee members present voted in favor, 5-0. The meeting was adjourned at 4:41.

Submitted 5/25/2021 by Amy Kaplan, Library Museum and Arts Committee Chair



WESTPORT, CONNECTICUT

JAMES S. MARPE First Selectman

BACK UP MATERIAL RTM ITEM #

TO:

Velma Heller, RTM Moderator

Jeff Dunkerton, Town Clerk

FROM:

James S. Marpe, First Selectman

DATE:

May 17, 2021

RE:

RTM Approval for 2021 Neighborhood Assistance Act Program Proposal

Kindly place the attached request on the upcoming Representative Town Meeting agenda for legislative approval of the applications for the 2021 Neighborhood Assistance Act Program Proposal, as attached.

Thank you.

cc: E. Daignault



Department of Human Services



Town Hall, 110 Myrtle Avenue Westport, CT 06880 Westportct.gov humansrv@westportct.gov Telephone (203) 341-1050

TO:

V James Warre, Flist Selectman

FROM:

Elaine Daignault, DHS Director

DATE:

May 14, 2021

RE:

Revised Items for the June 1, 2021 RTM Meeting

I respectfully request that the following items be placed on the June 1 RTM agenda for legislative approval, per CGS Sec 12-632:

The approval of the Westport Country Playhouse (two applications) and the Homes with Hope, Inc. application for the 2021 Neighborhood Act Tax Credit program.

This NAA program is designed to provide funding for municipal, and tax-exempt organizations, by providing a corporation business tax credit for businesses that make cash contributions to these non-profit organizations.

This program allows businesses to claim a State tax credit for cash contributions made to qualifying community programs conducted by tax exempt or municipal agencies.

Attachments

CC:

Eileen Flug, Assistant Town Attorney Jeff Dunkerton, Town Clerk

Municipality:	Westport	



Form NAA-01

2021 Connecticut Neighborhood Assistance Act (NAA) Program Proposal

This form must be completed and submitted to your municipality for approval. All items must be completed with as much detail as possible. If additional space is needed, attach additional sheets. Please type or print clearly. See attached instructions before completing. Do not submit this form directly to the Department of Revenue Services.

Part I — General Information				
Name of tax exempt organization/municipal agency:				
Westport Country Playhouse				
Address: 25 Powers Court, Westport, CT 06880				
Federal Employer Identification Number: 23-7357943				
Program title: Upgrade parking lot lights to LED lighting				
Name of contact person: Michele Crowley				
Telephone number: 203-571-1284				
Email address: mcrowley@westportplayhouse.org				
Total NAA funding requested (\$250 minimum, \$150,000 maximum): \$ 14,210.17				
Is your organization required to file federal Form 990 or 990EZ, Return of Organization Exempt from Income Tax?				
X Yes No				
If Yes, attach a copy of the first page of your most recent return.				
If No , attach a copy of your determination letter from the U.S. Treasury Department, Internal Revenue Service.				

Part (I — Program Information

Check the appropriate description of your program:				
100% credit percentage				
X Energy conservation; or				
Comprehensive college access loan forgiveness (see Conn. Gen. Stat. § 12-635(3)).				
50% credit percentage Job training/education for unemployed persons aged 50 or over; Job training/education for persons with physical disabilities; Program serving low-income persons; Child care services; Establishment of a child day care facility; Open space acquisition fund; or Other (specify):				
Description of program:				
The Westport Country Playhouse is looking to upgrade our parking lot lights to energy saving LED lighting. The retro fit will consist of 32 parking lot lights and 13 driveway lights.				
Need for program:				
.The Playhouse believes it is a good time to invest in energy-efficient lighting for the parking lot since the current lights are old and not energy-efficient.				
We are incorporating "green" technology into the Playhouse where possible, and ultimately the energy-efficient lighting will save our organization money while benefitting the natural environment.				
<u> </u>				
Neighborhood area to be served:				
·Town of Westport				
•				
Plan to implement the program:				
-Meet with a company to find out price of new lighting for the parking lot.				
Develop a time frame to replace the lighting that works with the Playhouse's season/schedule of shows.				
'Contract with a company to replace the old lights with new, energy-efficient lighting.				

Program start date: 9/1/2021					
Program completion date: 9/1/2023					
The program completion date must not be more than two years from post-project review is due to the municipality overseeing implementation date for all projects receiving \$25,000 or	tation no later than three months				
Part III — Financial Information					
Program Budget:					
Complete in full. Expenditures must equal or exceed total funding.					
Sources of Revenue:					
NAA funds requested	14210.17				
Other funding sources - itemized sources:	a a				
a)					
b)					
c)					
d)					
Total Funding:					
Proposed Program Expenditures:					
Direct operating expenses - itemized description:					
a) Lighting fixtures	4580.17				
b)					
c)					
d)					
Administrative expenses - itemized description:					
a) Facilities Manager @ 15%	9630				
b)					
c)					
d)					
Total Proposed Expenditures:	14,210.17				

Timetable:

Form NAA-01 (Rev. 02/21) Page 3 of 5

Part IV — Municipal Information

To be completed by the municipal agency overseeing implementation of the program

Name of municipal agency overseeing implementation of the program: Town of Westport Department of Human Services			
Maifing address:			
110 Myrtle Avenue, Room 200, Westport, CT 06880			
Name of municipal liaison:	Ms. Elaine Daignault, Director		
Telephone number:	203-341-1050		
Fax number:	203-341-1073		
Email address:	elained@westportct.gov		

Post-Proje	ect Review
Is a post-project review re	equired for this proposal?
Yes	(x)No
If Yes, date post-p	project review due:
Da	ite

Form NAA-01 (Rev. 02/21) Page 4 of 5

Form **990**

Department of the Treasury

Return of Organization Exempt From Income Tax

OMB No. 1545-0047

Under section 501(c), 527, or 4947(a)(1) of the Internal Revenue Code (except private foundations) Do not enter social security numbers on this form as it may be made public. > Go to www.irs.gov/Form990 for instructions and the latest information.

		GD OCIAION			<u> </u>		W 1 744			
<u>A.</u>							, 20 (9			
8	Check if	ck if applicable: C Name of organization WESTPORT COUNTRY PLAYHOUSE INC							D Emplo	yer identification number.
	Address	s change. Doing business as							23-7357943	
	Name ci	change Number and street (or P.O. box if mail is not delivered to street address) Room/sufte						E Teleph	one number	
	Initial ret	บก								203-227-5137
	Final retu	um/terminated City or town, state or province, country, and ZIP or foreign postal code								
$\overline{\Box}$	Amende	diretum	Westport, CT,						G Gross	receipts \$ 5,612,429
		ion pending,	F Name and add	ress of principal of	ficer: Michael Bark	er		H(a) is this a	group tetum fo	r subordinates? Yes 🗹 No
_	Market and	Tell (France)	- **	ourt, Westport, C	W			H(b) Are all	subordinati	s included? 🗌 Yes 🔲 No
ـــــــــــــــــــــــــــــــــــــ	Тах-ехе	mpt status:	√ 501(c)(3)	501(c) () < (insert no.);	4947(a)(1) or	527			e instructions)
J	Website	e Di www.w	estportplayhou	se om				H(c) Group	exemption	number >
				Trust Associa	ation Other >	L Yea	r of formatio			of legal domicile: CT
·	art I	Summa		Emmile .					.:1	
	1			nization's miss	sion or most signi	ficant activities:	To transf	orm people	's lives th	rough the power of
σ.	'									lents by presenting
Ę	1			D, Statement 2)		ac community of	HERMIN	4013, 011130	3 0110 3000	nation man by Court in 182
Ĕ	2				discontinued its	operations of d	iennead o	f more that	n 25% of	its net accets
š	3				eming body (Part					
۳					ers of the governir				4	32 32
Š	4				in calendar year 2				5	146
Ě	5			ais employed ers (estimate if		* * e ² :	zaj .		6	
Activities & Governance	6						* * *		7a	250
4	7a				Part VIII, column			, ., .	7b	5,046
<u> </u>	<u> </u>	Net unrela	ted business.	axable income	from Form 990-	i. me as	* * *	Prior Y		5,046 Curient Year
	ļ		and the second second	e <i>18</i> 51a. vezasa (1811	. See LCU:				ear 2:447;340	
9	8									2,909,914
6	9		service revenue (Part VIII, line 2g)						2,128,704	2,094,180
Неувлие	10		nt income (Part VIII, column (A), lines 3, 4, and 7d)						23,754 89,436	90,102
_	11		enue (Part VIII, column (A), lines 5, 6d, 8c, 9c, 10c, and 11e)							123,801
_	.12								4,689,234	5,217,997
	13		nd similar amounts paid (Part IX, column (A), lines 1–3)							
	14		paid to or for members (Part IX, column (A), line 4)							. 0
(i)	15		es, other compensation, employee benefits (Part IX, column (A), lines 5-10)							3,213,741
Expenses	16a		stonal fundralsing fees (Part IX, column (A), line 11e)							0
X	b				lumn (D), line 25)		2,309	* *** *** *** ***		
Щ	-17.				nes-11a-11d,-11f				2,469,102	
	18				t equal Part IX, co	olumn (A), line 2!	5)	<u> </u>	5,553,129	5,825,825
_	19	Revenue le	ess expenses.	Subtract line	18 from line 12	· · · · · ·	<u> </u>	3 100 W	4863,895	
at Assets or			<u> </u>	***			B	oginning of C		1
Set	20		ts (Part X, line		- 4 F F A A .		• • •		5,274,717	
2.7	21		ities (Part X, li						2,282,937	-
7 1	22			rces. Subtract	line 21 from line	20			2,991,780	12,481,131
	art.ll_		ıre Block	<u></u>			. <u></u>			· · · · · · · · · · · · · · · · · · ·
Ų	der pen	alties of perjury	, I declare that I I	nave examined this	return, including acc	ompanying schedule	s and staten	nents, and to	the best of	my knowledge and belief, it is
tn	ia, correc	it, and complet	le. Declaration of	busbarer forber me	n officer) is based on	as information of wi	ich preparer	nas any knov	viecge.	
4.5		1								and the second s
Si		Signat	ture of officer					3,	ate	
He	ere:			naging Directo	r					<u> </u>
_		2	or print name and	·						<u></u>
P:	iid	Print/Typ	e preparer's name)	Preparer's signatu	rė.	Da	to		☐ if: PTIN
	epare						<u> </u>		self-en	nployed
	_		miè 🕨					FI	m's EIN 🕨	
U:	se On	У	Medica No.					ы	one no	

May the IRS discuss this return with the preparer shown above? (see instructions)

Municipality: Westport



Form NAA-01

2021 Connecticut Neighborhood Assistance Act (NAA) Program Proposal

This form must be completed and submitted to your municipality for approval. All items must be completed with as much detail as possible. If additional space is needed, attach additional sheets. Please type or print clearly. See attached instructions before completing. Do not submit this form directly to the Department of Revenue Services.

Department of Revenue Services.				
Part I — General Information				
Name of tax exempt organization/municipal agency:				
Westport Country Playhouse				
Address: 25 Powers Court, Westport, CT 06880				
Federal Employer Identification Number: 23-7357943				
Program title: Purchase Assistive Listening System for the Jason Robards Theatre				
Name of contact person: Michele Crowley				
Telephone number: 203-571-1284				
Email address: mcrowley@westportplayhouse.org				
Total NAA funding requested (\$250 minimum, \$150,000 maximum): \$ 10,500				
Is your organization required to file federal Form 990 or 990EZ, Return of Organization Exempt from Income Tax?				
X Yes No				
If Yes, attach a copy of the first page of your most recent return.				
If No , attach a copy of your determination letter from the U.S. Treasury Department, Internal Revenue Service.				

Part II — Program Information

Check the appropriate description of your program:
100% credit percentage
Energy conservation; or
Comprehensive college access loan forgiveness (see Conn. Gen. Stat. § 12-635(3)).
60% credit percentage
Job training/education for unemployed persons aged 50 or over;
Job training/education for persons with physical disabilities;
Program serving low-income persons;
Child care services;
Establishment of a child day care facility;
Open space acquisition fund; or
Other (specify): Provide assistive listening devices for audience members who do not hear well
Description of program:
.To be compliance with the Americans with Disabilities Act, the Playhouse needs to upgrade its 15+ year-old assistive listening system to a more user-friendly/consistent model that allows patrons with hearing loss to better access the productions on stage. For our size building, we need to purchase a 29 receiver/headphone package with 11 "loops" that work with a patron's hearing aids. This Transmitter system includes Wave Cast - a free Download that lets patrons use their iPhones to listen to the show via their own headphones or Bluetooth hearing aids.
Need for program:
.The Playhouse has many patrons who truly need the assistive listening devices to hear our shows on stage. The Playhouse must also upgrade its system in accordance with the Americans with Disabilities Act.
Neighborhood area to be served:
·Town of Westport
Plan to implement the program:
-The Playhouse General Manager looked at various assistive listening devices and chose this one: https://williamsav.com/product/wf-sy5-pro-24/
Once purchased, the assistive listening system will be installed in the Jason Robards Theatre by Playhouse staff.
Playhouse patrons will be able to use the assistive listening devices to better hear the productions on stage.
Form NAA-01 (Rev. 02/21) Page 2 of 5

Program start date: 7/1/2021	
Program completion date: 7/1/2023	
The program completion date must not be more than two years post-project review is due to the municipality overseeing implanter program completion date for all projects receiving \$25,0	ementation no later than three months
Part III — Financial Information	
Program Budget:	
Complete in full. Expenditures must equal or exceed total funding.	
Sources of Revenue:	
NAA funds requested	10,500
Other funding sources - itemized sources:	
a)	
b)	
c)	
d)	
Total Funding:	10,500
Proposed Program Expenditures:	
Direct operating expenses - itemized description:	•
a) Assistive listening device system	10,500
b)	
c)	
d)	
Administrative expenses - itemized description:	
a)	
b)	
c)	
d)	
Total Proposed Expenditures:	10,500

Timetable:

Form NAA-01 (Rev. 02/21) Page 3 of 5

Part IV — Municipal Information

To be completed by the municipal agency overseeing implementation of the program

Name of municipal agency overseeing implementation of the program: Town of Westport Department of Human Services			
Mailing address:			
110 Myrtle Avenu	ue, Room 200, Westport, CT 06880		
Name of municipal liaison:	Ms. Elaine Daignault, Director		
Telephone number:	203-341-1050		
Fax number:	203-341-1073		
Email address:	elained@westportct.gov		

Post-Project Review								
Is a post-project review required for this proposal?								
Yes XNo								
If Yes , date post-project review due:								
Date								

Form NAA-01 (Rev. 02/21) Page 4 of 5

Return of Organization Exempt From Income Tax

Under section 501(c), 527, or 4947(a)(1) of the Internal Revenue Code (except private foundations)

OMB No. 1545-0047

Department of the Treasury Internal Revenue Service

A For the 2019 calendar year, or tax year beginning

Do not enter social security numbers on this form as it may be made public. ► Go to www.irs.gov/Form990 for instructions and the latest information.

Open to Public Inspection

A	For the	2019 calend	dar year, or tax year beginning	01/01	, 2019, and end	ing	12/31		20 19		
В	Check if	applicable:	C Name of organization. WESTPOR		Emplo:	yer Identification number					
	Address	change	Doing business as	1		23-7357943					
	Name ch	Number and street (or P.O. box if mail is not delivered to street address) Ro						Telepho	one number		
	Initial ret	រយា	25 Powers Court		_			203-227-5137			
	Final retu	inviterminated	City or town, state or province, cou	intry, and ZIP or foreign p	ostal code		ï				
	Amende	d return	Westport, CT, 06880	· · · · · · · · · · · · · · · · · · ·	<u>.</u>			Gross	receipts \$ 5,612,429		
	Applicati	on pending	F Name and address of principal offic	er: Michael Barker			H(a) is this a group	return for	subordinates? Yes Vo		
_			25 Powers Court, Westport, CT				H(b) Are all sub	ordinate	s Included? Yes No		
i.	Tax-exer	npt status:	√ 501(c)(3) 501(c) (4947(a)(1) or 🔲 527	Q T	If "No," attach	a list. (se	e instructions)		
J	Website	· Mwwww	estportplayhouse.org				H(c) Group exe	mptlon i	number >		
K	Form of a	organization: 🗸	Gorporation Trust Associati	on Other >	L Year of for	mation	: 1973	A State	of legal domicite: CT		
	artel"	Summai			- <u> </u>			***			
ext.	1		cribe the organization's mission	on or most significar	t activities: To in	ansio	rm people's i	ives thi	rough the power of		
Ö	1 -		ir eim is to enlighten, enrich an								
E E			on Schedule O, Statement 2)		***************************************						
E	2	Check this	box ▶ ☐ if the organization of	discontinued its one	rations or dispose	ed of	more than 2	5% of	its net assets.		
ò	3		voting members of the gover					3	32		
3	4		independent voting members			lb)		4	32		
5	5		per of individuals employed in			· er# ·		5	. 146		
Activities & Governance	6		per of volunteers (estimate if n		क्षेत्रहरू । स्टब्स् १४ - इ.स. १८ - इ.स. १८	15 K)		6	250-		
Ş	7a		ated business revenue from P		•	46		7a	5,046		
_	ь		ted business taxable income t					7b	5,046		
`		1 10 Catholes	tod additiona resident a least to		<u> </u>	1	Prior Year	'-	Current Year		
	8	Contributio	one and grante (Part VIII line 1	<u> </u>		7,340	2,909,914				
<u> </u>	9	Contributions and grants (Part VIII, line 1h). Program service revenue (Part VIII, line 2g)							2,094,180		
Rovenue	10	Investment	-		8,704 3,754	90,102					
æ	2000		-		9,436	123,801					
	12	the first the many through a control of the properties of the part of the properties of the part of th						39,234	,		
			l similar amounts paid (Part I)				4;00	0	<u>5,217,997</u>		
	14		aid to or for members (Part IX			-		0	<u> </u>		
12/2	15		her compensation, employee b	6.77		⋰├─	3,064,027		3,213,741		
863	16a		al fundraising fees (Part IX, co			'	3,00	0	3,213,741		
Expenses	I .		raising expenses (Part IX, colt					7.4.40	0		
ă	.17		enses (Part IX, column (A), line				2,4		0.640.004		
E-mi	18		nses: Add lines 13–17 (must o	-		53,129	2,612,084				
	19		ess expenses. Subtract line 1		53.895	5,825,825					
·		neveriue it	ess expenses, outliact line it	7 HOURING 12	· · · · · · ·		ginning of Cune		-607,828 End of Year		
200	20 21 22	Total aces	ts (Part X, line 16)			1 20	2	74,717	14,652,741		
20.2	21		ities (Part X, line 26)		an an jan ja sa a	' ├─		82.937	2,171,610		
2.5	22	. e	or fund balances. Subtract II	ne 24 from line 20		<u> </u>	200	91.780			
TO.	aa u		ire Block	NO ET MONTHING LO	<u>قى مى ما مات بە يۇ .</u>	<u> </u>	1.613	31,700:	12,701(101		
			, I declare that I have examined this re	ation including accompa	wind ochadulae and s	tatome	inte nod to the	boot of	me knowledge, and hellet little		
tru	io, correc	t, and complet	ie. Declaration of preparer (other than	officer) is based on all info	rmation of which pre	parer h	as any knowled	ge.	my raid alough mid perior is is		
		1 A	<u> </u>				1	<u>·</u>	-		
Sig	án	Signat	ure of officer				Date	-			
He		1.5	ael Barker, Managing Director	•							
4.10	,,,	Type	or print name and title		*						
_	1 1	117 12	preparer's name	Preparer's signature	. "	Date	, 	30 8.1	C (FIDN		
Pa	id		ساهم ساهم شوي شواق شفا و شماه مديم	The Contraction of the Contraction				Check self-em	<u> </u>		
Pr	epare	r	2 54	-		ــــــــــــــــــــــــــــــــــــــ			5 7 Tall.		
	se On	V Fim's nar			<u> </u>	<u> </u>	- 1,	EIN 🟲	**************************************		
	•	Firm's act		ulita hutitiavailt			Phone				
MIZ	y the II	a discuss	this return with the preparer s	nown above? (see i	nsaucauns)	-, .• 		<u>#. 4 (</u>	Yes No		

Municipality: Westport



Form NAA-01

2021 Connecticut Neighborhood Assistance Act (NAA) Program Proposal

This form **must** be completed and submitted to your municipality for approval. All items **must** be completed with as much detail as possible. If additional space is needed, attach additional sheets. Please type or print clearly. See attached instructions before completing. **Do not submit this form directly to the Department of Revenue Services**.

Part I — General Information
Name of tax exempt organization/municipal agency:
Homes with Hope, Inc.
Address: Mailing: PO Box 631, Westport, CT 06881; Physical: 59 Myrtle Avenue, Westport, CT 06880
Federal Employer Identification Number: 22-2534326
Program title: We've Got You Covered": Roof Replacement for Bacharach Community
Name of contact person: Janet Zamparo
Telephone number: (203) 226-3426
Email address: jzamparo@hwhct.org
Total NAA funding requested (\$250 minimum, \$150,000 maximum): \$ 24,340.00
Is your organization required to file federal Form 990 or 990EZ, Return of Organization Exempt from Income Tax?
∑ Yes □ No
If Yes , attach a copy of the first page of your most recent return. If No , attach a copy of your determination letter from the U.S. Treasury Department, Internal Revenue Service.

Part II — Program Information

Check the appropriate description of your program:

100% cre	dit percentage							
	Energy conservation; or							
	Comprehensive college access loan forgiveness (see Conn. Gen. Stat. § 12-635(3)).							
60% cred	it percentage							
	Job training/education for unemployed persons aged 50 or over;							
	Job training/education for persons with physical disabilities;							
Program serving low-income persons;								
	Child care services;							
	Establishment of a child day care facility;							
	Open space acquisition fund; or							
	Other (specify):							
Descriptio	on of program:							
Communit and remove underlaym	osed project consists of roof replacement for three free-standing homes that comprise Bacharach by Permanent Supportive Housing, #1, #3 and #5 Wassell Lane, and will include the following: Strip by existing roof; Install ice and water prevention membrane around perimeter; Install synthetic nent; Install new metal edge; Repair chimney flashing; Install new vent pipe flashing; Install new ridge all 30-year GAF timberline architectural shingles; and Install new gutters and down spouts —4 on each							
Need for p	program:							
help addre inexpensivex expenses,	ndividual homes that comprise Bacharach Community are in great need of roof replacements that will less issues around energy conservation and efficiency: These particular homes were built wely in 1957 as military housing on a slab thus no basement and no insulation in the attics. Heating incurred by Homes with Hope, can be very high. Sometimes, ice dams form along the eaves of the use heat that collects in the attic, warming the roof, causes snow to melt and trickle down to the cold if freeze.							
Neighbort	nood area to be served:							
•	arach Community is located on Wassell Lane in Westport, CT, on a residential street with an							
appearance an emerge Permanen	that is in keeping with the other houses that line this street. The Bacharach Community, formerly sency shelter program for homeless mothers and their children was transitioned to affordable at Supportive Housing for families in 2019. Residents live independently with a lease and pay 30% of the in rent. Case Managers work with tenants to help them achieve and sustain housing stability.							
Plan to im	plement the program:							
Contracting implement	ney, Homes with Hope Vice President and COO will work in conjunction with Peter Jennings, ig Property Manager and President of Bayberry Property Management, to oversee the project's tation and completion. For all three homes, the on-site roof replacement can take anywhere from one few. Billing and internal processing can take anywhere from one month to a few.							

Form NAA-01 (Rev. 02/21) Page 2 of 5

Program start date: April 15, 2022	
Program completion date: June 30, 2022	
The program completion date must not be more than two yost-project review is due to the municipality overseeing after program completion date for all projects receiving	implementation no later than three months
Part III — Financial Information	
Program Budget:	
Complete in full. Expenditures must equal or exceed total fund	ling.
Sources of Revenue:	
NAA funds requested	\$24,340.00
Other funding sources - itemized sources:	
a) First County Bank Foundation - pending	\$8,050.00
b)	
c)	<u> </u>
d)	
Total Funding:	\$32,390.00
Proposed Program Expenditures:	·
Direct operating expenses - itemized description:	
a) Roof Replacement 1	\$9,815.00
b) Roof Replacement 2	\$9,815.00
c) Roof Replacement 3	\$9,815.00
d)	
Administrative expenses - itemized description:	
a) Oversight and Processing	\$2,945.00
b)	
c)	
d)	
Total Proposed Expenditures:	\$32,390.00

Timetable:

Total Proposed Expenditures:

Form NAA-01 (Rev. 02/21) Page 3 of 5

Part IV — Municipal Information

To be completed by the municipal agency overseeing implementation of the program

Name of municipal agency overseeing implementation of the program: Town of Westport Department of Human Services								
Mailing address:								
110 Myrtle Avenue, Room 200, Westport, CT 06880								
Name of municipal liaison:	Ms. Elaine Daignault, Director	.						
Telephone number:	203-341-1050							
Fax number:								
Email address:	elained@westportct.gov							

Post-Proje	ect Review
ls a post-project review re	equired for this proposal?
☐ Yes	(x)No
lf Yes , date post-ք	project review due:
Da	ate

EXTENDED TO NOVEMBER 16, 2020

Return of Organization Exempt From Income Tax

(Rev. January 2020)

Department of the Treasury Internal Revenue Service

Under section 501(c), 527, or 4947(a)(1) of the Internal Revenue Code (except private foundations)

Do not enter social security numbers on this form as it may be made public.

► Go to www.irs.gov/Form990 for instructions and the latest information.

OMB No. 1545-0047 Open to Public Inspection

<u>A</u>	For the	2019 calendar year, or tax year beginning	and	ending				
В	Check if applicabl	C Name of organization			D Emp	loyer ic	lentific	cation number
	Addre chang Name	HOMES WITH HOPE, INC.			_	=		
L	chang				2:	<u>2-25</u>	<u>343</u> 2	26
L	Initial return	Number and street (or P.O. box if mail is not del	•	Room/suite		-		
	Final return		, SUITE 212		(:	203)	<u> 226-</u>	-3426
	termin ated	City or town, state or province, country, and a	ZIP or foreign postal code		G Gross	receipts \$		3,428,193.
L	Amene	WESTPORT, CT 00000			_ H(a) is 1	this a gr	oup re	turn
	Appile tion pendir	.a.			for	subord	inates'	? Yes 🗓 No
		149 RICHMONDVILLE AVE, WI	STPORT, CT 068	180	H(b) Are	all subord	inates in	cluded? Yes No
			(insert no.) 4947(a)(1)	or 527	7 <u> </u>	'No," att	ach a	list. (see instructions)
_		e: ► WWW.WWW.HWHCT.ORG	 		H(c) Gr	oup exe	mption	n number 🕨
			sociation Other 🕨	L Year	of formation	on: 19	84 N	State of legal domicile; CT
E	art I	Summary				_		
en.	1	Briefly describe the organization's mission or most						
Governance		MEN AND WOMEN, INCLUDING A	FOOD PANTRY AN	D COM	MUNIT	Y KI	TCH:	EN. CASE
rna	2	Check this box 🕨 🔲 if the organization discor	tinued its operations or dispos	sed of more	than 25%	6 of its r	et ass	ets.
SVC.	3	Number of voting members of the governing body (Part VI, line 1a)		***********		3	18
		Number of independent voting members of the gov	eming body (Part VI, line 1b)		*********		4	18
Activities &	5	Total number of individuals employed in calendar y						45
JII.	6	Total number of volunteers (estimate if necessary)						400.
cti	7 a	Total unrelated business revenue from Part VIII, col	umn (C), line 12				7a	0.
_	Ь	Net unrelated business taxable income from Form					7b	0.
						Year		Current Year
æ	8	Contributions and grants (Part VIII, line 1h)			2,3	70,9	82.	2,593,698.
Ž	9			2:	39,1	38.	273,366.	
Revenue	10	Investment income (Part VIII, column (A), lines 3, 4,			:	27,6	54.	76,999.
Œ	11	Other revenue (Part VIII, column (A), lines 5, 6d, 8c,		-:	34,1	-61,511.		
		Total revenue - add lines 8 through 11 (must equal			2,60	03,6	71.	2,882,552.
		Grants and similar amounts paid (Part IX, column (/					0.	0.
		Benefits paid to or for members (Part IX, column (A				0.	0.	
¢.	15	Salaries, other compensation, employee benefits (F	art IX, column (A), lines 5-10)		1,58	30,4	20.	1,573,293.
Expenses	16a	Professional fundraising fees (Part IX, column (A), li					0.	0.
Ď	ь	Total fundraising expenses (Part IX, column (D), line		45.		100		
ú	17	Other expenses (Part IX, column (A), lines 11a-11d,	11f-24e)		1,32	24,3	15.	1,271,722.
		Total expenses. Add lines 13-17 (must equal Part I)			2,90	04,7	35.	2,845,015.
	19	Revenue less expenses. Subtract line 18 from line			-30	01,0	64.	37,537.
Assets or	4			В	eginning of	Current	Year	End of Year
sets	20	Total assets (Part X, line 16)	***************************************		4,10	68,8	25.	4,200,858.
Ass	21	Total liabilities (Part X, line 26)			2,49	94,5	39.	2,492,513.
Net,		Net assets or fund balances. Subtract line 21 from	line 20	<u></u>	1,67	74,2	86.	1,708,345.
		Signature Block						
		lties of perjury, I declare that I have examined this return,			•		-	knowledge and belief, it is
true	, correc	t, and complete. Declaration of preparer (other than office	r) is based on all information of wh	nich prepare	r has any kr	nowledge	<u>. </u>	
Sig	ın	Signature of officer				Date		
He	re	HELEN MCALINDEN, PRESI	ENT/CEO					
_		Type or print name and title						
Print/Type preparer's name Preparer's signature Date Check PTIN								PTIN
Pai	d	MARY-EVELYN ANTONETTI				II	eff-employe	ы №00431862
Pre	parer	Firm's name MARCUM LLP				Firm's E	IN 🔽	11-1986323
Use	Only	Firm's address 185 ASYLUM STREE!	ŗ					
		HARTFORD, CT 0610				Phone n	<u>o. (</u> 8	<u>60) 760-0600</u>
Ma	y the li	RS discuss this return with the preparer shown about	/e? (see instructions)					X Yes No
	001 01-2			ons.				Form 990 (2019)

19381022 150872 S06085

Form **990** (2019)

			Yes	No
1	Is the organization described in section 501(c)(3) or 4947(a)(1) (other than a private foundation)?			
	If "Yes," complete Schedule A	1	<u>X</u>	
2	Is the organization required to complete Schedule B, Schedule of Contributors?	2	X	
3	Did the organization engage in direct or indirect political campaign activities on behalf of or in opposition to candidates for	_		3.7
	public office? If "Yes," complete Schedule C, Part I	3		<u> </u>
4	Section 501(c)(3) organizations. Did the organization engage in lobbying activities, or have a section 501(h) election in effect	_		37
_	during the tax year? If "Yes," complete Schedule C, Part II	4		<u> </u>
5	Is the organization a section 501(c)(4), 501(c)(5), or 501(c)(6) organization that receives membership dues, assessments, or	_		x
6	similar amounts as defined in Revenue Procedure 98-19? If "Yes," complete Schedule C, Part III	_5_		
0	provide advice on the distribution or investment of amounts in such funds or accounts? If "Yes," complete Schedule D, Part I	_		х
7	Did the organization receive or hold a conservation easement, including easements to preserve open space,	6		
•	the environment, historic land areas, or historic structures? If "Yes," complete Schedule D, Part II			x
8	Did the organization maintain collections of works of art, historical treasures, or other similar assets? If "Yes." complete		_	
0	, ,	8		x
9	Schedule D, Part III Did the organization report an amount in Part X, line 21, for escrow or custodial account liability, serve as a custodian for	-		
3	amounts not listed in Part X; or provide credit counseling, debt management, credit repair, or debt negotiation services?			
	If "Yes," complete Schedule D, Part IV	9	х	
10	Did the organization, directly or through a related organization, hold assets in donor-restricted endowments	9	41	
	or in quasi endowments? If "Yes," complete Schedule D, Part V	10	х	
11	If the organization's answer to any of the following questions is "Yes," then complete Schedule D, Parts VI, VII, VIII, IX, or X	10	47.3	7.7
••	as applicable.			
а	Did the organization report an amount for land, buildings, and equipment in Part X, line 10? If "Yes," complete Schedule D,	4.46	X.25.	11000000
_	Part VI	11a	х	
ь	Did the organization report an amount for investments - other securities in Part X, line 12, that is 5% or more of its total			
_	assets reported in Part X, line 16? If "Yes," complete Schedule D, Part VII	11b		X
c	Did the organization report an amount for investments - program related in Part X, line 13, that is 5% or more of its total			
	assets reported in Part X, line 16? If "Yes," complete Schedule D, Part VIII	11c		X
d	Did the organization report an amount for other assets in Part X, line 15, that is 5% or more of its total assets reported in			
	Part X, line 16? If "Yes," complete Schedule D, Part IX	11d	X	
e	Did the organization report an amount for other liabilities in Part X, line 25? If "Yes," complete Schedule D, Part X	11e	X	
f	Did the organization's separate or consolidated financial statements for the tax year include a footnote that addresses			
	the organization's liability for uncertain tax positions under FIN 48 (ASC 740)? If "Yes," complete Schedule D, Part X	11f	X	
12a	Did the organization obtain separate, independent audited financial statements for the tax year? If "Yes," complete			
	Schedule D, Parts XI and XII	12a		X
b	Was the organization included in consolidated, independent audited financial statements for the tax year?			
	If "Yes," and if the organization answered "No" to line 12a, then completing Schedule D, Parts XI and XII is optional	12b	X	<u> </u>
13	Is the organization a school described in section 170(b)(1)(A)(ii)? If "Yes," complete Schedule E	13		<u>X</u>
14a	Did the organization maintain an office, employees, or agents outside of the United States?	14a		X
b	Did the organization have aggregate revenues or expenses of more than \$10,000 from grantmaking, fundraising, business,			
	investment, and program service activities outside the United States, or aggregate foreign investments valued at \$100,000			v
4-	or more? If "Yes," complete Schedule F, Parts I and IV	14b		<u> </u>
15	Did the organization report on Part IX, column (A), line 3, more than \$5,000 of grants or other assistance to or for any	ا ہے ا		v
10	foreign organization? If "Yes," complete Schedule F, Parts II and IV	15		
16	Did the organization report on Part IX, column (A), line 3, more than \$5,000 of aggregate grants or other assistance to	16		X
47	or for foreign individuals? If "Yes," complete Schedule F, Parts III and IV Did the organization report a total of more than \$15,000 of expenses for professional fundraising services on Part IX,	16		
17	- · · · · · · · · · · · · · · · · · · ·	17		X
18	column (A), lines 6 and 11e? <i>If</i> "Yes," <i>complete Schedule G, Part I</i> Did the organization report more than \$15,000 total of fundraising event gross income and contributions on Part VIII, lines	1/		-42
10		18	х	
19	1c and 8a? If "Yes," complete Schedule G, Part II Did the organization report more than \$15,000 of gross income from gaming activities on Part VIII, line 9a? If "Yes."	- 10	-43	
13	· · · · · · · · · · · · · · · · ·	19		x
20a	complete Schedule G, Part III Did the organization operate one or more hospital facilities? If "Yes," complete Schedule H	20a		$\frac{x}{x}$
		20a	-	
21	Did the organization report more than \$5,000 of grants or other assistance to any domestic organization or			
	domestic government on Part IX, column (A), line 1? If "Yes," complete Schedule I. Parts I and II	21		x
	25. Complete Scripting Street, 49 and 7. II 169, Complete Scripting Falls I and II	<u> </u>		

Par	t IV Checklist of Required Schedules (continued)		_						
	,		Yes	No					
22	Did the organization report more than \$5,000 of grants or other assistance to or for domestic individuals on								
	Part IX, column (A), line 2? If "Yes," complete Schedule I, Parts I and III	22		х					
23	Did the organization answer "Yes" to Part VII, Section A, line 3, 4, or 5 about compensation of the organization's current								
	and former officers, directors, trustees, key employees, and highest compensated employees? If "Yes," complete								
	Schedule J	23	х						
24a	Did the organization have a tax-exempt bond issue with an outstanding principal amount of more than \$100,000 as of the								
	last day of the year, that was issued after December 31, 2002? If "Yes," answer lines 24b through 24d and complete								
	Schedule K. If "No," go to line 25a	24a		x					
h	b Did the organization invest any proceeds of tax-exempt bonds beyond a temporary period exception?								
	Did the organization maintain an escrow account other than a refunding escrow at any time during the year to defease	24b							
•									
ч	any tax-exempt bonds? d Did the organization act as an "on behalf of" issuer for bonds outstanding at any time during the year?								
		24d							
2 5a	Section 501(c)(3), 501(c)(4), and 501(c)(29) organizations. Did the organization engage in an excess benefit	00-		x					
	transaction with a disqualified person during the year? If "Yes," complete Schedule L, Part I	25a							
D	Is the organization aware that it engaged in an excess benefit transaction with a disqualified person in a prior year, and	\	,						
	that the transaction has not been reported on any of the organization's prior Forms 990 or 990-EZ? If "Yes," complete			77					
	Schedule L, Part I	25b		<u>X</u>					
26	Did the organization report any amount on Part X, line 5 or 22, for receivables from or payables to any current			-					
	or former officer, director, trustee, key employee, creator or founder, substantial contributor, or 35%								
	controlled entity or family member of any of these persons? If "Yes," complete Schedule L, Part II	26		<u>X</u>					
27	Did the organization provide a grant or other assistance to any current or former officer, director, trustee, key employee,								
	creator or founder, substantial contributor or employee thereof, a grant selection committee member, or to a 35% controlled								
	entity (including an employee thereof) or family member of any of these persons? If "Yes," complete Schedule L, Part III	27		X					
28	Was the organization a party to a business transaction with one of the following parties (see Schedule L, Part IV								
	instructions, for applicable filing thresholds, conditions, and exceptions):								
а	a A current or former officer, director, trustee, key employee, creator or founder, or substantial contributor? If								
	"Yes," complete Schedule L, Part IV								
b	b A family member of any individual described in line 28a? If "Yes," complete Schedule L, Part IV								
C	A 35% controlled entity of one or more individuals and/or organizations described in lines 28a or 28b? If								
	"Yes," complete Schedule L, Part IV	28c		X					
29	Did the organization receive more than \$25,000 in non-cash contributions? If "Yes," complete Schedule M	29	X						
30	Did the organization receive contributions of art, historical treasures, or other similar assets, or qualified conservation								
	contributions? If "Yes," complete Schedule M	30	<u> </u>	X					
31	Did the organization liquidate, terminate, or dissolve and cease operations? If "Yes," complete Schedule N, Part I	31	<u> </u>	X					
32	Did the organization sell, exchange, dispose of, or transfer more than 25% of its net assets? If "Yes," complete								
	Schedule N, Part II	32		X					
33	Did the organization own 100% of an entity disregarded as separate from the organization under Regulations		1	-					
	sections 301.7701-2 and 301.7701-3? If "Yes," complete Schedule R, Part I	33		X					
34	Was the organization related to any tax-exempt or taxable entity? If "Yes," complete Schedule R, Part II, III, or IV, and		1	1					
	Part V, line 1	34	X						
35a	Did the organization have a controlled entity within the meaning of section 512(b)(13)?	35a		X					
b	If "Yes" to line 35a, did the organization receive any payment from or engage in any transaction with a controlled entity								
	within the meaning of section 512(b)(13)? If "Yes," complete Schedule R, Part V, line 2	35b							
36	Section 501(c)(3) organizations. Did the organization make any transfers to an exempt non-charitable related organization?								
	If "Yes," complete Schedule R, Part V, line 2	36		X					
37	Did the organization conduct more than 5% of its activities through an entity that is not a related organization								
	and that is treated as a partnership for federal income tax purposes? If "Yes," complete Schedule R, Part VI	37		X					
38	Did the organization complete Schedule O and provide explanations in Schedule O for Part VI, lines 11b and 19?			-					
	Note: All Form 990 filers are required to complete Schedule O	38	x						
Par	t V Statements Regarding Other IRS Filings and Tax Compliance								
	Check if Schedule O contains a response or note to any line in this Part V								
			Yes	No					
1a	Enter the number reported in Box 3 of Form 1096. Enter -0- if not applicable	for etc.							
	Enter the number of Forms W-2G included in line 1a. Enter -0- if not applicable 1b 0	- Comment 1	6.						
	Did the organization comply with backup withholding rules for reportable payments to vendors and reportable gaming	1	196	PA.					
•	(gambling) winnings to prize winners?	1c	x	-					
932004	4 01-20-20		1	(2019)					

	_			Yes	No			
2a	Enter the number of employees reported on Form W-3, Transmittal of Wage and Tax Statements,							
	filed for the calendar year ending with or within the year covered by this return	2a 45			15.0			
b	if at least one is reported on line 2a, did the organization file all required federal employment tax return		2b	X				
	Note: If the sum of lines 1a and 2a is greater than 250, you may be required to e-file (see instructions		1.1	1.7	524			
			3a		X			
	If "Yes," has it filed a Form 990-T for this year? If "No" to line 3b, provide an explanation on Schedule		3b					
4a	At any time during the calendar year, did the organization have an interest in, or a signature or other a			•	l			
	financial account in a foreign country (such as a bank account, securities account, or other financial account,	count)?	4a	AND TO	<u>X</u>			
þ	If "Yes," enter the name of the foreign country			met.	14			
_	See instructions for filing requirements for FinCEN Form 114, Report of Foreign Bank and Financial Ac	counts (FBAR).	1		3.4			
_			<u>5a</u> 5b		X			
_	b Did any taxable party notify the organization that it was or is a party to a prohibited tax shelter transaction?							
_	c If "Yes" to line 5a or 5b, did the organization file Form 8886-T?							
6a	Does the organization have annual gross receipts that are normally greater than \$100,000, and did the	-			37			
	any contributions that were not tax deductible as charitable contributions?		6a	_	X			
b	If "Yes," did the organization include with every solicitation an express statement that such contribution	-						
-	were not tax deductible?	***************************************	6b	312	Subsective 1			
7	Organizations that may receive deductible contributions under section 170(c).		salis.	X	all art :			
a	Did the organization receive a payment in excess of \$75 made partly as a contribution and partly for goods and send	, , ,	7a	X				
b	If "Yes," did the organization notify the donor of the value of the goods or services provided? Did the organization sell, exchange, or otherwise dispose of tangible personal property for which it wa		7b		<u>. </u>			
C		•	7.		x			
d		7d	7c		-A			
e	If "Yes," indicate the number of Forms 8282 filed during the year Did the organization receive any funds, directly or indirectly, to pay premiums on a personal benefit co		7e		X			
f	Did the organization, during the year, pay premiums, directly or indirectly, on a personal benefit contra		7 1		x			
g g	If the organization received a contribution of qualified intellectual property, did the organization file For		7g	_				
h	If the organization received a contribution of cars, boats, airplanes, or other vehicles, did the organization		7h					
8	Sponsoring organizations maintaining donor advised funds. Did a donor advised fund maintained				y WM			
	sponsoring organization have excess business holdings at any time during the year?	- ,	8		-			
9	Sponsoring organizations maintaining donor advised funds.				Av.a			
а								
b	Did the sponsoring organization make a distribution to a donor, donor advisor, or related person?		9b					
10	Section 501(c)(7) organizations. Enter:		3 +02		1944			
а	Initiation fees and capital contributions included on Part VIII, line 12	10a		- 75				
b	Gross receipts, included on Form 990, Part VIII, line 12, for public use of club facilities	10b						
11	Section 501(c)(12) organizations. Enter:	•		1	÷ 3-			
а	Gross income from members or shareholders	11a	- 3-,	*. 1				
b	Gross income from other sources (Do not net amounts due or paid to other sources against		W.C.					
	amounts due or received from them.)	116	NV JA	1.5	6.73			
	Section 4947(a)(1) non-exempt charitable trusts. Is the organization filing Form 990 in lieu of Form		12a	800. [323] Long.	and bulleting			
	,	12b	envi e					
13	Section 501(c)(29) qualified nonprofit health insurance issuers.		are to 1 Mr.	Second Second	A 48 3			
а	Is the organization licensed to issue qualified health plans in more than one state?		13a	/: <u>**********************************</u>	organism in			
	Note: See the instructions for additional information the organization must report on Schedule O.				W-12			
O	Enter the amount of reserves the organization is required to maintain by the states in which the	ا ممر ا			3.94			
	organization is licensed to issue qualified health plans	13b						
	Enter the amount of reserves on hand	_13c	44-		X			
14a		. ^	14a	 	├^			
	If "Yes," has it filed a Form 720 to report these payments? If "No," provide an explanation on Schedul		14b	├	 			
15	Is the organization subject to the section 4960 tax on payment(s) of more than \$1,000,000 in remuner		45		x			
	excess parachute payment(s) during the year?		15	1871-12	A.			
16	If "Yes," see instructions and file Form 4720, Schedule N.	incomo?		کک	X			
16	Is the organization an educational institution subject to the section 4968 excise tax on net investment If "Yes," complete Form 4720, Schedule O.	income:	16		42			
	is 165, Complete Form 4720, Gonedule O.		11.07 × 1.3754000	PART BE	Para an ye			

HOMES WITH HOPE, INC. 22-2534326 Part VI Governance, Management, and Disclosure For each "Yes" response to lines 2 through 7b below, and for a "No" response to line 8a, 8b, or 10b below, describe the circumstances, processes, or changes on Schedule O. See instructions. Check if Schedule O contains a response or note to any line in this Part VI Section A. Governing Body and Management 1a Enter the number of voting members of the governing body at the end of the tax year If there are material differences in voting rights among members of the governing body, or if the governing body delegated broad authority to an executive committee or similar committee, explain on Schedule O. **b** Enter the number of voting members included on line 1a, above, who are independent Did any officer, director, trustee, or key employee have a family relationship or a business relationship with any other officer, director, trustee, or key employee? Did the organization delegate control over management duties customarily performed by or under the direct supervision of officers, directors, trustees, or key employees to a management company or other person? X 3 X Did the organization make any significant changes to its governing documents since the prior Form 990 was filed? 4 Did the organization become aware during the year of a significant diversion of the organization's assets? 5 Did the organization have members or stockholders? 6 7a Did the organization have members, stockholders, or other persons who had the power to elect or appoint one or more members of the governing body? 7a b Are any governance decisions of the organization reserved to (or subject to approval by) members, stockholders, or persons other than the governing body? 7b 44 Did the organization contemporaneously document the meetings held or written actions undertaken during the year by the following: X a The governing body? Яа b Each committee with authority to act on behalf of the governing body? Is there any officer, director, trustee, or key employee listed in Part VII, Section A, who cannot be reached at the 9 organization's mailing address? If "Yes." provide the names and addresses on Schedule O Section B. Policies (This Section B requests information about policies not required by the Internal Revenue Code.) Yes No 10a Did the organization have local chapters, branches, or affiliates? Х 10a b If "Yes." did the organization have written policies and procedures governing the activities of such chapters, affiliates, and branches to ensure their operations are consistent with the organization's exempt purposes? 10b 11a Has the organization provided a complete copy of this Form 990 to all members of its governing body before filing the form? 11a b Describe in Schedule O the process, if any, used by the organization to review this Form 990. 12a Did the organization have a written conflict of interest policy? If "No," go to line 13 X 12a X b Were officers, directors, or trustees, and key employees required to disclose annually interests that could give rise to conflicts? 12b c Did the organization regularly and consistently monitor and enforce compliance with the policy? if "Yes." describe 12c in Schedule O how this was done X 13 Did the organization have a written whistleblower policy? 13 14 Did the organization have a written document retention and destruction policy? X 14 15 Did the process for determining compensation of the following persons include a review and approval by independent persons, comparability data, and contemporaneous substantiation of the deliberation and decision? X 15a a The organization's CEO, Executive Director, or top management official **b** Other officers or key employees of the organization 15b If "Yes" to line 15a or 15b, describe the process in Schedule O (see instructions). 16a Did the organization invest in, contribute assets to, or participate in a joint venture or similar arrangement with a taxable entity during the year? 16a b If "Yes," did the organization follow a written policy or procedure requiring the organization to evaluate its participation in joint venture arrangements under applicable federal tax law, and take steps to safeguard the organization's 16b exempt status with respect to such arrangements? Section C. Disclosure List the states with which a copy of this Form 990 is required to be filed ▶CT Section 6104 requires an organization to make its Forms 1023 (1024 or 1024-A, if applicable), 990, and 990-T (Section 501(c)(3)s only) available for public inspection. Indicate how you made these available. Check all that apply. X Own website Another's website X Upon request Other (explain on Schedule O) Describe on Schedule O whether (and if so, how) the organization made its governing documents, conflict of interest policy, and financial statements available to the public during the tax year. State the name, address, and telephone number of the person who possesses the organization's books and records HELEN MCALINDEN, PRESIDENT/CEO - 203-226-3426 212. WESTPORT. 49 RICHMONDVILLE AVE, SUITE

Part VII Compensation of Officers, Directors, Trustees, Key Employees, Highest Compensated Employees, and Independent Contractors

Check if Schedule O contains a response or note to any line in this Part VII

Section A. Officers, Directors, Trustees, Key Employees, and Highest Compensated Employees

- 1a Complete this table for all persons required to be listed. Report compensation for the calendar year ending with or within the organization's tax year.
- List all of the organization's current officers, directors, trustees (whether individuals or organizations), regardless of amount of compensation. Enter -0- in columns (D), (E), and (F) if no compensation was paid.
 - List all of the organization's current key employees, if any. See instructions for definition of "key employee."
- List the organization's five current highest compensated employees (other than an officer, director, trustee, or key employee) who received reportable compensation (Box 5 of Form W-2 and/or Box 7 of Form 1099-MISC) of more than \$100,000 from the organization and any related organizations.
- List all of the organization's former officers, key employees, and highest compensated employees who received more than \$100,000 of reportable compensation from the organization and any related organizations.
- List all of the organization's former directors or trustees that received, in the capacity as a former director or trustee of the organization, more than \$10,000 of reportable compensation from the organization and any related organizations.

 See instructions for the order in which to list the persons above.

Check this box if neither the organization nor any related organization compensated any current officer, director, or trustee.

(A)	(B)	(C)					(D)	(E)	(F)	
Name and title	Average	l (do	Position (do not check more than one				one.	Reportable	Reportable	Estimated
	hours per	box	nless person is both an rand a director/trustee)				compensation	compensation	amount of	
	week	-		and a director/rustee)			te e)	from	from related	other
	(list any	recto						the	organizations	compensation
	hours for	ō	66			ated		organization	(W-2/1099-MISC)	from the
	related organizations	ustee	trust		 æ	iben:		(W-2/1099-MISC)		organization and related
	below	ag 타	tional) Se	ag ag	_			organizations
	line)	Individual trustee or director	Institutional trustee	Officer	Key employee	Highest compensated employee	Former	1		organizations
(1) JOHN WALSH	0.40	_	Ī	۲	1		┢			
CHAIR	0.30	x		X				0.	.0.	0.
(2) JENNIFER FERRANTE	0.40								-	
VICE CHAIR	0.30	X		X				0.	0.	0.
(3) BRUCE HENNEMUTH	0.40									
TREASURER	0.30	X		X				.0.	0.	0.
(4) CAROL RANDEL	0.40									_
DIRECTOR/SECRETARY	0.30	X		X				0.	0.	0.
(5) BRIAN BRAXENDALE	0.40									
DIRECTOR	0.30	X						0.	0.	0.
(6) DALE FREDSTON	0.40									_
DIRECTOR	0.30	X						0.	0.	0.
(7) BRUCE GAYLORD	0.40									
DIRECTOR	0.30	X						0.	0.	0.
(8) ALLYSON GOTTLIEB	0.40									
DIRECTOR (FROM 9/2019)	0.30	X		<u> </u>	L	L	_	0.	0.	0.
(9) JEFFREY GURREN	0.40	l								
DIRECTOR	0.30	X	_					0.	0.	
(10) KIMBERLY LAKE	0.40	1		l						
DIRECTOR (TO 9/2019)	0.30	X		_			_	0.	0.	0.
(11) MONICA LAZARO	0.40							<u> </u>	_	_
DIRECTOR (TO 5/2019)	0.30	X		<u> </u>	L	Щ		0.	0.	0.
(12) NATASHA LIPCAN	0.40				1]	_	_	_
DIRECTOR	0.30	Х					L	0.	0.	0.
(13) REBECCA MARTIN	0.40								1	
DIRECTOR (FROM 9/2019)	0.30	X				<u> </u>		0.	0.	0.
(14) BETH MASSOUD	0.40									
DIRECTOR (FROM 2/2019)	0.30	X	L	L	L		<u> </u>	0.	0.	0.
(15) RACHEL MEISEL	0.40]								
SECRETARY/DIRECTOR	0.30	X		X				0.	0.	0.
(16) IAN O'MALLEY	0.40]								
DIRECTOR	0.30	X						0.	0.	0.
(17) HAROLD SHUPACK	0.40]		1]	
DIRECTOR	0.30	X	L_	L			L_	0.	.0.	0.
932007_01-20-20										Form 990 (2019)

932007 01-20-20

Form **990** (2019)

Form 990 (2019) HOMES WIT									22-25	34	326	Page 8
Part VII Section A. Officers, Directors, Trus	tees, Key Emp	oloy	ees,	and	Hig	ghes	st C	ompensated Employees	s (continued)			
(A)								F)				
Name and title	Name and title Average Po		Pos				Reportable	Reportable			nated	
	hours per	(do not check more than one box, unless person is both an officer and a director/trustee)				than d s both	one nan	compensation	compensation	n I		unt of
	week							from	from related			her
	(list any	햠						the	organizations			ensation
	hours for	r dire				eg eg		organization	(W-2/1099-MIS		-	n the
	related	tee o	nstee			ensat		(W-2/1099-MISC)			orgar	ization
	organizations	l trus	nal tr		eg ée	E .					and i	elated
	below	ndividual trustee or director	nstitutional trustee	5	Key employee	lest c Toyee	늘				organ	izations
	line)	i ba	Insti	.Officer	Key	Highest compensated employee	Former					
(18) LAUREN SOLOFF	0.40											
DIRECTOR	0.30	X						0.		0.		0.
(19) CHAN WHEELER	0.40											
DIRECTOR	0.30	X						0.		0.		0.
(20) MISSY ZAHLER	0.40											
DIRECTOR	0.30	x						0.		0.		0.
(21) JEFFREY N. WIESER	40.00	-		-	┢	 	┢					
PRESIDENT/CEO	0.30	ł		х				145,000.		0.	20	112
(22) JACQUELINE HOGAN	35.00	⊢	\vdash	42	├	-	-	143,000.		٠.		<u>,112.</u>
COMPTROLLER	0.30	ł		٠,				E0 402			2.4	co.c
		⊢	_	X	┝		⊢	58,493.		0.		<u>,686.</u>
(23) PARIS LOONEY	40.00	ŀ						00.440		ا ہ	•	
VICE PRESIDENT/COO	0.30	<u> </u>	<u> </u>	Х		<u> </u>	_	90,143.		0.	9	<u>,477.</u>
		ļ										
		_			_	_	<u> </u>					· ·
-	<u> </u>										-	••
							_	<u></u>				
_												
1b Subtotal								293,636.		0.	54	,275.
c Total from continuation sheets to Part VII	l, Section A							0.		0.		0.
d Total (add lines 1b and 1c)							•	293,636.		0.	54	,275.
2 Total number of individuals (including but no							o re	eceived more than \$100.0	000 of reportable			
compensation from the organization						,		,				1.
							_				Ty	es No
3 Did the organization list any former officer,	director trusts	امما		mnl	OVO	0 Or	hio	heet compensated ample	Oved on	í		3. C. (3. V.)
					-		_		=			Х
line 1a? If "Yes," complete Schedule J for si	ucn individuai et renestebl					٠٠٠٠٠٠					3	criticism of lagrancing stress
4 For any individual listed on line 1a, is the su												X
and related organizations greater than \$150												
5 Did any person listed on line 1a receive or a	-				-			- ·				
rendered to the organization? If "Yes." com	plete Schedule	J f	or st	ıch ı	oers	on .					5	X X
Section B. Independent Contractors								 	 			
Complete this table for your five highest cor		-								ensat	tion from	J
the organization. Report compensation for t	the calendar ye	ear e	ndir	<u>ıg w</u>	<u>ith c</u>	y wi	<u>thin</u>	the organization's tax ye	ear.			
(A)			_					(B)		_	(C)	
Name and business	address	Ŋ	NI	3			_	Description of se	ervices	<u> </u>	ompens	ation
aa												
							\neg					
							- 1					
							\dashv				-	
							- {					
2 Total number of independent contract—— #-	noludina but	at 1:-	nito:	1 40	ther	na 11-	٠	ahaya) wha reast and	ro than	- 1 C		**; L. 4.1
2 Total number of independent contractors (in	-	טון זונו	me	1 (0)	_		red	above) who received mo				
\$100,000 of compensation from the organization	zation 📂							<u> </u>		Car.		-7-249/

Page 9

Statement of Revenue

Check if Schedule O contains a response or note to any line in this Part VIII Related or exempt Unrelated Revenue excluded Total revenue from tax under function revenue business revenue sections 512 - 514 1 a Federated campaigns 1a **b** Membership dues 1b c Fundraising events 508,982 1c d Related organizations 1d 726,407 e Government grants (contributions) f All other contributions, gifts, grants, and 1,358,309 similar amounts not included above 375,590 g Noncash contributions included in lines 1a-1f h Total. Add lines 1a-1f Business Code 624200 266,347 2 a HOUSING FEES 266,347 SHELTER FEES 624200 f All other program service revenue 273,366. g Total. Add lines 2a-2f Investment income (including dividends, interest, and 19,745. 19,745. other similar amounts) Income from investment of tax-exempt bond proceeds 5 Royalties (i) Real (ii) Personal 6 a Gross rents b Less: rental expenses c Rental income or (loss) d Net rental income or (loss) 7 a Gross amount from sales of (i) Securities (ii) Other 7a 438,665. assets other than inventory b Less: cost or other basis 7ь 381,411 and sales expenses c Gain or (loss) d Net gain or (loss) 8 a Gross income from fundraising events (not including \$ 508,982. of contributions reported on line 1c). See 90,166 Part IV, line 18 вь 164,230. b Less: direct expenses c Net income or (loss) from fundraising events 9 a Gross income from gaming activities. See Part IV, line 19 b Less: direct expenses c Net income or (loss) from gaming activities 10 a Gross sales of inventory, less returns and allowances 10b **b** Less: cost of goods sold c Net income or (loss) from sales of inventory **Business Code** scellaneous 11 a OTHER INCOME 900099 12,553. 12,553. d All other revenue 12,553. Total. Add lines 11a-11d 2,882,552. 2,935. 285,919 Total revenue. See instructions

Section 501(c)(3) and 501(c)(4) organizations must complete all columns. All other organizations must complete column (A). Check if Schedule O contains a response or note to any line in this Part IX (A) Total expenses (B) Program service (D) Do not include amounts reported on lines 6b, Management and general expenses Fundraising 7b. 8b. 9b. and 10b of Part VIII. expenses expenses Grants and other assistance to domestic organizations e de la companya de and domestic governments, See Part IV, line 21 AND DESCRIPTION Grants and other assistance to domestic individuals. See Part IV, line 22 Grants and other assistance to foreign organizations, foreign governments, and foreign individuals. See Part IV, lines 15 and 16 uz ve slove (1). Benefits paid to or for members Compensation of current officers, directors, 347,911. 127,436. 128,079. 92,396. trustees, and key employees Compensation not included above to disqualified 6 persons (as defined under section 4958(f)(1)) and persons described in section 4958(c)(3)(B) Other salaries and wages 973,896. 900,900. 7,972. 65,024. Pension plan accruals and contributions (include section 401(k) and 403(b) employer contributions) Other employee benefits 144,370. 115,426. 12,266. 16,678. 9 12,374. 107,116. 85,641. 9.101. 10 Payroll taxes Fees for services (nonemployees): 7,248. 7.248. a Management 845. 845. b Legal 36,580. 36,580. c Accounting d Lobbying teran generatana kuratan kuratan da Professional fundraising services. See Part IV. line 17 investment management fees _____ g Other. (If line 11g amount exceeds 10% of line 25, 110,317. 109,620. 697. column (A) amount, list line 11g expenses on Sch O.) Advertising and promotion 12 25,376. 24.058. 117.924. 68,490. 13 Office expenses Information technology 14 Royalties 15 224,943. 223,808. 1,135. Occupancy 16 17,444. 13,082. 4.362. 17 Travel 18 Payments of travel or entertainment expenses for any federal, state, or local public officials ... 540. 405. 135. Conferences, conventions, and meetings 19 87,371. 87,371. 20 Interest Payments to affiliates 21 117,111. 103,722. 13,389. 22 Depreciation, depletion, and amortization 18,031.72.418. 54,387. 23 Insurance Other expenses. Itemize expenses not covered above (List miscellaneous expenses on line 24e. If line 24e amount exceeds 10% of line 25, column (A) ar and amount, list line 24e expenses on Schedule O.) 461,776. 461,776. SHELTER & OTHER PROGRAM 9,997. 9,997. FUNDRAISING EXPENSES 623. 2,498. 1,875 MEMBERSHIP/SUBSCRIPTION d 4,710. $\frac{4}{.710}$ e All other expenses 264,521. 2,845,015. 2,358,649. 221,845. Total functional expenses. Add lines 1 through 24e Joint costs. Complete this line only if the organization reported in column (B) joint costs from a combined educational campaign and fundraising solicitation. Check here if following SOP 98-2 (ASC 958-720)

Form **990** (2019)

	2	Check if Schedule O contains a response or note	to anv	line in this Part X				
						(A) Beginning of year		(B) End of year
-	1	Cash - non-interest-bearing				528,226.	1	473,429.
J	2	Savings and temporary cash investments				120,183.	2	217,508.
	3	Pledges and grants receivable, net				28,891.	3	113,849.
	4	Accounts receivable, net				11,015.	4	100,548.
	5	Loans and other receivables from any current or	e e e e e e e e e e e e e e e e e e e		Canada de la casa de la			
		trustee, key employee, creator or founder, substa		200				
		controlled entity or family member of any of thes		5				
	6	Loans and other receivables from other disqualif			Selected Accepts			
		under section 4958(f)(1)), and persons described	in secti	on 4958(c)(3)(B)	*****		6	
2	7	Notes and loans receivable, net					7	
Assets	8	Inventories for sale or use		•••••	· · · · · · · · ·		8	
₹	9			*************		24,288.	9	10,113.
	10a	Land, buildings, and equipment: cost or other						
		basis. Complete Part VI of Schedule D	10a	3,899,8	362.			
	b	Less: accumulated depreciation	10b	1,398,0)35.	2,606,572.	10c	
l	11	Investments - publicly traded securities		******************		498,537.	11	368,675.
1	12	Investments - other securities. See Part IV, line 1	1				12	
	13	Investments - program-related. See Part IV, line 1	1				13	
	14	Intangible assets		• • • • • • • • • • • • • • • • • • • •			14	
	15	Other assets. See Part IV, line 11		351,113.	15	414,909.		
	16	Total assets. Add lines 1 through 15 (must equa				4,168,825.	16	4,200,858.
	17	Accounts payable and accrued expenses	92,584.	17	83,243.			
	18	Grants payable		•••••			18	
	19	Deferred revenue		37,862.	19	49,323.		
	20	Tax-exempt bond liabilities					20	
	21	Escrow or custodial account liability. Complete F				1,752.	21	1,752.
ဖ္က	22	Loans and other payables to any current or form	er office	r, director,			41.0	
Liabilities		trustee, key employee, creator or founder, subst	antiál co	intributor, or 35%				
jabi		controlled entity or family member of any of thes					22	
<u> </u>	23	Secured mortgages and notes payable to unrelate	ted third	l parties		2,351,770.	23	2,347,624.
	24	Unsecured notes and loans payable to unrelated					24	
	25	Other liabilities (including federal income tax, pay	/ables to	o related third				
		parties, and other liabilities not included on lines	17-24).	Complete Part X				
		of Schedule D		•••••		10,571.	25	10,571.
_	26					2,494,539.	26	2,492,513.
ا ۾		Organizations that follow FASB ASC 958, che	ck here	► X		En 1753 de 1760 fabricas (1504) de 1606 en 1770 La la Magnetia de la la Calebra (1504) de 1606 de 1770 de		er gelgich i Argen eine gan, der in der ge-
ĕ		and complete lines 27, 28, 32, and 33.				1 140 000	- 1	1 102 251
ag	27	Net assets without donor restrictions		1,149,292.	27	1,183,351.		
Net Assets or Fund Balances	28	Net assets with donor restrictions			 1	524,994.	28	524,994.
ַבַּ		Organizations that do not follow FASB ASC 958, check here						
튛		and complete lines 29 through 33.						
ş	29						29	
sse	30	Paid-in or capital surplus, or land, building, or eq	•			 	30	
ا پا	31	Retained earnings, endowment, accumulated inc				1 674 206	31	1 700 245
S	32	Total net assets or fund balances				1,674,286.	32	1,708,345.
	33	Total liabilities and net assets/fund balances				4,168,825.	33	4,200,858. Form 990 (2019)

Both consolidated and separate basis

c If "Yes" to line 2a or 2b, does the organization have a committee that assumes responsibility for oversight of the audit,

review, or compilation of its financial statements and selection of an independent accountant?

If the organization changed either its oversight process or selection process during the tax year, explain on Schedule O. 3a As a result of a federal award, was the organization required to undergo an audit or audits as set forth in the Single Audit Act and OMB Circular A-133?

b If "Yes," did the organization undergo the required audit or audits? If the organization did not undergo the required audit

or audits, explain why on Schedule O and describe any steps taken to undergo such audits

X 3a

2c

2

3

4

ß

7

8

9

10

SCHEDULE A

(Form 990 or 990-EZ)

Department of the Treasury Internal Revenue Service

Public Charity Status and Public Support

Complete if the organization is a section 501(c)(3) organization or a section 4947(a)(1) nonexempt charitable trust.

➤ Attach to Form 990 or Form 990-EZ.

Go to www.irs.gov/Form990 for instructions and the latest information.

OMB No. 1545-0047 Open to Public Inspection

Name of the organization

Employer identification number HOMES WITH HOPE, INC. 22-2534326 Reason for Public Charity Status (All organizations must complete this part.) See instructions. The organization is not a private foundation because it is: (For lines 1 through 12, check only one box.) 1 A church, convention of churches, or association of churches described in section 170(b)(1)(A)(i). 2 A school described in section 170(b)(1)(A)(ii), (Attach Schedule E (Form 990 or 990-EZ),) 3 A hospital or a cooperative hospital service organization described in section 170(b)(1)(A)(iii). A medical research organization operated in conjunction with a hospital described in section 170(b)(1)(A)(iii). Enter the hospital's name, city, and state: An organization operated for the benefit of a college or university owned or operated by a governmental unit described in section 170(b)(1)(A)(iv). (Complete Part II.) A federal, state, or local government or governmental unit described in section 170(b)(1)(A)(v). X An organization that normally receives a substantial part of its support from a governmental unit or from the general public described in section 170(b)(1)(A)(vi). (Complete Part II.) A community trust described in section 170(b)(1)(A)(vi). (Complete Part II.) An agricultural research organization described in section 170(b)(1)(A)(ix) operated in conjunction with a land-grant college or university or a non-land-grant college of agriculture (see instructions). Enter the name, city, and state of the college or university: An organization that normally receives: (1) more than 33 1/3% of its support from contributions, membership fees, and gross receipts from activities related to its exempt functions - subject to certain exceptions, and (2) no more than 33 1/3% of its support from gross investment income and unrelated business taxable income (less section 511 tax) from businesses acquired by the organization after June 30, 1975. See section 509(a)(2). (Complete Part III.) An organization organized and operated exclusively to test for public safety. See section 509(a)(4). An organization organized and operated exclusively for the benefit of, to perform the functions of, or to carry out the purposes of one or more publicly supported organizations described in section 509(a)(1) or section 509(a)(2). See section 509(a)(3). Check the box in lines 12a through 12d that describes the type of supporting organization and complete lines 12e, 12f, and 12g. Type I. A supporting organization operated, supervised, or controlled by its supported organization(s), typically by giving the supported organization(s) the power to regularly appoint or elect a majority of the directors or trustees of the supporting organization. You must complete Part IV. Sections A and B. Type II. A supporting organization supervised or controlled in connection with its supported organization(s), by having control or management of the supporting organization vested in the same persons that control or manage the supported organization(s). You must complete Part IV, Sections A and C. Type III functionally integrated. A supporting organization operated in connection with, and functionally integrated with, its supported organization(s) (see instructions). You must complete Part IV, Sections A, D, and E. Type III non-functionally integrated. A supporting organization operated in connection with its supported organization(s) that is not functionally integrated. The organization generally must satisfy a distribution requirement and an attentiveness requirement (see instructions). You must complete Part IV, Sections A and D, and Part V. Check this box if the organization received a written determination from the IRS that it is a Type I, Type II, Type III functionally integrated, or Type III non-functionally integrated supporting organization. f Enter the number of supported organizations g Provide the following information about the supported organization(s). (i) Name of supported (ii) EIN (iii) Type of organization (v) Amount of monetary (vi) Amount of other your gov ng documen organization (described on lines 1-10 support (see instructions) support (see instructions) Yes No above (see instructions)) Total

Part II Support Schedule for Organizations Described in Sections 170(b)(1)(A)(iv) and 170(b)(1)(A)(vi)

(Complete only if you checked the box on line 5, 7, or 8 of Part I or if the organization failed to qualify under Part III. If the organization fails to qualify under the tests listed below, please complete Part III.)

Sec	tion A. Public Support	,,,					
	ndar year (or fiscal year beginning in)	(a) 2015	(b) 2016	(c) 2017	(d) 2018	(e) 2019	(f) Total
	Gifts, grants, contributions, and	1-,	,	12,	1-1-1-1-1	1-1	177.3
	membership fees received. (Do not						
	include any "unusual grants.")	1786625.	1872952.	2264091.	2370982.	2593698.	10888348.
2	Tax revenues levied for the organ-		·				
	ization's benefit and either paid to						
	or expended on its behalf						
3	The value of services or facilities						
	furnished by a governmental unit to						
	the organization without charge	301,406.	305,324.	311,736.	319,218.	324,964.	1562648.
4	Total. Add lines 1 through 3	2088031.	2178276.	2575827.	2690200.	2918662.	12450996.
5	The portion of total contributions			10 10 10 10 10 10 10 10 10 10 10 10 10 1		let a arv a v	
	by each person (other than a				Constant Constant Const		
	governmental unit or publicly		And Annual Con-	ne da como de se el			
	supported organization) included	1481.	- 1297	aleman kara			
	on line 1 that exceeds 2% of the			and the section is the			
	amount shown on line 11,						
	column_(f)	DEPENDEN.	3.004.00		And A proper		113,497.
6	Public support. Subtract line 5 from line 4.	FF CARNE					12337499.
Se	ction B. Total Support						
Cale	ndar year (or fiscal year beginning in) 🕨	(a) 2015	(b) 2016	(c) 2017	(d) 2018	(e) 2019	(f) Total
7	Amounts from line 4	2088031.	2178276.	2575827.	2690200.	2918662.	12450996.
8	Gross income from interest,						
	dividends, payments received on						
	securities loans, rents, royalties,						
	and income from similar sources	16,986.	16,737.	14,493.	16,369.	19,745.	84,330.
9	Net income from unrelated business						-
	activities, whether or not the						
	business is regularly carried on						
10	Other income. Do not include gain						
	or loss from the sale of capital						
	assets (Explain in Part VI.)	1,191.	10,629.	7,009.	8,440.	12,553.	39,822.
11	Total support. Add lines 7 through 10		and a second	a zou definition	A A C THE		12575148.
12	Gross receipts from related activities,	etc. (see instruction	ons)	***************************************		12 1	,283,445.
13	First five years. If the Form 990 is for	r the organization's	s first, second, third	d, fourth, or fifth ta	x year as a section	501(c)(3)	
<u>c.</u>	organization, check this box and stor	here					
	ction C. Computation of Publi		-				00 11
	Public support percentage for 2019 (I		-			14	98.11 % 80.51 %
	Public support percentage from 2018					15	
16a	33 1/3% support test - 2019. If the						
	stop here. The organization qualifies						
	33 1/3% support test - 2018. If the						
44.	and stop here. The organization qual						
178	10% -facts-and-circumstances test	=					
	and if the organization meets the "fac			•	•	-	
_	meets the "facts and circumstances"						
ŀ	10% -facts-and-circumstances test	_					
	more, and if the organization meets the				-		e ⊾
	organization meets the "facts-and-circ		' - '	·			▝
<u>18</u>	Private foundation. If the organization	on did not check a	pox on line 13, 16	a, 160, 1/a, or 17b		-	
					Sche	suule A (FOTM 991) or 990-EZ) 2019

Schedule A (Form 990 or 990-EZ) 2019 HOMES WITH HOPE, INC. [Part III | Support Schedule for Organizations Described in Section 509(a)(2)

(Complete only if you checked the box on line 10 of Part I or if the organization failed to qualify under Part II. If the organization fails to qualify under the tests listed below, please complete Part II.)

Sec	ction A. Public Support	ciow, picaco cemp	Jote Furt II.,				_	
Cale	ndar year (or fiscal year beginning in) 🕨	(a) 2015	(b) 2016	(c) 2017	(d) 2018	(e) 2019	(f) Total	
1	Gifts, grants, contributions, and					1		
	membership fees received. (Do not				l			
	include any "unusual grants.")							
2	Gross receipts from admissions,							
	merchandise sold or services per-							
	formed, or facilities furnished in any activity that is related to the					{		
	organization's tax-exempt purpose	i						
3	Gross receipts from activities that							
	are not an unrelated trade or bus-	•		1				
	iness under section 513							
4	Tax revenues levied for the organ-		1	1				
•	ization's benefit and either paid to							
	or expended on its behalf							
5	The value of services or facilities		 	 				
J	furnished by a governmental unit to							
	the organization without charge					j		
6	Total. Add lines 1 through 5		 					
	Amounts included on lines 1, 2, and					 		
,,	3 received from disqualified persons		Ì					
	Amounts included on lines 2 and 3 received		-	 	-			
-	from other than disqualified persons that		1					
	exceed the greater of \$5,000 or 1% of the		1					
	amount on line 13 for the year	<u> </u>						
	Add lines 7a and 7b	- in	e cui e escrib					
	Public support. (Subtract line 7c from line 6.)	<u> </u>	3,27		<u> </u>			
	ction B. Total Support		1				 	
	ndar year (or fiscal year beginning in)	(a) 2015	(b) 2016	(c) 2017	(d) 2018	(e) 2019	(f) Totai	
	Amounts from line 6		 					
าบล	Gross income from interest, dividends, payments received on							
	securities loans, rents, royalties.							
	and income from similar sources			ļ	ļ. -			
t	Unrelated business taxable income			1				
	(less section 511 taxes) from businesses						1	
	acquired after June 30, 1975]				
	Add lines 10a and 10b		ļ					
11	Net income from unrelated business activities not included in line 10b,]				
	whether or not the business is						İ	
	regularly carried on							
12	Other income. Do not include gain	ļ						
	or loss from the sale of capital assets (Explain in Part VI.)							
13	Total support. (Add lines 9, 10c, 11, and 12.)							
14	First five years. If the Form 990 is fo	r the organization's	s first, second, thir	d, fourth, or fifth ta	ix year as a sectio	n 501(c)(3) organiza	ition,	
	check this box and stop here							
Se	ction C. Computation of Publi							
15	Public support percentage for 2019 (ine 8, column (f), d	livided by line 13,	column (f))		15	%	
16	Public support percentage from 2018					16	%	
Sec	ction D. Computation of Inves	stment Income	e Percentage					
17	Investment income percentage for 20)19 (line 10c, colu	mn (f), divided by li	ine 13, column (f))		17	%	
18	Investment income percentage from	2018 Schedule A,	Part III, line 17			18	%	
198	33 1/3% support tests - 2019. If the	organization did r				33 1/3%, and line 17	is not	
	more than 33 1/3%, check this box as						▶ □	
Ŀ	33 1/3% support tests - 2018. If the	-	-				••	
	line 18 is not more than 33 1/3%, che	=						
20			-					
	20 Private foundation. If the organization did not check a box on line 14, 19a, or 19b, check this box and see instructions							

Vas No

Part IV | Supporting Organizations

(Complete only if you checked a box in line 12 on Part I. If you checked 12a of Part I, complete Sections A and B. If you checked 12b of Part I, complete Sections A and C. If you checked 12c of Part I, complete Sections A, D, and E. If you checked 12d of Part I, complete Sections A and D, and complete Part V.)

Section A. All Supporting Organizations

- 1 Are all of the organization's supported organizations listed by name in the organization's governing documents? If "No," describe in Part VI how the supported organizations are designated. If designated by class or purpose, describe the designation. If historic and continuing relationship, explain.
- 2 Did the organization have any supported organization that does not have an IRS determination of status under section 509(a)(1) or (2)? If "Yes," explain in Part VI how the organization determined that the supported organization was described in section 509(a)(1) or (2).
- 3a Did the organization have a supported organization described in section 501(c)(4), (5), or (6)? If "Yes," answer (b) and (c) below.
- b Did the organization confirm that each supported organization qualified under section 501(c)(4), (5), or (6) and satisfied the public support tests under section 509(a)(2)? If "Yes," describe in Part VI when and how the organization made the determination.
- c Did the organization ensure that all support to such organizations was used exclusively for section 170(c)(2)(B) purposes? If "Yes," explain in Part VI what controls the organization put in place to ensure such use.
- 4a Was any supported organization not organized in the United States ("foreign supported organization")? If "Yes," and if you checked 12a or 12b in Part I, answer (b) and (c) below.
- b Did the organization have ultimate control and discretion in deciding whether to make grants to the foreign supported organization? If "Yes," describe in Part VI how the organization had such control and discretion despite being controlled or supervised by or in connection with its supported organizations.
- c Did the organization support any foreign supported organization that does not have an IRS determination under sections 501(c)(3) and 509(a)(1) or (2)? If "Yes," explain in Part VI what controls the organization used to ensure that all support to the foreign supported organization was used exclusively for section 170(c)(2)(B) purposes.
- 5a Did the organization add, substitute, or remove any supported organizations during the tax year? If "Yes," answer (b) and (c) below (if applicable). Also, provide detail in Part VI, including (i) the names and EIN numbers of the supported organizations added, substituted, or removed; (ii) the reasons for each such action; (iii) the authority under the organization's organizing document authorizing such action; and (iv) how the action was accomplished (such as by amendment to the organizing document).
- **b** Type I or Type II only. Was any added or substituted supported organization part of a class already designated in the organization's organizing document?
- c Substitutions only. Was the substitution the result of an event beyond the organization's control?
- 6 Did the organization provide support (whether in the form of grants or the provision of services or facilities) to anyone other than (i) its supported organizations, (ii) individuals that are part of the charitable class benefited by one or more of its supported organizations, or (iii) other supporting organizations that also support or benefit one or more of the filing organization's supported organizations? If "Yes," provide detail in Part VI.
- 7 Did the organization provide a grant, loan, compensation, or other similar payment to a substantial contributor (as defined in section 4958(c)(3)(C)), a family member of a substantial contributor, or a 35% controlled entity with regard to a substantial contributor? If "Yes," complete Part I of Schedule L (Form 990 or 990-EZ).
- 8 Did the organization make a loan to a disqualified person (as defined in section 4958) not described in line 7? If "Yes," complete Part I of Schedule L (Form 990 or 990-EZ).
- 9a Was the organization controlled directly or indirectly at any time during the tax year by one or more disqualified persons as defined in section 4946 (other than foundation managers and organizations described in section 509(a)(1) or (2))? If "Yes," provide detail in Part VI.
- b Did one or more disqualified persons (as defined in line 9a) hold a controlling interest in any entity in which the supporting organization had an interest? If "Yes," provide detail in Part VI.
- c Did a disqualified person (as defined in line 9a) have an ownership interest in, or derive any personal benefit from, assets in which the supporting organization also had an interest? If "Yes," provide detail in Part VI.
- 10a Was the organization subject to the excess business holdings rules of section 4943 because of section 4943(f) (regarding certain Type II supporting organizations, and all Type III non-functionally integrated supporting organizations)? If "Yes," answer 10b below.
 - b Did the organization have any excess business holdings in the tax year? (Use Schedule C, Form 4720, to determine whether the organization had excess business holdings.)

		· · · · · · · · · · · · · · · · · · ·
1		
	judaj.	it in it
i		
2		
, ,		
34040m.mm		<i>\$60.00</i>
3a		
<u> </u>		- 1
1		1
i l		1
3b		

3c		
	.725.13	3.5
		1
4		I
4a	. ~	
1		L I
[[
J		<u> </u>
4b	l	
	 	— ;
	4.3	L I
i '		lv l
[. :		i i i i i i i i i i i i i i i i i i i
	i hadiş	7
************	·	
4c		
		**. * - 1
1		
1	9	.
1		1
1		· 1
-		
5a		
	r La estaĝ	45.13.7
	ئىنىدى <u>نىن</u> ىن	Section Section
5b	د در	Salarine Salarine
	Y 1133	
5b 5c		Collina Spiriton
		Control Statistics
		-
		-
		-
		-
		-
5c		-
5c		
5c 6 7		
5c		
6 7		
6 7		
6 7		
6 		
6 7 8		
6 7 8		
6 - 7 - 8		
6 7 8		
6 - 7 - 8		
6 - 7 - 8		
6 - 7 - 8 - 9a - 9b - 9b		
6 - 7 - 8		
6 - 7 - 8 - 9a - 9b - 9b		
6 - 7 - 8 - 9a - 9b - 9b		
6 - 7 - 8 - 9a - 9b - 9b		
6 6 7 8 9a 9b 9c		
6 6 7 8 9a 9b 9c		
6 6 7 8 9a 9b 9c		
6 6 7 8 9a 9b 9c		

Pai	TTIV Supporting Organizations (continued)			
			Yes	No
11	Has the organization accepted a gift or contribution from any of the following persons?		1	
а	A person who directly or indirectly controls, either alone or together with persons described in (b) and (c)			ar vā
	below, the governing body of a supported organization?	11a		
b	A family member of a person described in (a) above?	11b		
c	A 35% controlled entity of a person described in (a) or (b) above? If "Yes" to a, b, or c, provide detail in Part VI.	11c		
Sec	tion B. Type I Supporting Organizations			
			Yes	No
1	Did the directors, trustees, or membership of one or more supported organizations have the power to		-	,
	regularly appoint or elect at least a majority of the organization's directors or trustees at all times during the			
	tax year? If "No," describe in Part VI how the supported organization(s) effectively operated, supervised, or			
	controlled the organization's activities. If the organization had more than one supported organization,			*
	describe how the powers to appoint and/or remove directors or trustees were allocated among the supported			
	organizations and what conditions or restrictions, if any, applied to such powers during the tax year.	1	· -	
2	Did the organization operate for the benefit of any supported organization other than the supported	 -	 	
4			,	1
	organization(s) that operated, supervised, or controlled the supporting organization? If "Yes," explain in			
	Part VI how providing such benefit carried out the purposes of the supported organization(s) that operated,		·	
Sec	supervised, or controlled the supporting organization. tion C. Type II Supporting Organizations	2	<u> </u>	<u> </u>
	and at 12ho it authorities at structure in	 .	\ <u>v</u>	NI
,	Mary a majority of the grandinations dispersive as to late a dispersive the territories of the dispersive		Yes	No
1	Were a majority of the organization's directors or trustees during the tax year also a majority of the directors			
	or trustees of each of the organization's supported organization(s)? If "No," describe in Part VI how control	ŀ	l .	
	or management of the supporting organization was vested in the same persons that controlled or managed		·	 -
500	the supported organization(s). tion D. All Type III Supporting Organizations	_1_	L	Ь
Sec	don b. All Type III Supporting Organizations		1,,	
	Did the constitution and the control of the constitution of the control of the co		Yes	No
1	Did the organization provide to each of its supported organizations, by the last day of the fifth month of the			
	organization's tax year, (i) a written notice describing the type and amount of support provided during the prior tax			
	year, (ii) a copy of the Form 990 that was most recently filed as of the date of notification, and (iii) copies of the	<u> </u>		
	organization's governing documents in effect on the date of notification, to the extent not previously provided?	1		
2	Were any of the organization's officers, directors, or trustees either (i) appointed or elected by the supported		1	
	organization(s) or (ii) serving on the governing body of a supported organization? If "No," explain in Part VI how			
	the organization maintained a close and continuous working relationship with the supported organization(s).	2		
3	By reason of the relationship described in (2), did the organization's supported organizations have a]		
	significant voice in the organization's investment policies and in directing the use of the organization's	j		
	income or assets at all times during the tax year? If "Yes," describe in Part VI the role the organization's	<u></u> -		
	supported organizations played in this regard.	3	l	L
	tion E. Type III Functionally Integrated Supporting Organizations			
1	Check the box next to the method that the organization used to satisfy the Integral Part Test during the year (see instructions)			
a	The organization satisfied the Activities Test. Complete line 2 below.			
b	The organization is the parent of each of its supported organizations. Complete line 3 below.			
C	The organization supported a governmental entity. Describe in Part VI how you supported a government entity (see inst	ructions		T
2	Activities Test. Answer (a) and (b) below.		Yes	No
а			1	
	the supported organization(s) to which the organization was responsive? If "Yes," then in Part VI identify	!		
	those supported organizations and explain how these activities directly furthered their exempt purposes,		:	
	how the organization was responsive to those supported organizations, and how the organization determined			-
	that these activities constituted substantially all of its activities.	2a	1000	160
þ	Did the activities described in (a) constitute activities that, but for the organization's involvement, one or more		\$24.3	
	of the organization's supported organization(s) would have been engaged in? If "Yes," explain in Part VI the]		
	reasons for the organization's position that its supported organization(s) would have engaged in these		.[
	activities but for the organization's involvement.	2b		-
3	Parent of Supported Organizations. Answer (a) and (b) below.			ľ.
а			·	ļ
	trustees of each of the supported organizations? Provide details in Part VI.	3a		<u> </u>
b				
	of its supported organizations? If "Yes." describe in Part VI the role played by the organization in this regard.	3b		<u> </u>

Par	t V Type III Non-Functionally Integrated 509(a)(3) Supporting	g Orga	nizations	-	
1	Check here if the organization satisfied the Integral Part Test as a qualifying	g trust o	n Nov. 20, 1970 (e	explain in Part	VI). See instructions. Al
	other Type III non-functionally integrated supporting organizations must co	mplete S	Sections A through	h E.	
Secti	ion A - Adjusted Net Income		(A) Prior Year		(B) Current Year (optional)
1	Net short-term capital gain	1			
_2	Recoveries of prior-year distributions	2			
3	Other gross income (see instructions)	3			
4	Add lines 1 through 3.	4			
_5	Depreciation and depletion	5			
6	Portion of operating expenses paid or incurred for production or			-	
	collection of gross income or for management, conservation, or				
	maintenance of property held for production of income (see instructions)	6			
_7	Other expenses (see instructions)	7			
8	Adjusted Net Income (subtract lines 5, 6, and 7 from line 4)	8	T		
Secti	ion B - Minimum Asset Amount	·	(A) Prior	/ear	(B) Current Year (optional)
1	Aggregate fair market value of all non-exempt-use assets (see			.24 ¢	
	instructions for short tax year or assets held for part of year):	n sing			
а	Average monthly value of securities	1a	T		
	Average monthly cash balances	1b			
	Fair market value of other non-exempt-use assets	1c			•
	Total (add lines 1a, 1b, and 1c)	1d			
	Discount claimed for blockage or other	1 -	•		
•	factors (explain in detail in Part VI):	į.			
2	Acquisition indebtedness applicable to non-exempt-use assets	2			***************************************
3	Subtract line 2 from line 1d.	3		- I	
4	Cash deemed held for exempt use. Enter 1-1/2% of line 3 (for greater amount,	Ť			
•	see instructions).	4			
5	Net value of non-exempt-use assets (subtract line 4 from line 3)	5			
6	Multiply line 5 by .035.	6			
7	Recoveries of prior-year distributions	7			-
8	Minimum Asset Amount (add line 7 to line 6)	8			
	ion C - Distributable Amount	, 0	1		Current Year
1	Adjusted net income for prior year (from Section A, line 8, Column A)	1			
2	Enter 85% of line 1.	2	F		•
3	Minimum asset amount for prior year (from Section B, line 8, Column A)	3			
4	Enter greater of line 2 or line 3.	4	:	2	
5	Income tax imposed in prior year	5		\$	
6	Distributable Amount. Subtract line 5 from line 4, unless subject to	<u>ا آ</u>	1	7. 350 mm	
•	emergency temporary reduction (see instructions).	6	a ar		
7	Check here if the current year is the organization's first as a non-functional		ated Type III supp		ation (see
•	instructions).	.,	, p vapp		

Schedule A (Form 990 or 990-EZ) 2019

Par	t V Type III Non-Functionally Integrated 509	(a)(3) Suppo	rting Orga	nizations	(continued)		
Secti	on D - Distributions					Cur	rent Year
1	Amounts paid to supported organizations to accomplish exe	mpt purposes					
2	Amounts paid to perform activity that directly furthers exemp	ot purposes of s	supported				
	organizations, in excess of income from activity						,
3	Administrative expenses paid to accomplish exempt purpose	es of supported	l organizations				
4	Amounts paid to acquire exempt-use assets						
5	Qualified set-aside amounts (prior IRS approval required)						
6	Other distributions (describe in Part VI). See instructions.						
7	Total annual distributions. Add lines 1 through 6.			-			
8	Distributions to attentive supported organizations to which ti	he organization	is responsive				
	(provide details in Part VI). See instructions.				_		
9	Distributable amount for 2019 from Section C, line 6						
10	Line 8 amount divided by line 9 amount						
Secti	on E - Distribution Allocations (see instructions)	(i Excess Dis		Underdi	(ii) stributions -2019		(iii) tributable unt for 2019
1	Distributable amount for 2019 from Section C, line 6	61.			, ķiri		_
2	Underdistributions, if any, for years prior to 2019 (reason-	and the second s	1-467 26 3-1-54			* :	
	able cause required- explain in Part VI). See instructions.						
3	Excess distributions carryover, if any, to 2019			Maria di T			KIN .
а	From 2014	1 Table - 1	. 46	* *	\$ \$. ** 		e Partir de la companya de la compa
b	From 2015	E _a E _a			: :	*	·
С	From 2016	2		·	· #	<i>;</i>	
d	From 2017	3 200	- :		<u>.</u>		
	From 2018	A 15.	* 15°				H 9 1, 1 2
	Total of lines 3a through e				ine dell'i		E 9
	Applied to underdistributions of prior years	Mark .	· Fine			ing . With .	
	Applied to 2019 distributable amount			,	, s		
<u>; .</u>	Carryover from 2014 not applied (see instructions)						1940
i	Remainder. Subtract lines 3g, 3h, and 3i from 3f.						
4	Distributions for 2019 from Section D.		- · · · · ·	^		-	
•	line 7: \$		ä.		ga Sara		
a	Applied to underdistributions of prior years				V 65(07) 5607	*	11.00
	Applied to 2019 distributable amount	***	# 65 P 13			,~ ·	
	Remainder, Subtract lines 4a and 4b from 4.		_	. ***	Y- 85 C	2	
	Remaining underdistributions for years prior to 2019, if	**************************************	F . 1 8 %			124	
-	any. Subtract lines 3g and 4a from line 2. For result greater					H	* a*
	than zero, explain in Part VI. See instructions.		مُ الله				, ki
6	Remaining underdistributions for 2019. Subtract lines 3h		٠		. 2002 2000		
	and 4b from line 1. For result greater than zero, explain in	:		c ¥	· -= 4 · · · · ·		
	Part VI. See instructions.			P rop			
7	Excess distributions carryover to 2020. Add lines 3		- 211 313141 141 24			S	4 2 4 4 4
-	and 4c.			Andi'' '		10 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
8	Breakdown of line 7:	V	* (/)		1	4.7	
	Excess from 2015			-			279.000
	Excess from 2016				* * *		
	Excess from 2017	1.5			٠,٤.		
	Excess from 2018	7. T	A\$7.0. A\$7.00A	(** · · · · · · · · · · · · · · · · · ·			
	Excess from 2010	Part of the second					

Schedule A (Form 990 or 990-EZ) 2019

Schedule B

(Form 990, 990-EZ, or 990-PF)

Department of the Treasury Internal Revenue Service

Name of the organization

Schedule of Contributors

Attach to Form 990, Form 990-EZ, or Form 990-PF. ► Go to www.irs.gov/Form990 for the latest information. OMB No. 1545-0047

2019

Employer identification number

НО	MES WITH HOPE, INC.	22-2534326
Organization type (check or	ne):	
Filers of:	Section:	
Form 990 or 990-EZ	X 501(c)(3) (enter number) organization	
	4947(a)(1) nonexempt charitable trust not treated as a private found	ation
	527 political organization	
Form 990-PF	501(c)(3) exempt private foundation	
	4947(a)(1) nonexempt charitable trust treated as a private foundation	
	501(c)(3) taxable private foundation	-
	s covered by the General Rule or a Special Rule. (7), (8), or (10) organization can check boxes for both the General Rule and a	Special Rule. See instructions.
General Rule		
-	n filing Form 990, 990-EZ, or 990-PF that received, during the year, contributions one contributor. Complete Parts I and II. See instructions for determining a contributor.	
Special Rules		
sections 509(a)(1) a any one contributo	n described in section 501(c)(3) filing Form 990 or 990-EZ that met the 33 1/3 and 170(b)(1)(A)(vi), that checked Schedule A (Form 990 or 990-EZ), Part II, lir or, during the year, total contributions of the greater of (1) \$5,000; or (2) 2% or line 1. Complete Parts I and II.	ne 13, 16a, or 16b, and that received from
year, total contribu	n described in section 501(c)(7), (8), or (10) filing Form 990 or 990-EZ that receitions of more than \$1,000 exclusively for religious, charitable, scientific, literatly to children or animals. Complete Parts I, II, and III.	
year, contributions is checked, enter h purpose. Don't cor	n described in section 501(c)(7), (8), or (10) filing Form 990 or 990-EZ that recommend the exclusively for religious, charitable, etc., purposes, but no such contribution here the total contributions that were received during the year for an exclusive mplete any of the parts unless the General Rule applies to this organization e, etc., contributions totaling \$5,000 or more during the year	s totaled more than \$1,000. If this box rely religious, charitable, etc., because it received <i>nonexclusively</i>
but it must answer "No" on	nat isn't covered by the General Rule and/or the Special Rules doesn't file Sch Part IV, line 2, of its Form 990; or check the box on line H of its Form 990-EZ he filing requirements of Schedule B (Form 990, 990-EZ, or 990-PF).	
LHA For Paperwork Reduction	on Act Notice, see the instructions for Form 990, 990-EZ, or 990-PF.	Schedule B (Form 990, 990-EZ, or 990-PF) (2019)

Name of organization

Employer identification number

HOMES WITH HOPE, INC

22-2534326

Part I	Contributors (see instructions). Use duplicate copies of Part I if additional	I space is needed.	
(a) No.	(b) Name, address, and ZIP + 4	(c) Total contributions	(d) Type of contribution
1_	CT DEPT. OF HOUSING 25 SIGOURNEY STREET HARTFORD, CT 06106	\$ <u>159,423.</u>	Person X Payroll
(a) No.	(b) Name, address, and ZIP + 4	(c) Total contributions	(d) Type of contribution
2	CT DEPT. OF MENTAL HEALTH & ADDICTION SERVICES 410 CAPITOL AVE, PO BOX 341431 HARTFORD, CT 06134	\$342,921.	Person X Payroll
(a) No.	(b) Name, address, and ZIP + 4	(c) Total contributions	(d) Type of contribution
3	SUPPORTIVE HOUSING WORKS 387 CLINTON AVENUE BRIDGEPORT, CT 06605	\$ <u>155,268.</u>	Person X Payroll Noncash (Complete Part II for noncash contributions.)
(a) No.	(b) Name, address, and ZIP + 4	(c) Total contributions	(d) Type of contribution
4_	ESTATE OF DONALD T. HEGEDUS C/O HOMES WITH HOPE, INC., 49 RICHMONDVILLE AVENUE, SUITE 112 WESTPORT, CT 06880	\$ <u>197,700.</u>	Person X Payroll Noncash (Complete Part II for noncash contributions.)
(a) No.	(b) Name, address, and ZIP + 4	(c) Total contributions	(d) Type of contribution
		\$	Person Payroll Oncash (Complete Part II for noncash contributions.)
(a) No.	(b) Name, address, and ZIP + 4	(c) Total contributions	(d) Type of contribution
		\$	Person Payroll Oncash Occash (Complete Part II for noncash contributions.)

Name of organization

Employer identification number

HOMES WITH HOPE, INC.

22-2534326

art II	Noncash Property (see instructions). Use duplicate copies of Par	rt II if additional space is needed.	
(a) No. from Part I	(b) Description of noncash property given	(c) FMV (or estimate) (See instructions.)	(d) Date received
		 \$	
(a) No. rom	(b) Description of noncash property given	(c) FMV (or estimate) (See instructions.)	(d) Date received
		\$	*
(a) No. rom	(b) Description of noncash property given	(c) FMV (or estimate) (See instructions.)	(d) Date received
		<u></u>	
(a) No. from Part I	(b) Description of noncash property given	(c) FMV (or estimate) (See instructions.)	(d) Date received
 		\$	
(a) No. from Part I	(b) Description of noncash property given	(c) FMV (or estimate) (See instructions.)	(d) Date received
 -			
(a) No. from Part I	(b) Description of noncash property given	(c) FMV (or estimate) (See instructions.)	(d) . Date received

Schedule B (Form 990, 990-EZ, or 990-PF) (2019) **Employer identification number** Name of organization HOMES WITH HOPE, 22-2534326 Exclusively religious, charitable, etc., contributions to organizations described in section 501(c)(7), (8), or (10) that total more than \$1,000 for the year from any one contributor. Complete columns (a) through (e) and the following line entry. For organizations completing Part III, enter the total of exclusively religious, charitable, etc., contributions of \$1,000 or less for the year. (Enter this info. once.) Use duplicate copies of Part III if additional space is needed. (a) No. from Part I (c) Use of gift (d) Description of how gift is held (b) Purpose of gift (e) Transfer of gift Transferee's name, address, and ZIP + 4 Relationship of transferor to transferee (a) No. from Part I (c) Use of gift (d) Description of how gift is held (b) Purpose of gift (e) Transfer of gift Transferee's name, address, and ZIP + 4 Relationship of transferor to transferee (a) No. (d) Description of how gift is held from (c) Use of gift (b) Purpose of gift Part I (e) Transfer of gift Transferee's name, address, and ZIP + 4 Relationship of transferor to transferee (a) No. from (b) Purpose of gift (d) Description of how gift is held (c) Use of gift Part_ (e) Transfer of gift Transferee's name, address, and ZIP + 4 Relationship of transferor to transferee

923454 11-06-19

Schedule B (Form 990, 990-EZ, or 990-PF) (2019)

SCHEDULE D

(Form 990)

Department of the Treasury Internal Revenue Service

Supplemental Financial Statements

► Complete if the organization answered "Yes" on Form 990, Part IV, line 6, 7, 8, 9, 10, 11a, 11b, 11c, 11d, 11e, 11f, 12a, or 12b. ► Attach to Form 990. ► Go to www.irs.gov/Form990 for instructions and the latest information.

OMB No. 1545-0047 Inspection

Name of the organization

HOMES WITH HOPE TNC **Employer identification number** 22-2534326

Par	Organizations Maintaining Donor Advised	Funds or Other Similar Funds	or Accounts. Complete if the
\$1.10Ac-1177	organization answered "Yes" on Form 990, Part IV, line		Complete in the
		(a) Donor advised funds	(b) Funds and other accounts
1	Total number at end of year		
2	Aggregate value of contributions to (during year)		
3	Aggregate value of grants from (during year)		<u> </u>
4	Aggregate value at end of year		
5	Did the organization inform all donors and donor advisors in v	vriting that the assets held in donor advis-	ed funds
	are the organization's property, subject to the organization's e	exclusive legal control?	Yes No
6	Did the organization inform all grantees, donors, and donor ad	dvisors in writing that grant funds can be	used only
	for charitable purposes and not for the benefit of the donor or	r donor advisor, or for any other purpose of	conferring
			
Par	Conservation Easements. Complete if the org	ganization answered "Yes" on Form 990, I	Part IV, line 7.
1	Purpose(s) of conservation easements held by the organization	on (check all that appl <u>y).</u>	
	Preservation of land for public use (for example, recreat	tion or education) Preservation of	a historically important land area
	Protection of natural habitat	Preservation of	a certified historic structure
	Preservation of open space		•
2	Complete lines 2a through 2d if the organization held a qualifi	ied conservation contribution in the form	of a conservation easement on the last
	day of the tax year.		Held at the End of the Tax Year
а	Total number of conservation easements		2a
Ь	Total acreage restricted by conservation easements		2b
C	Number of conservation easements on a certified historic stru	ucture included in (a)	2c -
d	Number of conservation easements included in (c) acquired a	ifter 7/25/06, and not on a historic structu	ıre
	listed in the National Register		
3	Number of conservation easements modified, transferred, rele	eased, extinguished, or terminated by the	organization during the tax
	year ▶		
4	Number of states where property subject to conservation eas	· · · · · · · · · · · · · · · · · · ·	
5	Does the organization have a written policy regarding the per		<u></u>
	violations, and enforcement of the conservation easements it		
6	Staff and volunteer hours devoted to monitoring, inspecting,	handling of violations, and enforcing cons	servation easements during the year
_	• — — — — — — — — — — — — — — — — — — —		v.
7	Amount of expenses incurred in monitoring, inspecting, hand	ling of violations, and enforcing conserva-	tion easements during the year
•	Described an line Old show		L-\/4\/\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\
8	Does each conservation easement reported on line 2(d) above		
9	and section 170(h)(4)(B)(ii)? In Part XIII, describe how the organization reports conservation		
9	balance sheet, and include, if applicable, the text of the footn		
	organization's accounting for conservation easements.	iote to the organization's infancial statem	ents that describes the
Pai	Till Organizations Maintaining Collections of	Art. Historical Treasures, or Ot	her Similar Assets.
2.00	Complete if the organization answered "Yes" on Form		
	If the organization elected, as permitted under FASB ASC 95		and balance sheet works
	of art, historical treasures, or other similar assets held for pub	•	
	service, provide in Part XIII the text of the footnote to its finar	· · · · · · · · · · · · · · · · · · ·	,
b	If the organization elected, as permitted under FASB ASC 95		
_	art, historical treasures, or other similar assets held for public		
	provide the following amounts relating to these items:	or indicate in the second of t	totalion of positions,
	(i) Revenue included on Form 990, Part VIII, line 1		> \$
	(ii) Assets included in Form 990, Part X		
2	If the organization received or held works of art, historical treat		
4	the following amounts required to be reported under FASB A		3
,	Revenue included on Form 990, Part VIII, line 1	-	> \$
	Assets included in Form 990, Part X		
	For Paperwork Reduction Act Notice, see the Instructions		Schedule D (Form 990) 2019

932051 10-02-19

Schedule D (Form 990) 2019

2,501,827.

15,313.

e Other.

Total. Add lines 1a through 1e. (Column (d) must equal Form 990. Part X. column (B). line 10c.)

Λ.

15,313.

Ð	(Form 9	90) 20	219	H	OMES	WITH	HOPE,	INC
	-				_	844		

Complete if the organization answered "Yes"			
(a) Description of security or category (including name of security)	(b) Book value	(c) Method of valuation: Cost or end	-of-year market value
1) Financial derivatives			
2) Closely held equity interests			
3) Other		·	
(A)		<u> </u>	
(B)			
(C)			• _
(D)			·
(E)		-	
(F)			
(G)			
(H)			
otal. (Col. (b) must equal Form 990, Part X, col. (B) line 12.)		The Committee of the Co	
Part VIII Investments - Program Related.	<u> </u>		
	F 000 D+ N/ II-	- 44 - O Farm 000 Bart V Bar 40	4
Complete if the organization answered "Yes" (a) Description of investment	(b) Book value	(c) Method of valuation: Cost or end	of your market value
	(B) BOOK VAIUE	(c) Method of Valdation. Cost of end	-or-year market value
	 	- 	
(2)		<u> </u>	
(3)	ļ		
(4)			
(5)			<u></u>
(6)			
(7)			· ·
(8)			
(9)			*
Total. (Col. (b) must equal Form 990, Part X, col. (B) line 13.)		and the said has the said the said to the	
Part IX Other Assets.			<u></u>
Complete if the organization answered "Yes"	on Form 990, Part IV, line	e 11d. See Form 990, Part X, line 15.	
	Description		(b) Book value
(1) DUE FROM AFFILIATE		· · · · · · · · · · · · · · · · · · ·	141,555.
(2) CHFA RESERVES			273,354.
(3)			
(4)			-
(5)			-
	-	****	· — · · · · · · · · · · · · · · · · · ·
(6)			
	··-	·	
(8)			
(9)			414,909.
Total. (Column (b) must equal Form 990. Part X. col. (B) lir Part X Other Liabilities.	<u>e 15.)</u>	······································	414,303
Complete if the organization answered "Yes"	on Form 990 Part IV lin	e 11e or 11f See Form 990 Part X line 25	
4 (a) Description of liability	Offi Chin 550, Fare IV, Ital	5 (16 OF 111, COS (SKII 550, 1 EECX, MIS 25,	(b) Book value
	<u> </u>		1-1
(1) Federal income taxes (2) SECURITY DEPOSITS			10,571.
			10,011,
(3)			
	•		
(5)			
(6)			
(7)			
(8)			
(9)			
Total. (Column (b) must equal Form 990. Part X. col. (B) lin	ne 25.)		10,571.
2. Liability for uncertain tax positions. In Part XIII, provid		to the organization's financial statements to	hat reports the
organization's liability for uncertain tax positions unde			
- ga andress o manney for an oor sum that positions direct			edule D (Form 990) 20

Sche	dule D (Form 990) 2019 HOMES WITH HOPE, INC.		. <u>.</u> .	22-2	534326 Page 4
Par	Reconciliation of Revenue per Audited Financial Statemen	ts With	Revenue per Re	turn.	
	Complete if the organization answered "Yes" on Form 990, Part IV, line 12a.				
1	Total revenue, gains, and other support per audited financial statements			1	3,851 <u>,605</u> .
2	Amounts included on line 1 but not on Form 990, Part VIII, line 12:			2.33	
а	Net unrealized gains (losses) on investments	2a	<u>-3,478.</u>		
b	Donated services and use of facilities	2b	672,823.		
С		2c		Chr.	
d	Other (Describe in Part XIII.)	2d	135,478.	(1)	
e	Add lines 2a through 2d			2e	804,823.
3	Subtract line 2e from line 1			3	3,046,782.
4	Amounts included on Form 990, Part VIII, line 12, but not on line 1:				
а	Investment expenses not included on Form 990, Part VIII, line 7b	4a		15.0	
b	Other (Describe in Part XIII.)		-164,230.		
С	Add lines 4a and 4b			4c	-164,230.
5	Total revenue, Add lines 3 and 4c. (This must equal Form 990, Part I, line 12.)			5	2,882,552.
Pa	T XII Reconciliation of Expenses per Audited Financial Statemen	nts Wit	h Expenses per F	Return	l.
.,	Complete if the organization answered "Yes" on Form 990, Part IV, line 12a.				
1	Total expenses and losses per audited financial statements			1	3,875,871.
2	Amounts included on line 1 but not on Form 990, Part IX, line 25:				
a	Donated services and use of facilities	2a	672,823.	200.04	-
b	— • • • • • •	2b	<u> </u>	1	
c		_		1	4.5
d			358,033.		
	Add lines 2a through 2d		,	2e	1,030,856
_				3	2,845,015
3	Subtract line 2e from line 1			3	2,013,013.
4	Amounts included on Form 990, Part IX, line 25, but not on line 1:	المما		9201-3	
	Investment expenses not included on Form 990, Part VIII, line 7b			1 1	
	Other (Describe in Part XIII.)				0.
_	Add lines 4a and 4b			4c	2,845,015
5	Total expenses. Add lines 3 and 4c. (This must equal Form 990. Part I, line 18.)			5	2,045,015
	ide the descriptions required for Part II, lines 3, 5, and 9; Part III, lines 1a and 4; Part IV			i; Part X	i, line 2; Part XI,
lines	2d and 4b; and Part XII, lines 2d and 4b. Also complete this part to provide any additi	ional info	rmation.		
					-
PA	RT IV, LINE 2B:				
					GGGTTTTG
HOI	MES WITH HOPE HOLD FUNDS OF SHELTER CLIENTS	'I'HA'I	HAVE NO BA	NK A	ACCOUNTS.
<u>PA</u>	RT V, LINE 4:				
					_
TO	PROVIDE A CONTINGENCY FOR UNEXPECTED OPERA	TION	T DEAETOBWE	NTS	AND FOR
PO:	SSIBLE FUTURE INCREASES IN BENEFIT STRUCTUR	<u>e ani</u>	D EXPENSE LE	VELS	S AS WELL
<u>AS</u>	UNEXPECTED DOWNTURNS IN FUTURE LEVELS OF A	NNUAI	L CONTRIBUTI	ONS	AND
<u>FU</u>	NDING. TO BUILD AN INVESTMENT RESERVE FOR	FUTUI	RE UNFORESEE	N P	ROJECTS
TH	AT WOULD REQUIRE A CAPITAL CONTRIBUTION.				
					-

PART X, LINE 2:

BELOW IS AN EXCERPT FROM FOOTNOTE 1 OF THE AUDITED FINANCIAL STATEMENTS

FOR HOMES WITH HOPE, INC. AND AFFILIATE.

HWH AND IHA WERE INCORPORATED UNDER THE LAWS OF THE STATE OF CONNECTICUT

AND SERVE THE PUBLIC WITHIN THE MEANING OF CHARITABLE, EDUCATIONAL

ORGANIZATIONS AS DEFINED BY SECTION 501(C)(3) OF THE INTERNAL REVENUE CODE

(THE CODE). AS SUCH HWH AND IHA ARE EXEMPT FROM STATE AND FEDERAL INCOME

TAXES PURSUANT TO SECTION 509(A) OF THE CODE. ACCOUNTING PRINCIPLES

GENERALLY ACCEPTED IN THE UNITED STATES OF AMERICA REQUIRE MANAGEMENT TO

EVALUATE TAX POSITIONS TAKEN AND RECOGNIZE A TAX LIABILITY (OR ASSET) IF

THE ORGANIZATION HAS TAKEN AN UNCERTAIN POSITION THAT MORE LIKELY THAN NOT

WOULD NOT BE SUSTAINED UPON EXAMINATION BY TAXING AUTHORITIES. MANAGEMENT

HAS ANALYZED THE TAX POSITIONS TAKEN AND HAS CONCLUDED THAT AS OF DECEMBER

31, 2019, THERE ARE NO UNCERTAIN POSITIONS TAKEN OR EXPECTED TO BE TAKEN

THAT WOULD REQUIRE RECOGNITION OF A LIABILITY (OR ASSET) OR DISCLOSURE IN

HWH AND IHA ARE SUBJECT TO ROUTINE AUDITS BY TAXING JURISDICTIONS;

HOWEVER, THERE ARE CURRENTLY NO AUDITS FOR ANY TAX PERIODS PENDING OR IN PROGRESS.

PART XI, LINE 2D - OTHER ADJUSTMENTS:

INCOME FROM CONSOLIDATED COMPANY 135,478.

PART XI, LINE 4B - OTHER ADJUSTMENTS:

DIRECT EXPENSES FROM FUNDRAISING EVENTS -164,230.

PART XII, LINE 2D - OTHER ADJUSTMENTS:

EXPENSES OF CONSOLIDATED COMPANY

193,803.

Schedule D (Form 990) 2019

Schedule D (Form 990) 2019 HOMES WITH HOPE, INC.	22-253432 <u>6 Page 5</u>
Schedule D (Form 990) 2019 HOMES WITH HOPE, INC. Part XIII Supplemental Information (continued)	
DIRECT EXPENSES FROM FUNDRAISING EVENTS	164,230.
MOMAI MO COURDITE D. DADM VII IING 2D	358,033.
TOTAL TO SCHEDULE D, PART XII, LINE 2D	330,033.
· · · · · · · · · · · · · · · · · · ·	
<u> </u>	
	·
	-
 	
	
*	
	
	-

Schedule D (Form 990) 2019

SCHEDULE G

Department of the Treasury

Internal Revenue Service

(Form:990 or 990-EZ)

Supplemental Information Regarding Fundraising or Gaming Activities

Complete if the organization answered "Yes" on Form 990, Part IV, line 17, 18, or 19, or if the organization entered more than \$15,000 on Form 990-EZ, line 6a.

OMB No. 1545-0047

Attach to Form 990 or Form 990-EZ.

► Go to www.irs.gov/Form990 for instructions and the latest information.

Inspection

Name of the organization						Employer ide	ntification number	
HOMES W	ITH HOPE, INC.					22-2534	326	
Part II. Fundraising Activities. Complete if the organization answered "Yes" on Form 990, Part IV, line 17. Form 990-EZ filers are not required to complete this part.								
Indicate whether the organization rais a Mail solicitations b Internet and email solicitations c Phone solicitations d In-person solicitations 2 a Did the organization have a written or key employees listed in Form 990, Pab If "Yes," list the 10 highest paid individuals.	ed funds through any of the followin e Solicitat f Solicitat g Special r oral agreement with any individual art VII) or entity in connection with priduals or entities (fundraisers) pursui	ion of ion of fundra (includ	non-ga govern ising of ing of onal fu	overnment grants nment grants events ficers, directors, trus undraising services?		☐ Yes		
compensated at least \$5,000 by the	organization.						•	
(i) Name and address of individual or entity (fundraiser)	(ii) Activity	(iii) fundr have cr or con contribe	Did alser ustody trol of rtions?	(iv) Gross receipts from activity	to (0	Amount paid or retained by) fundraiser ted in col. (i)	(vi) Amount paid to (or retained by) organization	
		Yes	No					
							-	
							-	
							,	
Total			>					
3 List all states in which the organization or licensing.			utions	or has been notified	l it is	exempt from re	gistration	
			•					
	-							
						-		
								
		-						
								

LHA For Paperwork Reduction Act Notice, see the Instructions for Form 990 or 990-EZ.

Schedule G (Form 990 or 990-EZ) 2019

7. N. O.	LL I	of fundraising event contributions and gro				
			(a) Event #1	(b) Event #2	(c) Other events	(d) Total events
			STAND UP	GRT LUNCHEON	1.	(add col. (a) through
			(event type)	(event type)	(total number)	col. (c))
nue				, ,,,	,	
Revenue	1	Gross receipts	443,678.	81,796.	73,674.	599,148.
ш.	2	Less: Contributions	385,978.	64,330.	58,674.	508,982.
	3	Gross income (line 1 minus line 2)	57,700.	17,466.	15,000.	90,166.
	4	Cash prizes				
	5	Noncash prizes				
benses	6	Rent/facility costs				
Direct Expenses	7	Food and beverages				
	8	Entertainment				
	9	Other direct expenses			19,122.	164,230.
	10	Direct expense summary. Add lines 4 through		•••••		164,230.
P	11	Net income summary. Subtract line 10 from li Gaming. Complete if the organization				-74,064.
# J.S.		\$15,000 on Form 990-EZ, line 6a.	answered res dirioni	1330, Fait IV, IIII C 13, OF I	eported more than	
_		4.0,000 0.000 0.		(b) Pull tabs/instant		(d) Total gaming (add
nue			(a) Bingo	bingo/progressive bingo	(c) Other gaming	col. (a) through col. (c))
Revenue						
	1	Gross revenue				
es	2	Cash prizes				
Direct Expenses	з	Noncash prizes				
Direct	4	Rent/facility costs				
	5	Other direct expenses				
	6	Volunteer labor	Yes %	Yes% No	☐ Yes % ☐ No	
	7	Direct expense summary. Add lines 2 through	n 5 in column (d)		>	
	8	Net gaming income summary. Subtract line 7	' from line 1, column (d)		>	
_	_					
	ı İs	iter the state(s) in which the organization condu the organization licensed to conduct gaming a "No," explain:	ctivities in each of these	states?		Yes No
		ere any of the organization's gaming licenses re "Yes," explain:	•		rear?	Yes No
	_			-		
	_			-		
9320	182 G	9-11-19			Schedule G (Fo	rm 990 or 990-EZ) 201 9

Schedule G (Form 990 or 990-EZ) 2019 HOMES WITH HOPE, INC.	22-2534326 Page 3
11 Does the organization conduct gaming activities with nonmembers?	
12 Is the organization a grantor, beneficiary or trustee of a trust, or a member of a partnership or other entity formed	
to administer charitable gaming?	Yes No
13 Indicate the percentage of gaming activity conducted in:	
a The organization's facility	13a %
	• • • • • • • • • • • • • • • • • • •
 b An outside facility Enter the name and address of the person who prepares the organization's gaming/special events books and recor 	
14 Enter the name and address of the person who prepares the organization's gaming/special events books and recor	as:
Name ▶	
Address	
15a Does the organization have a contract with a third party from whom the organization receives gaming revenue?	Yes No
b If "Yes," enter the amount of gaming revenue received by the organization ▶ \$ and the am	ount
of gaming revenue retained by the third party >\$	
c if "Yes," enter name and address of the third party:	•
on roo, onto hamo and dedicto of the time party.	
Name	. ~
	4
Address	
16 Gaming manager information:	∴
16 Gaming manager information:	÷
Name ▶	
Gaming manager compensation > \$	
Description of services provided	
	- <u>-</u> -
	<u></u>
Director/officer Employee Independent contractor	
17 Mandatory distributions:	
a is the organization required under state law to make charitable distributions from the gaming proceeds to	
retain the state gaming license?	Yes No
b Enter the amount of distributions required under state law to be distributed to other exempt organizations or spent	in the
organization's own exempt activities during the tax year > \$	
Part IV Supplemental Information. Provide the explanations required by Part I, line 2b, columns (iii) and (v); and Part III, lines 9, 9b, 10b,
15b, 15c, 16, and 17b, as applicable. Also provide any additional information. See instructions.	
· · · · · · · · · · · · · · · · · · ·	
	<u></u>

932083 09-11-19

Schedule G	(Form 990 or 990-EZ)	HOMES	WITH	HOPE,	INC.			22-2534326	Page 4
Part IV	(Form 990 or 990-EZ) Supplemental Infor	mation _{(co}	ntinued)						
	- -								
							_		
	<u> </u>								
	<u>-</u>								
									•
									_
					-				
								-	
					· · · · · · · · · · · · · · · · · · ·				
									
						-			
•=								·, -	-
-									
	-								
					<u>-</u>				ath.
-									

SCHEDULE J (Form 990)

Compensation Information

For certain Officers, Directors, Trustees, Key Employees, and Highest
Compensated Employees

Complete if the organization answered "Yes" on Form 990, Part IV, line 23.

► Attach to Form 990. ► Go to www.irs.gov/Form990 for instructions and the latest information.

OMB No. 1545-0047

Inspection 11

Internal Revenue Service Name of the organization

Department of the Treasury

HOMES WITH HOPE, INC.

Employer identification number 22-2534326

Pa	rt L Questions Regarding Compensation			
			Yes	No
1a	Check the appropriate box(es) if the organization provided any of the following to or for a person listed on Form 990,			
	Part VII, Section A, line 1a. Complete Part III to provide any relevant information regarding these items.	5.00	ani ya	773 334
	First-class or charter travel			
	Travel for companions Payments for business use of personal residence			
	Tax indemnification and gross-up payments Health or social club dues or initiation fees		1	100
	Discretionary spending account Personal services (such as maid, chauffeur, chef)			3.0
ь	If any of the boxes on line 1a are checked, did the organization follow a written policy regarding payment or	5.7	a zakania	4
	reimbursement or provision of all of the expenses described above? If "No," complete Part III to explain	1b		
2	Did the organization require substantiation prior to reimbursing or allowing expenses incurred by all directors,			
	trustees, and officers, including the CEO/Executive Director, regarding the items checked on line 1a?	2		PARTY VINENTALVAL
		111-12-11	19.4	ide, is
3	Indicate which, if any, of the following the organization used to establish the compensation of the organization's			
_	CEO/Executive Director. Check all that apply. Do not check any boxes for methods used by a related organization to			
	establish compensation of the CEO/Executive Director, but explain in Part III.		7	
	X Compensation committee Written employment contract			
	Independent compensation consultant Compensation survey or study			
	Form 990 of other organizations X Approval by the board or compensation committee			*
	Point 990 of other diganizations		200	September 1
	During the year, did any person listed on Form 990, Part VII, Section A, line 1a, with respect to the filing			
4			4	
_	organization or a related organization:	4a		X
	Receive a severance payment or change-of-control payment? Participate in, or receive payment from, a supplemental nonqualified retirement plan?	4b		X
b	Participate in, or receive payment from, a supplemental hondulained retirement plan: Participate in, or receive payment from, an equity-based compensation arrangement?	4c		X
G	If "Yes" to any of lines 4a-c, list the persons and provide the applicable amounts for each item in Part III.	4C	10111	AR 2700
	if tes to any of lines 4a-c, list the persons and provide the applicable amounts for each item in Fart in.			300
	Only anation 504/2/00 504/2/40 and 504/2/000 annuinctions much complete lines 5.0		3,0	
_	Only section 501(c)(3), 501(c)(4), and 501(c)(29) organizations must complete lines 5-9. For persons listed on Form 990, Part VII, Section A, line 1a, did the organization pay or accrue any compensation		. # L M	4.7
5		1, 0		
_	contingent on the revenues of: The organization?	5a	\$1,000 photos	X
		5b		X
D	Any related organization? If "Yes" on line 5a or 5b, describe in Part III.	200		
6	For persons listed on Form 990, Part VII, Section A, line 1a, did the organization pay or accrue any compensation			
0	contingent on the net earnings of:		4114	¥.1
•	The appropriation 2	6a		X
	Any related organization?	6b	<u> </u>	X
	If "Yes" on line 6a or 6b, describe in Part III.	**	(A)	#7 K44
7	For persons listed on Form 990, Part VII, Section A, line 1a, did the organization provide any nonfixed payments			300
•	not described on lines 5 and 6? If "Yes," describe in Part III	7	- Walking	X
8	Were any amounts reported on Form 990, Part VII, paid or accrued pursuant to a contract that was subject to the	100	0.08	5474
0	The state of the s	8		Х
^	initial contract exception described in Regulations section 53.4958-4(a)(3)? If "Yes," describe in Part III	•	Giller Fr	
9		9		
_	Regulations section 53.4958-6(c)?	1 9		

LHA For Paperwork Reduction Act Notice, see the Instructions for Form 990.

Schedule J (Form 990) 2019

Part II. Officers, Directors, Trustees, Key Employees, and Highest Compensated Employees. Use duplicate copies if additional space is needed.

For each individual whose compensation must be reported on Schedule J, report compensation from the organization on row (i) and from related organizations, described in the instructions, on row (ii). Do not list any individuals that aren't listed on Form 990, Part VII.

Note: The sum of columns (B)(i)-(iii) for each listed individual must equal the total amount of Form 990, Part VII, Section A, line 1a, applicable column (D) and (E) amounts for that individual.

		(B) Breakdown of	W-2 and/or 1099-MI	SC compensation	(C) Retirement and other deferred	(D) Nontaxable benefits	(E) Total of columns (B)(i)-(D)	(F) Compensation in column (B)
(A) Name and Title		(i) Base compensation	(ii) Bonus & incentive compensation	(iii) Other reportable compensation	compensation	Denents	(6)(1)-(0)	reported as deferred on prior Form 990
(1) JEFFREY N. WIESER	(i)	145,000.	0.	0.	0.	20,112.	165,112.	0.
PRESIDENT/CEO	(ii)	0.	0.	0.	0.	0.	0.	0.
	(i)							
	(ii) [
	(i)							
	(ii)							
	(i)							
	(ii) [_						
	(i)							
	(ii)							<u>-</u>
-	(i)							
	(ii)							_
-	(i)	·						
	(ii)					_		_
	(i)		_					
<u></u>	(ii) [
	(i)							
	(ii)							
	(i)							<u>_</u>
	(ii)						_	_
 _	(i)	,						<u>_</u>
	(ii) [
	(i)			-				
	(ii)		_			<u></u>		
	(i)					<u> </u>		
	(ii)							
	(i)							
	(ii)							
-	(i)							
	(ii)							
	(i)				_		<u> </u>	
	(ii)							

Schedule J (Form 990) 2019	HOMES WITH HOPE, INC	• •		22-2534326	Page 3
Part III Supplemental Informati	ion				
Provide the information, explanatio	on, or descriptions required for Part I, lines 1	a, 1b, 3, 4a, 4b, 4c, 5a, 5b, 6a, 6b, 7	, and 8, and for Part II. Also complete th	nis part for any additional information.	
				· -	
-	· ·				
			•		
					
					
				-	
	<u> </u>				
					
		· · · · ·			-
					

Schedule J (Form 990) 2019

SCHEDULE M (Form 990)

Noncash Contributions

OMB No. 1545-0047

Inspection

Department of the Treasury Internal Revenue Service

Name of the organization

► Complete if the organizations answered "Yes" on Form 990, Part IV, lines 29 or 30.

Attach to Form 990.

Go to www.irs.gov/Form990 for instructions and the latest information.

Employer identification number 22-2534326 HOMES WITH HOPE, INC.

Par	Types of Property					
		(a) Check if applicable	(b) Number of contributions or items contributed	(c) Noncash contribution amounts reported on Form 990, Part VIII, line 1g	Method of	(d) determining ibution amounts
1	Art - Works of art					
2	Art - Historical treasures					
3	Art - Fractional interests			•		
4	Books and publications		ong Argan Lag		·	
5	Clothing and household goods	Х		129,190.	ESTIMATED	COST
6	Cars and other vehicles			·	-	•
7	Boats and planes					
8	Intellectual property			· -		•
9	Securities - Publicly traded			-		<u>.</u>
10	Securities - Closely held stock					
11	Securities - Partnership, LLC, or			•		-
	trust interests					,
12	Securities - Miscellaneous					
13	Qualified conservation contribution -					
	Historic structures					
14	Qualified conservation contribution - Other					
15	Real estate - Residential					
16	Real estate - Commercial					
17	Real estate - Other					
18	Collectibles					•
19	Food inventory	X	3	246,400.	ESTIMATED	COST
20	Drugs and medical supplies					
21	Taxidermy					
22	Historical artifacts					
23	Scientific specimens					
24	Archeological artifacts					
25	Other					
26	Other • ()					
27	Other ()				1	
28	Other (L		<u> </u>	
29	Number of Forms 8283 received by the organization completed Form 828			I I		0
	During the year, did the organization receive by must hold for at least three years from the date exempt purposes for the entire holding period?	of the initia	l contribution, and		sed for	
	If "Yes," describe the arrangement in Part II.					
31	Does the organization have a gift acceptance p	-	•		tions?	31 X
32a	Does the organization hire or use third parties of contributions?					32a X
b	If "Yes," describe in Part II.					
33	If the organization didn't report an amount in codescribe in Part II.	olumn (c) fo	r a type of property	y for which column (a) is che	cked,	

For Paperwork Reduction Act Notice, see the Instructions for Form 990. LHA

Schedule M (Form 990) 2019

Schedule M (Form 990) 2019

932142 09-27-19

SCHEDULE O

(Form 990 or 990-EZ)

Supplemental Information to Form 990 or 990-EZ

Complete to provide information for responses to specific questions on Form 990 or 990-EZ or to provide any additional information.

OMB No. 1545-0047 Open to Public Inspection

Attach to Form 990 or 990-EZ. Department of the Treasury Go to www.irs.gov/Form990 for the latest information. Internal Revenue Service Employer identification number Name of the organization 22-2534326 HOMES WITH HOPE, INC. LINE 1, DESCRIPTION OF ORGANIZATION MISSION: FORM 990. PART I. MANAGEMENT SERVICES, SUPPORTIVE HOUSING UNITS, MENTORING AND YOUTH ACADEMIC SUPPORT. LINE 1, DESCRIPTION OF ORGANIZATION MISSION: FORM 990, PART III. ACHIEVE INDEPENDENT LIVES. PART III, LINE 3, CHANGES IN PROGRAM SERVICES: CONVERTED BACHARACH COMMUNITY FROM SHELTER SERVICES TO SUPPORTIVE HOUSING. FORM 990. PART VI. SECTION A. LINE 3: HOMES WITH HOPE, INC. CONTRACTS WITH A PROPERTY MANAGEMENT COMPANY WHO COLLECTS RENTS, MAKES DEPOSITS, AND REPORTS/MAINTAINS A GENERAL LEDGER (DEMARÇO MANAGEMENT CORP.). FORM 990, PART VI, SECTION A, LINE 4: HOMES WITH HOPE. INC. MADE CHANGES TO ITS BYLAWS AND CREATED SUCCESSION PLAN. FORM 990, PART VI, SECTION A, LINE 8B: THE MEETINGS ARE DOCUMENTED ON A CASE-BY-CASE BASIS. EXAMPLE: THE FINANCE COMMITTEE MEETS ON A MONTHLY BASIS. A COMPLETE FINANCIAL MANAGEMENT REPORTING PACKAGE IS DEVELOPED BY STAFF AND REVIEWED WITH THE FC MEMBERS. TWO SELECTED REPORTS FROM THAT PACKAGE ARE THEN DISTRIBUTED TO THE FULL

Schedule O (Form 990 or 990-EZ) (2019) LHA For Paperwork Reduction Act Notice, see the Instructions for Form 990 or 990-EZ.

BOARD AND THE "TREASURER'S REPORT" IS GIVEN AT THE SUBSEQUENT BOARD

SCHEDULE R (Form 990)

Related Organizations and Unrelated Partnerships

Complete if the organization answered "Yes" on Form 990, Part IV, line 33, 34, 35b, 36, or 37.

Attach to Form 990.

Department of the Treasury Internal Revenue Service

▶ Go to www.irs.gov/Form990 for instructions and the latest information.

OMB No. 1545-0047 Open to Public Inspection

Employer identification number

Name of the organization HOMES WITH	HOPE, INC.				E	Employer identifi 22-25343		ımber
Part le Identification of Disregarded Entities. Co	omplete if the organization answered "Y	es" on Form 990, Part IV, line 3	3.					_
(a) Name, address, and EIN (if applicable) of disregarded entity	(b) Primary activity	(c) Legal domicile (state of foreign country)	(d) or Total inco	me End-of-yea		ts Direct o	(f) controlling ntity)
						-		
						 		
Part II Identification of Related Tax-Exempt Orgonizations during the tax year.	ganizations. Complete if the organization	on answered "Yes" on Form 996	D, Part IV, line 34, i	pecause it had one	e or mo	re related tax-exe	mpt	
(a) Name, address, and EiN of related organization	(b) Primary activity	(c) Legal domicile (state or foreign country)	(d) Exempt Code section	(e) Public charity status (if section		(f) irect controlling entity	cont	g) 512(b)(13) trolled tity?
IHA, INC 06-1566759			-	501(c)(3))	+		Yes	No_
49 RICHMONDVILLE AVENUE, SUITE 112 WESTPORT, CT 06880	TO PROVIDE AFFORDABLE,	CONNECTICUT	501(C)(3)	LINE 9	N/A			x
							1	
		_				_	1	-
For Paperwork Reduction Act Notice, see the Instr	uctions for Form 990.		<u> </u>			Schedule R	 (Form 9	90) 2019

For Paperwork Reduction Act Notice, see the Instructions for Form 990.

Part III Identification of Related Organizations Taxable as a Partnership. Complete if the organization answered "Yes" on Form 990, Part IV, line 34, because it had one or more related organizations treated as a partnership during the tax year.

(a)	(b)	(c)	(d)	(e)	(f)	(g)	(1	h)	(i)	(j)	(k)
Name, address, and EIN of related organization	Primary activity	Legal domicile (state or foreign	Direct controlling entity	Predominant income (related, unrelated, excluded from tax under sections 512-514)	Share of total income	Share of end-of-year assets	alloca	ortionate tions?	Code V-UBI amount in box 20 of Schedule	General or managing partner?	Percentage ownership
		country)		sections 512-514)			Yes	No	K-1 (Form 1065)	Yes No	
							1				
		ł									
		Ì									
		<u> </u>	_				+		· · · -	 	-
<u>-</u>	ł					ı		ļ			
	1										
_	1										_
										1 1 -	
	1										
	1										
											_
:			_								ļ
]			•			}			1 1	
]		ļ	1			İ		}		
				<u></u>					<u> </u>		

Part IV Identification of Related Organizations Taxable as a Corporation or Trust. Complete if the organization answered "Yes" on Form 990, Part IV, line 34, because it had one or more related organizations treated as a corporation or trust during the tax year.

(a)	(b)	(c)	(d)	(e)	(f)	(g)	(h)	(i Sec	i) :tion
Name, address, and EIN of related organization	Primary activitý	Legal domicile (state or foreign	Direct controlling entity	Type of entity (C corp, S corp, or trust)	Share of total income	Share of end-of-year assets	Percentage ownership		i) ction b)(13) rolled dty?
		country)	-					Yes	No
		ļ							
									—
				1					
									<u> </u>
									l
	<u> </u>		<u> </u>			<u> </u>	ļ		—
									1
						1	İ		
							_		├──
	<u> </u>			<u> </u>		<u></u>	1.57		

1a

1b

Page 3

Note: Complete line 1 if any entity is listed in Parts II, III, or IV of this schedule.

Part V. Transactions With Related Organizations. Complete if the organization answered "Yes" on Form 990, Part IV, line 34, 35b, or 36.

During the tax year, did the organization engage in any of the following transactions with one or more related organizations listed in Parts II-IV?

a Receipt of (i) interest, (ii) annuities, (iii) royalties, or (iv) rent from a controlled entity

b Gift, grant, or capital contribution to related organization(s)

c Gift, grant, or capital contribution from related organization(s)			***************************************	1c X
d Loans or loan guarantees to or for related organization(s)				1d X
e Loans or loan guarantees by related organization(s)				1 . 137
				3.5
f Dividends from related organization(s)		······································		1f X
g Sale of assets to related organization(s)				1g X
h Purchase of assets from related organization(s)				1h X
i Exchange of assets with related organization(s)				1 1 37
i Lease of facilities, equipment, or other assets to related organization(s)				
Lease of facilities, equipment, or other according to related digarination (-)	***************************************	***************************************	***************************************	
k Lease of facilities, equipment, or other assets from related organization(s)				1k X
Performance of services or membership or fundraising solicitations for related organizations.				
m Performance of services or membership or fundraising solicitations by related org				1 1 77
n Sharing of facilities, equipment, mailing lists, or other assets with related organizations.				
				v
6 Sharing of paid employees with related organization(s)				
p Reimbursement paid to related organization(s) for expenses				. l v
·				···· •
q Reimbursement paid by related organization(s) for expenses	•••••			
				. V
r Other transfer of cash or property to related organization(s)				77
s Other transfer of cash or property from related organization(s) 1 If the answer to any of the above is "Yes," see the instructions for information on		is the including payared relation	enchine and transaction thresholds	.,, <u>,</u> <u>15 </u>
2 If the answer to any of the above is "Yes," see the instructions for information on	Muo must complete th	is line, including covered relation		
(a) Name of related organization	(b) Transaction	(c) Amount involved	(d) Method of determining amou	nt involved
Natile of fedated organization	type (a-s)	Amount involved	Monitor of Colornia in g allice	
	 			_
<u>(1)</u>				
(2)	_	-		
		_		
		}		
(4)				
(5)				
(6)	<u> </u>			
932163 09-10-19			Sche	dule R (Form 990) 2019
	45			

Page 4

Part VI Unrelated Organizations Taxable as a Partnership. Complete if the organization answered "Yes" on Form 990, Part IV, line 37.

Provide the following information for each entity taxed as a partnership through which the organization conducted more than five percent of its activities (measured by total assets or gross revenue) that was not a related organization. See instructions regarding exclusion for certain investment partnerships.

at was not a related organization. See ins (a) Name, address, and EIN of entity	(b) Primary activity	(c)	(e) Are all partners sec 501(c)(3) orgs.? Yes No	(f) Share of total income	(g) Share of end-of-year assets	Dispr tion alloca Yes	n) opor- nate tions?	(i) Code V-UBI amount in box 20 of Schedule K-1 (Form 1065)	(j) General o managing partner? Yes NO	(k) Percentage ownership
		e.								
									,	
										_
							_			
						<u> </u>				
				<u> </u>		1	,a			
	<u> </u>									m 990) 2019

Schedule R	(Form 990) 2019	HOMES	WITH	HOPE,	INC.	22-2534326	Page 5
Part VII	(Form 990) 2019 Supplemental Info	rmation					
	Bradda additional inform				- Oakadula B. Oaa isataaniisaa		
	Provide additional inform	iation for rest	onses to	questions of	n Schedule R. See instructions.		
_							
			_				
	•						
						-	-
-							
						•	
	 						
-						-	
-							
				•			
							
							
	 						
			-		-	-	

Proposal to Revise the Retirement Plan For Non-Union Supervisory Employees Of The Town of Westport.

Submitted by Gary Conrad, Finance Director and Ralph Chetcuti, Personnel/Human Resources Director

Westport RTM

April 6, 2021

This is a proposal to revise the Non-Union Supervisor Pension Plan to include the current management of the Police and Fire Departments allowing them to maintain the benefits as stated in the Police Pension Fund of the Town of Westport, prior to July 1, 2017. The individuals involved are the following:

Police Chief Foti Koskinas
Police Deputy Chief Sam Arciola
Police Captain David Farrell
Police Captain Ryan Paulson
Police Lieutenant Jillian Cabana
Police Lieutenant Matthew Gouveia
Police Lieutenant Anthony Prezioso
Police Lieutenant David Wolf
Police Lieutenant Eric Woods
Fire Chief Robert Yost
Fire Deputy Chief Michael Kronick
Fire Marshal Nate Gibbons

In November of 2017, the Assistant Fire Chiefs voted to become members of Local 2081, Westport Firefighters Union. As a result of this action, the Fire Department lost an important level of Non-Union managers. This also resulted in the five Assistant Chiefs became eligible for overtime pay. Since then, the Town has paid that cohort \$772,549 in overtime wages.

The individuals involved, besides being the management team of their respective departments, work a considerable amount of time beyond the normal work week, in sometimes very difficult situations, without additional compensation. You may be aware some of the officers whom they supervise are eligible for overtime pay which results in some of them earning substantially more than their supervisors.

In order to acknowledge the effort and service that each of these senior management personnel give to the Town of Westport, we propose this change in the pension plan. It will also be an incentive to retain them as Town employees and to remain Non-Union managers.

The Town actuarial firm, Milliman, has calculated that by making this change, it will require approximately an additional \$215,000 contribution to the Towns annual Actuarially Determined Contributions ("ADC"). We believe that the potential cost of overtime for anyone below the title of Deputy Chief in the police department, should they decide to become members of the union, would be substantial higher than the increase in the ADC.

!!

Finally, it should be noted that as these individuals leave their positions with the Town, their replacements will continue to participate in the Police Pension Fund of the Town of Westport and will not participate in the Non-Union Supervisor Pension Plan.



RTM Employee Compensation, Finance, and Public Protection Committee Joint Report Proposed Revision to the Retirement Plan for Non-Union Supervisory Employees May 19, 2021

RTM Members Present:

Employee Compensation Committee (7 of 9)

Louis Mall, Chair; Peter Gold; Jimmy Izzo; Rick Jaffe; Jay Keenan; Nicole Klein; Sal Liccione

Finance Committee (6 of 9)

Jeff Wieser, Chair; Seth Braunstein; Rick Jaffe; Nicole Klein; Stephen Shackelford; Cathy Talmadge

Public Protection Committee (8 of 9)

Jimmy Izzo, Chair; Candace Banks; Seth Braunstein; Andrew Colabella; Kristan Hamlin; Rick Jaffe; Dick Lowenstein; Louis Mall

Other Presenters in Attendance:

Ralph Chetcuti, Personnel / Human Resources Director, Town of Westport Gary Conrad, Finance Director, Town of Westport Floyd Dugas, Attorney, Berchem, Moses & Devlin and labor council for the Town Foti Koskinas, Chief, Police Department

Others in Attendance

Nick Marsan, Lieutenant, Fire Department David Farrell, Captain, Police Department Matthew Gouveia, Lieutenant, Police Department Ryan Paulsson, Captain, Police Department Eric Woods, Lieutenant, Police Department Velma Heller, Moderator, RTM

The RTM Employee Compensation, Finance, and Public Protection committees met together via Zoom to consider the following:

"To make a recommendation to the RTM to take such action as the meeting may determine, upon a request by the Finance Director and the Personnel/Human Resources Director, to revise the Retirement Plan for Non-Union Supervisory Employees of the Town of Westport to include current management of the Police and Fire Departments."

We are near the culmination of a process that has been in planning and development for several years.

We are asked to approve a proposed revision to the above-named Retirement Plan to allow certain of our Fire and Police Department management to be covered by the Retirement Plan, and to receive benefits under the plan. These members will no longer have future benefits accrue under other Westport pension plans. The impetus for the proposed revision is a perceived need to acknowledge the contribution of these individuals to their departments, and to the town, thereby helping Westport to maintain a strong level of management in our Fire and Police Departments.

Without the proposed revision, management personnel in our Fire and Police Departments, classified as "exempt" employees, are not eligible for compensation for overtime. In the recent past, also excluded from the Retirement Plan, some of our formerly "exempt" Fire Department management personnel elected to join the union in order to qualify for overtime pay and other benefits. It is considered by Police Chief Koskinas, Human Resources Director Chetcuti, and Finance Director Conrad that approving the proposed Retirement Plan revisions, thereby including our remaining Fire and Police management personnel in the Retirement Plan, will be:

- Substantially less costly to the Town than paying overtime and other union benefits, should those non-union Police management personnel elect to join the union;
- A strong and definite benefit in the Town's effort, not only to hire the best Public Protection people, but also to retain them.

In short, approval of the proposed Retirement Plan revisions is one of the steps our Town must take to encourage qualified Public Protection personnel to join Westport's workforce, and to remain with Westport.

The annual cost of extending the Retirement Plan to the seven Police and one Fire Department personnel who would qualify is expected to be approximately \$215,000 in fiscal year 2022. Chief Koskinas, one of the guest speakers who discussed the plan, is not among the Police Department management personnel who could be impacted by the proposed revisions.

Questions

- Do our three presenting Town employee guests recommend approval? Answer: All three
 Town employee guest speakers with knowledge and expertise in this matter, Chief
 Koskinas, Human Resources Director Chetcuti, and Finance Director Conrad, expressed
 strong recommendations in favor.
- Would the Fire Department management personnel who joined the union be able to leave the union and elect to join the revised Retirement Plan? If so, would it be cost effective for the Town of Westport? Answer: Once the union is involved, it's not that simple.

Reservations expressed by Committee members included the following:

 The text of the proposed revisions was not delivered timely to the members of the three RTM committees. Several members from all three committees expressed reservations at not having been given adequate time to review the details of the proposed revisions.
 Other committee members put trust in the strong and unanimous recommendations of our experts in this complex and sensitive matter. Motions for all three Committees in favor of recommending approval of the proposed revisions, all made by Rick Jaffe, and seconded by Sal Liccione (EE Comp), Seth Braunstein (Finance), and Andrew Colabella (Pub Protection) passed as follows:

- Employee Compensation: 6 0 in favor, with one abstention (Gold) for the stated purpose of wanting to read the documents more carefully
- Finance: 5 0 in favor, with one abstention (Braunstein)

 for the stated purpose of reading through and understanding the documents in detail
- Public Protection: 5-0 in favor, with three abstentions (Braunstein, Hamlin, Lowenstein) one of which (Braunstein) was for the stated purpose of reading through and understanding the documents in detail

Respectfully submitted,

Rick Jaffe, RTM 1

RETIREMENT PLAN FOR NON-UNION AND NON-SUPERVISORY EMPLOYEES OF THE TOWN OF WESTPORT, CONNECTICUT As Amended and Restated Effective as of July 1, 2019 2019 Amendment and Restatement

Table of Contents

	Table of Contents	
	•	Page
	ARTICLE 1 DEFINITIONS	
DEF	FINITIONS	1
	ARTICLE 2 ELIGIBILITY	
2.1	IN GENERAL	9
2.2	REEMPLOYMENT FOLLOWING RETIREMENT	9
	ARTICLE 3 RETIREMENT DATES	
3.1	NORMAL RETIREMENT	10
3.2	DEFERRED RETIREMENT	10
	ARTICLE 4 PENSION BENEFITS	
4.1	NORMAL OR DEFERRED RETIREMENT PENSION	11
4.2	MAXIMUM PENSION	11
4.3	MINIMUM PENSION	11
4.4	COVERAGE UNDER MEDICAL PLAN FOR PARTICIPANTS RETIRING BEFORE NOVEMBER 1, 2005	11
4.5	4.4A CONTINUED COVERAGE UNDER MEDICAL PLAN FOR PARTICIPANTS RETIRING ON OR AFTER NOVEMBER 1, 2005	12
4.6	[INSERT HEADING] Error! Bookmark not de	fined.
4.7	LIMITS ON BENEFITS	13
	ARTICLE 5 FORM OF RETIREMENT PAYMENT AND ELECTION PROCEDURES	
5.1	RETIREMENT PAYMENT FORMS	14
5.2	ELECTION AND REVOCATION OF OPTIONAL FORMS	14
5.3	DEATH OF JOINT ANNUITANT	14
5.4	UNCLAIMED BENEFITS	14
5.5	DIRECT ROLLOVERS OF ELIGIBLE ROLLOVER DISTRIBUTIONS	15
5.6	OUALIFIED MILITARY SERVICE	16

Table of Contents

	-	age
	ARTICLE 6 DISABILITY BENEFITS	
6.1	ELIGIBILITY AND DURATION	17
6.2	AMOUNT OF BENEFIT	17
6.3	DETERMINATION OF DISABILITY BENEFITS BY PENSION COMMITTEE	18
6.4	ELECTION AS TO FORM OF PAYMENT	18
	ARTICLE 7 VESTED BENEFITS AND DEATH BENEFITS	
7.1	VESTED BENEFITS	19
7.2	DEATH BENEFITS PRIOR TO RETIREMENT	19
7.3	DEATH BENEFITS AFTER RETIREMENT	20
	ARTICLE 8 REQUIRED DISTRIBUTIONS	
8.1	7A.1 GENERAL RULES	21
8.2	7A.2 TIME AND MANNER OF DISTRIBUTION	21
8.3	7A.3 DETERMINATION OF AMOUNT TO BE DISTRIBUTED EACH YEAR	22
8.4	7A.4 REQUIREMENTS FOR ANNUITY DISTRIBUTIONS COMMENCING DURING A PARTICIPANT'S LIFETIME	22
8.5	7A.5 REQUIREMENTS FOR MINIMUM DISTRIBUTIONS WHERE PARTICIPANT DIES BEFORE DATE DISTRIBUTIONS BEGIN	23
8.6	7A.6 LIFE EXPECTANCY	23
8.7	7A.7 DEFINITIONS	23
	ARTICLE 9 CONTRIBUTIONS AND PENSION FUND	
9.1	PARTICIPANT CONTRIBUTIONS	25
9.2	EMPLOYER CONTRIBUTIONS	.25
9.3	PENSION FUND	.25
9.4	RESTORATION OF FORFEITURE	25

Table of Contents

	ARTICLE 10	<u>Page</u>
	ALIENATION, FACILITY OF PAYMENT AND REEMPLOYMENT	
10.1	ALIENATION	27
10.2	FACILITY OF PAYMENT	27
10.3	SUSPENSION OF BENEFITS	27
	ARTICLE 11 ADMINISTRATION OF PLAN	
11.1	PENSION COMMITTEE	28
11.2	MEETINGS OF PENSION COMMITTEE	28
11.3	COMPENSATION OF PENSION COMMITTEE	28
11.4	AUTHORITY, POWERS AND DUTIES OF PENSION COMMITTEE	28
11.5	MODIFICATION OF PLAN	30
11.6	PROTECTION OF PENSION COMMITTEE	30
11.7	REVIEW PROCEDURES OF PENSION COMMITTEE	31
11.8	FINAL AND BINDING DECISIONS OF PENSION COMMITTEE	32
	ARTICLE 12 RETIREE MEDICAL EXPENSES	
12.1	EFFECTIVE DATE	33
12.2	PAYMENT OF RETIREE MEDICAL EXPENSES	
12.3	SEPARATE ACCOUNT	33
12.4	KEY EMPLOYEE	33
12.5	CO-ORDINATION WITH OTHER COVERAGE	34
12.6	USE OF FORFEITURES	34
12.7	EXCESS PENSION ASSETS	34
12.8	DEFINITIONS	35
12.9	ANNUAL LIMITATIONS	35
	ARTICLE 13 AMENDMENT, TERMINATION, BENEFIT LIMITATIONS, MERGER A APPROVAL OF PLAN	ND
13.1	AMENDMENT OF PLAN	37

Table of Contents

		Page Page
13.2	TERMINATION OR PARTIAL TERMINATION OF PLAN	37
13.3	ALLOCATION OF ASSETS UPON PLAN TERMINATION	37
13.4	INVALID PROVISIONS	38
13.5	NO GUARANTY OF EMPLOYMENT; EFFECT OF DENIAL OF BENEFITS	38
13.6	USAGE	38
13.7	ELECTION FOR CERTAIN PARTICIPANTS TO BE COVERED UNDER PLAN AS AMENDED AND RESTATED EFFECTIVE AS OF NOVEMBER 1,	
	2005 Error! Bookmark not de	fined.

PREAMBLE

The Retirement Plan for Non-Union and Non-Supervisory Employees of the Town of Westport (the "Plan") was originally adopted on July 1, 1981. It was subsequently amended effective as of July 1, 1986, effective as of July 1, 1998, effective as of November 1, 2005 and effective as of November 1, 2013. In accordance with Section 13.1 of the Plan, the Town of Westport, Connecticut hereby amends and restates the Plan to read as follows effective as of July 1, 2019 for persons who are actively employed on or after that date, except as otherwise provided herein. The purpose of this amendment and restatement is to reflect applicable amendments required under the 2015 Cumulative List (IRS Notice 2015-84), the 2016 Required Amendments List (IRS Notice 2016-80) and the 2017 Required Amendments List (IRS Notice 201-72). If any provision of the Plan should be subject to more than one interpretation, such provision shall be interpreted in a manner which shall be consistent with the Plan being regarded as a qualified pension plan and the Trust being exempt from tax as aforesaid.

ARTICLE 1 DEFINITIONS

As used herein, the words and phrases below shall have the following meanings:

- 1.1 "Actuarial Equivalent" or "Equivalent Actuarial Value" means a benefit of equivalent value when computed on the basis of the rate of interest of 7% and mortality rates in accordance with the 1983 Group Annuity Mortality Table (unisex based on 50% male/50% female).
- 1.2 "Actuary" means the Enrolled Actuary retained by the Pension Committee in connection with the administration of the Plan.
- 1.3 "Approved Absence" means the period during which a Participant is on leave of absence approved in writing by the Employer. Approved Absence shall also include the period during which the Participant is in military service with the Armed Forces of the United States (including Coast Guard and Merchant Marine Service) if he has reemployment rights under applicable laws and complies with the requirements of the law as to reemployment.
- 1.4 "Average Final Compensation" means, with respect to a Participant, the greater of (a) his Compensation during the calendar year in which his Compensation was the highest, or (b) his final 12 months of Compensation.
 - 1.5 "Code" means the Internal Revenue Code of 1986, as amended.
 - 1.6 "Compensation" means, with respect to a Participant, the following:
- (A) For all purposes other than Section 4.6, the basic salary regularly paid by the Employer to such Participant,
- (1) Exclusive of any overtime pay, bonuses, gratuities, commissions, retainer fees, benefits, severance pay, allowance for expenses or other special remuneration paid to such Participant; and

- (2) Increased by any "pick-up" contributions with respect to such Participant which are designated as Employer contributions in accordance with Section 414(h)(2) of the Code.
- (B) For purposes of Section 4.6, such Participant's wages for the Plan Year paid by the Employer of the type reported in Box 1 of Form W-2. Such wages shall include amounts within the meaning of Section 3401(a) of the Code plus any other amounts paid to him by the Employer for which the Employer is required to furnish a written statement under Sections 6041(d) and 6051(a)(3) of the Code, determined without regard to any rules that limit the amount required to be reported based on the nature or location of the employment or services performed,
- (1) Exclusive of any amounts paid or reimbursed by the Employer for moving expenses which the Employer reasonably believes at the time of such payment to be deductible by the Employee under Section 217 of the Code;
- (2) Increased by the amount of any contributions made by the Employer under any salary reduction or similar arrangement to: (a) a qualified cash or deferred arrangement under Code Section 401(k); (b) a simplified employee pension plan described in Section 408(k) of the Code; (c) a SIMPLE arrangement under Code Section 408(p); (d) an annuity contract described in Section 403(b) of the Code; (e) a deferred compensation plan within the meaning of Section 457(b) of the Code; (f) a cafeteria plan under Code Section 125; and (g) a deferred compensation plan under Code Section 457; and
- (3) Increased by any amounts contributed or deferred by the Employer at his election and which is not includable in his gross income under Section 132(f)(4) of the Code.

Compensation shall be limited as follows:

In addition to any other applicable limitations set forth in the Plan and notwithstanding any other provision in the Plan to the contrary, for Plan Years beginning after January 1, 1989, and prior to January 1, 1994, the annual Compensation of each Participant taken into account under the Plan shall not exceed \$200,000, as adjusted by the Secretary for increases in the cost of living at the same time and in the same manner as under Section 415(d) of the Code (the "TRA 1986 Limit"). The cost-of-living adjustment for a calendar year shall apply to any period (a "Determination Period") not exceeding 12 months, over which Compensation is determined, beginning in such calendar year. If a Determination Period consists of fewer than 12 months, the TRA 1986 Limit shall be multiplied by a fraction whose numerator is the number of months in the Determination Period and whose denominator is 12. If Compensation for any prior Determination Period is taken into account in determining any Participant's benefits accruing in any Plan Year beginning after January 1, 1989, and prior to January 1, 1994, the Compensation for such prior Determination Period shall be subject to the TRA 1986 limit in effect for such prior Determination Period. For purposes of the preceding sentence, for Determination Periods beginning before the first day of the Plan Year beginning on or after January 1, 1989, the TRA 1986 Limit shall be \$200,000.

(Y) In addition to any other applicable limitations set forth in the Plan and notwithstanding any other provision in the Plan to the contrary, for Plan Years beginning on or after January 1, 1994, the annual Compensation of each Participant taken into account under the Plan shall not exceed \$150,000, as adjusted by the Secretary for increases in the cost of living in accordance with Section 401(a)(17)(B) of the Code (the "OBRA 1993 Limit"). The cost-of-living adjustment for a calendar year shall apply to any Determination Period beginning in such calendar year. If a Determination Period consists of fewer than 12 months, the OBRA 1993 Limit shall be multiplied by a fraction whose numerator is the number of months in the Determination Period and whose denominator is 12. If Compensation for any prior Determination Period is taken into account in determining any Participant's benefits accruing in the current Plan Year, the Compensation for such prior Determination Period shall be subject to the OBRA 1993 Limit in effect for such prior Determination Period. For purposes of the preceding sentence, for Determination Periods beginning before the first day of the Plan Year beginning on or after January 1, 1994, the OBRA 1993 Limit shall be \$150,000.

Notwithstanding the preceding provisions, the annual Compensation of each Participant taken into account in determining benefit accruals in any Plan Year beginning after December 31, 2001, shall not exceed \$200,000. The \$200,000 limit in the preceding sentence shall be adjusted for cost-of-living increases in accordance with Section 401(a)(17)(B) of the Code. The cost-of-living adjustment in effect for a calendar year applies to annual Compensation for the determination period that begins with or within such calendar year.

Effective for Limitation Years beginning after July 1, 2007, there shall be included in a Participant's Compensation for purposes of Subsections (A) and/or (B) above, as the case may be, any payment after termination of his employment only if such payment (i) is regular compensation for services during his regular working hours, or compensation for services outside such regular working hours (such as overtime), bonuses or other similar payments; (ii) would have been paid to him before termination of employment if he had continued in employment with the Employer and if so paid, would have been included in "Compensation" under said Subsections (A) and/or (B), as the case may be; and (iii) is paid by the later of 21/2 months after termination of employment or the end of the Limitation Year including the date of termination of employment.

- 1.7 "Credited Service" means Years of Service during which a Participant is eligible for the Plan in accordance with the eligibility provisions set forth in Article 2, including any applicable probationary period, and during which such Participant makes the required contributions as set forth in Article 9.
- (A) Prior to January 1, 2012, if such Participant was a participant in another pension plan to which the Town contributed immediately prior to such Participant's participation in this Plan, his years of Credited Service under this Plan shall include his years of credited service under such other plan, provided that all such Participant's contributions to such other plan with Interest are transferred to this Plan (the "Transfer"). The provisions of this Section 1.7 shall not apply to a Participant unless he shall have waived in writing the right to receive any benefit under such other plan. Such waiver must have been filed with the Town's Personnel Department no later than the later to occur of (1) 90 days after November 1, 2005 or (2) the date such Participant begins participation in this Plan. If the total of such Participant's contributions

to such other plan exceeds the total of what such Participant would have contributed to this Plan had he participated in this Plan during the time he was a participant in such other plan, after the Transfer the excess amount shall be applied against contributions otherwise required from such Participant under this Plan and until such excess amount has been fully so applied such Participant shall not be required to contribute to this Plan.

- (B) The Pension Committee, at its sole discretion, may allow a Participant to buy back all or part of his prior service which is Governmental and Other Related Service (as defined in Section 1.17) under the following provisions:
- (1) For Governmental and Other Related Service immediately prior to employment with the Town: For any prior service allowed to be credited by the Town, such Participant must pay into the Town Plan (as hereinafter defined) the amount he would have paid in contributions for any such period had the Employee been a participant in the Town Plan, plus Interest thereon from the date of such prior service credit to the date payment is made to the Plan.
- (2) For Governmental and Other Related Service not immediately prior to employment with the Town: For any prior service allowed to be credited by the Town, such Participant must pay into the Town Plan, an amount equal to the sum of Clauses (a) and (b) below:
- (a) The amount he would have paid in contributions for any such period had the employee been a Participant in the Town Plan, plus Interest thereon from the date of such prior service credit to the date payment is made to the Plan.
- (b) The amount the Town would have contributed for any such period had the employee been a Participant in the Town Plan, plus interest thereon from the date of such prior service credit to the date payment is made to the Plan. Interest on amounts described in this Section 1.7(2)(b) shall be calculated at the rate of assumed investment income used by the Actuary in determining Plan costs for investment purposes.

Payment of the required amount must be completed at least one year prior to such Participant's termination of employment on account of retirement or otherwise.

- (3) A Participant may buy back up to 10 years of Governmental and other Related Service. The service which is bought back by a Participant pursuant to this Section 1.7 shall not be taken into account in determining his Normal Retirement Date under Section 3.1 nor his vested interest under Section 7.1(A).
- (4) A Participant's pension attributable to years of Governmental and other Related Service bought back by him shall be treated as vested upon his having completed the vesting requirement in Section 7.1(A).
- (5) Such Participant need not have been a participant in the plan of the other employer, but he must have been holding a position that would have qualified him for participation in the Town Plan if the Town were such other employer.

- (6) Any benefits determined under the Town Plan shall be offset by any benefits to which such Participant is entitled from another employer's plan to the extent any Credited Service is granted under this provision.
- (7) For the purposes of this Section 1.7, "Town Plan" shall mean this Plan for prior service on or after July 1, 1981 and the Retirement Plan for Municipal Employees of the Town for prior service before July 1, 1981.
- (8) A Participant desiring to buy back prior service pursuant to this Section 1.7 shall submit a written request therefor to the Town's Personnel Department, which shall forward such request to the Pension Committee for its approval or disapproval in its sole discretion. Such request must be submitted no later than 30 days after the successful completion of his 6-month probationary period. The decision of the Pension Committee as to such a Participant's request shall be conclusive and binding on such Participant and, if any, his spouse, Joint Annuitant, and Designated Beneficiary. If the Pension Committee approves such a request by a Participant, he shall have 90 days from the date of approval to pay to the Plan the required amount of contributions and Interest, as determined by the Town's Finance Department.
- (9) If a Participant's employment with the Town should terminate for any reason prior to the time he is fully vested under Section 7.1(A), the amount paid by such Participant pursuant to this Section 1.7 shall be returned to him, together with Interest thereon.
 - 1.8 "Deferred Retirement Date" means the date specified in Section 3.2.
- 1.9 "Dependent" means a Participant's child (including an adopted child) who has not attained age 21. A person who is a Dependent under the preceding sentence shall cease to be a Dependent upon attaining age 21.
- 1.10 "Designated Beneficiary" means the beneficiary designated by a Participant, subject to change from time to time by such Participant, on forms provided by the Pension Committee; provided, however, that if there be no Designated Beneficiary at the date of death of the Participant (e.g., due to the earlier death of the Designated Beneficiary or due to failure of the Participant to designate a beneficiary) then the Pension Committee shall, for any death benefit that may be available under the Plan (other than a monthly benefit payable to any Joint Annuitant or surviving spouse), designate a beneficiary from the following list taken in priority order:
 - (A) the Participant's legal spouse, or, if none survives;
- (B) the Participant's lineal descendants (including any children legally adopted) per stirpes; or if none survives;
 - (C) the Participant's parents, share and share alike; or if none survives;
 - (D) the Participant's estate,

subject to Sections 10.2 and 11.8. In the event any amount shall become payable from the Plan to a Designated Beneficiary or the executor or administrator of any deceased person and if, after written notice from the Pension Committee mailed to such person's last known address,

such person or such executor or administrator shall not have presented himself to the Pension Committee within two years after the mailing of such notice, the Pension Committee shall distribute such amount due to such Beneficiary or such executor or administrator among one or more of the spouse and blood relatives of such deceased person designated by the Pension Committee.

- 1.11 "Disabled" means, with respect to a Participant, having a Disability.
- 1.12 "Disability" means, with respect to a Participant, the inability to perform the duties of any occupation for which he is reasonably fitted by reason of training, education or experience.
- 1.13 "Effective Date" means July 1, 2019, as to Participants actively employed on or after that date, except as otherwise provided herein. The Original Effective Date was July 1, 1981.
- 1.14 "Employee" means any person who is employed on a permanent basis by the Town, Library, or Board of Education who is in a non-supervisory position as determined by the Employer, and who is not covered by a collective bargaining agreement, and excluding any teacher or other Employee who is eligible for membership in the Connecticut State Teachers Retirement System or is covered by the pension system established by Special Act No. 430 or its successor. The term "Employee" shall not include a "leased employee," meaning, with respect to the Employer, any person (other than an employee of the Employer) who pursuant to an agreement between the Town and any other person has performed services for the Employer (or for the Employer and related persons determined in accordance with Section 414(n)(6) of the Code) and such services are performed under primary direction or control by the Employer.
 - 1.15 "Employer" means the Town of Westport, Connecticut.
- 1.16 "Enrolled Actuary" means a person designated as an Enrolled Actuary by the Joint Board for Enrollment of Actuaries.
- 1.17 "Governmental And Other Related Service", as used in Section 1.7 means, with respect to a Participant, employment with a municipal, state or federal government, or agency thereof, which is related, by nature of the work and the duties performed, to the work for which such Employee is hired by the Town; and which could reasonably be expected to have helped to prepare such Participant for his present job responsibilities with the Town.
- 1.18 "Interest," when used in connection with a Participant's contributions, means such rate per annum as may from time to time be fixed by the Town. Interest on a Participant's Contributions during a Plan Year accrues from July 1 of the following Plan Year to the first of the month in which the pension payments commence, or death or termination of Service occurs, unless some other applicable date is specified in the Plan.
- 1.19 "Joint Annuitant" means the person designated by a Participant to receive benefits pursuant to Section 5.1(B)(1).
 - 1.20 "Limitation Year" means the period July 1 to June 30.

- 1.21 "Medical Plan" means, collectively, the plan or plans maintained by the Town pursuant to which medical, prescription drug and/or dental benefits are made available to particular Employees, Retirees and their respective spouses and eligible Dependents. If any such plan is referred to herein by name, such reference shall also include any comparable successor plan.
- 1.22 "Normal Retirement" means termination of employment (except by death) of a Participant on the first day of the month coincident with or next following his Normal Retirement Date as set forth in Article 3.
- 1.23 "Participant" means an Employee satisfying the eligibility requirements set forth in Article 2.
- 1.24 "Pension Committee" means the Chairman of the Board of Finance, the First Selectman or First Selectwoman, one Elector appointed by the Board of Selectmen and one Participant of the Plan appointed by the First Selectman or First Selectwoman. The term of appointment of the Elector shall be 3 years.
- 1.25 "Pension Fund" means the trust forming a part of the Plan, as described in Article 9.
- 1.26 "Plan" means the Retirement Plan for Non-Union and Non-Supervisory Employees of the Town of Westport, Connecticut, as amended and restated as set forth herein, or as may be amended from time to time.
 - 1.27 "Plan Year" means July 1 to June 30.
- 1.28 "Qualified Military Service" means any service in the "uniformed services" (as defined in Chapter 43 of Title 38, United States Code) by any individual if such individual is entitled to reemployment rights under such chapter as to such service.
- 1.29 "Retiree" means a Participant who retires on or after his Normal Retirement Date and who is eligible for pension benefits under the Plan.
- 1.30 "Retirement Date" means the date as of which a Participant's Normal or Deferred Retirement occurs under the Plan.
- 1.31 "Retirement Option" means the optional forms of pension as provided in Article 5.
 - 1.32 "RTM" means the Representative Town Meeting of the Town.
 - 1.33 "Secretary" means the Secretary of the Treasury.
- 1.34 "Service" means the period during which a person is in the status of an Employee, including Approved Absence.

- 1.35 "Spouse" means any individual to whom the Participant is legally married under the laws of the state or country in which the marriage was celebrated, without regard to whether such marriage is recognized under the laws of the state or country in which the Participant resides.
- 1.36 "Town" means the Town of Westport in the County of Fairfield, State of Connecticut.
- 1.37 "Trustee" means such person or persons as trustee or trustees of the Pension Fund, as may be appointed by the Pension Committee.
- 1.38 "Vested Participant" means a Participant who has completed at least 10 Years of Service and is vested according to Section 7.1(A) but is not yet eligible for Retirement under Article 3.
- 1.39 "Year of Service" means, with respect to an Employee, a 12-consecutive-month period commencing with such Employee's date of employment and succeeding 12-consecutive-month periods during which such Employee was employed by the Employer. Each completed month of employment shall constitute 1/12 of a Year of Service. In the case of Employees who regularly work 9 months or more but less than 12 months per year, continuous employment for the period of approximately 9 months, which constitutes the entire school year of an Employee's normal work year, shall be deemed to constitute a Year of Service (with partial years to be credited based on complete months at the rate of 1/9 year per complete month). Upon retirement or other termination of employment, a Participant who has accrued at least 60 days of unused sick leave shall be entitled to use such accrual, up to a maximum of 120 days, as an additional fractional Year of Service, as follows:

Accrued Unused Sick Days	Additional Fractional Year of Service
At least 60 days but less than 80 days	1/4
At least 80 days but less than 100 days	1/3
At least 100 days but less than 120 days	5/12
120 or more days	1/2

ARTICLE 2 ELIGIBILITY

2.1 IN GENERAL.

- (A) Each Employee shall become a Participant of the Plan provided he:
 - (1) Works 20 or more hours per week on a consistent basis;
 - (2) Works 9 months or more per year;
- (3) Has completed the probationary period required for his job classification; and
- (4) Is not a participant of another retirement plan to which the Employer contributes.
- (B) Participation in the Plan is mandatory, except that an Employee hired at age 60 or over has the option of not participating.
- (C) Notwithstanding the preceding provisions of this Section 2.1, no Employee who is hired or rehired on or after January 1, 2012 shall become a Participant.

2.2 REEMPLOYMENT FOLLOWING RETIREMENT.

A Retiree receiving a pension under the Plan who again becomes a Participant following his reemployment by the Town shall not have his pension discontinued.

ARTICLE 3 RETIREMENT DATES

3.1 NORMAL RETIREMENT.

A Participant's Normal Retirement Date is the first of the month coincident with or next following:

- (A) Age 55 and completion of at least 10 Years of Service; or
- (B) Completion of 25 Years of Service with no age requirement.

For purposes of this Section 3.1, "Years of Service" do not include any Credited Service bought back pursuant to Section 1.7.

3.2 DEFERRED RETIREMENT.

A Participant may defer retirement beyond his Normal Retirement Date. A Participant's Deferred Retirement Date is the first day of the month following termination of employment after Normal Retirement Date.

ARTICLE 4 PENSION BENEFITS

4.1 NORMAL OR DEFERRED RETIREMENT PENSION.

The monthly pension payable at a Participant's Normal Retirement Date or Deferred Retirement Date shall equal (i) 2% of the Participant's monthly Average Final Compensation multiplied by the number of his completed years and months (each month counting as 1/12 of a year) of Credited Service, including any Credited Service granted under the provisions of Section 1.7, to a maximum of 20 years; plus (ii) 2.25% of the Participant's monthly Average Final Compensation multiplied by the number of his completed years and months (each month counting as 1/12 of a year) of Credited Service in excess of 20; less (iii) the amount of monthly pension benefit that such Participant is entitled to receive from another plan in which he participated, payable at its normal retirement date, to the extent any Credited Service is granted under this Plan for participation in such other plan.

4.2 MAXIMUM PENSION.

In no instance may the sum of (i) a Participant's monthly pension benefit from this Plan, (ii) his monthly pension benefit from any other Town retirement plan and (iii) his monthly pension benefit from another plan (other than a Town retirement plan) if Credited Service is granted under this Plan with respect to such other plan, as provided in Section 1.7, exceed 100% of such Participant's monthly Average Final Compensation.

4.3 MINIMUM PENSION.

Notwithstanding anything to the contrary contained herein, the minimum monthly pension payable at Normal Retirement Date or Deferred Retirement Date under the Plan for a Participant who has completed at least 10 Years of Service shall be \$83.33 per month, or \$1,000 per year.

4.4 COVERAGE UNDER MEDICAL PLAN FOR PARTICIPANTS RETIRING BEFORE NOVEMBER 1, 2005.

Each Participant retiring before November 1, 2005 shall be entitled to continue his coverage under the Medical Plan in effect at the time of his retirement for himself and his spouse and eligible Dependents by paying the total cost of such coverage until July 1 following the attainment of age 49 and thereafter by paying 50% of the cost of such coverage.

4.5 CONTINUED COVERAGE UNDER MEDICAL PLAN FOR PARTICIPANTS RETIRING ON OR AFTER NOVEMBER 1, 2005.

Notwithstanding Section 4.4, the following provisions concerning coverage under the Town's Medical Plan shall apply to all Participants retiring on or after November 1, 2005 and who are immediately entitled to benefits under this Plan, and to the spouses and eligible Dependents of such Participants:

- (A) Each Retiree shall be entitled to continue his coverage for himself and his spouse and eligible Dependents under the Town's Medical Plan in effect at the time of his retirement by paying the total cost of such coverage until the July I following the attainment of age 49. Effective as of July 1 following his attainment of age 49, such Retiree shall pay 40% of the cost of such coverage, as determined based upon the "allocation rate," except that the cost of coverage for any year shall not increase by more than 10% over the cost of coverage for the preceding year.
- (B) In the event of a Retiree's death, his surviving spouse and surviving eligible Dependents shall pay 40% of the cost of coverage for them, as determined based upon the "allocation rate," except that the cost of coverage for any year shall not increase by more than 10% over the cost of coverage for the preceding year. For purposes of this Section 4.5, the term "Retiree" also includes a Participant with respect to whom a death benefit is payable pursuant to Section 7.2(A).
- (C) At such time as such Retiree or the spouse of a Retiree reaches the age of eligibility for Medicare (even if such Retiree or spouse is not eligible for Medicare):
- (1) Such Retiree or spouse shall no longer be able to participate in the Medical Plan, except as provided in Section 4.5(C)(3) and (4), then covering him and, provided that he has applied and been granted coverage under Medicare Part A and Part B, he shall instead be eligible to enroll in the Blue Cross/Blue Shield Century Carve Out Plan. Retirees and spouses shall pay 40% of the cost for such Blue Cross/Blue Shield Century Carve Out Plan, except that the cost of coverage for any year shall not increase by more than 10% over the cost of coverage for the preceding year. The term "Blue Cross/Blue Shield Century Carve Out Plan" includes any other health plan which the Town may offer in its place from time-to-time.
- (2) Such Retiree or spouse shall be responsible for payment of the full premiums for any Medicare coverage, including without limitation Part B coverage and also Part A coverage (unless such Retiree or spouse is entitled to Part A coverage without the payment of premiums).
- (3) Such Retiree or spouse may continue to be covered for the dental benefits under the Town's Medical Plan and the premium which he shall pay for such coverage shall be the amount he would be required to pay under Section 4.5(A) as if he had not reached the age of eligibility for Medicare, except that the cost of coverage for any year shall not increase by more than 10% over the cost of coverage for the preceding year. This Section 4.5(C)(3) shall apply only to a Retiree or spouse who was covered for such dental benefits at the time of such Retiree's retirement.

- (4) Such Retiree or spouse may continue to be covered for the prescription drug benefits under the Town's Medical Plan and the premium which he shall pay for such coverage shall be the amount he would be required to pay under Section 4.5(A) as if he had not reached the age of eligibility for Medicare, except that the cost of coverage for any year shall not increase by more than 10% over the cost of coverage for the preceding year. This Section 4.5(C)(4) shall apply only to a Retiree or spouse who was covered for such prescription drug benefits at the time of such Retiree's retirement.
- (D) A Retiree and the spouse of a Retiree shall be treated separately for purposes of this Section 4.5. For purposes of this Section 4.5, if a person is the spouse of a Retiree at the time of the Retiree's retirement, (i) such person shall continue to be treated as a spouse notwithstanding the termination of the marriage of such Retiree and spouse by reason of the death of such Retiree and (ii) such person shall not continue to be treated as a spouse in the event of the termination of the marriage of such Retiree and spouse by reason of divorce.

4.6 LIMITS ON BENEFITS.

Notwithstanding any other provision of the Plan, the benefits of a Participant hereunder shall be subject to the following limitation: The maximum Annual Benefit payable to a Participant under the Plan, and under any other defined benefit plan sponsored by the Employer (or any entity aggregated with the Employer pursuant to Section 414 of the Code), for any Limitation Year shall not exceed the limitation set forth in Section 415(b) of the Code. For purposes of Sections 415(b)(2)(E)(i) and Section 415(b)(2)(E)(ii) of the Code, the interest rate shall be 5% per annum, compounded annually. The term "Annual Benefit" shall mean a benefit payable annually in the form of a qualified joint and survivor annuity as defined in Section 417(b) of the Code and Treas. Reg. Sec. 1.401(a)-20, Q & A-25 (with no ancillary benefits) under a plan to which employees do not contribute and under which no rollover contributions are made.

ARTICLE 5 FORM OF RETIREMENT PAYMENT AND ELECTION PROCEDURES

5.1 RETIREMENT PAYMENT FORMS.

- (A) The normal form of payment to a Retiree is a standard life annuity, payable monthly for his life. This benefit ceases with the death of the Retiree. Should the Retiree die before receiving benefits equal to his accumulated Employee contributions plus Interest, such remaining amount shall be paid in a lump sum to the Retiree's Designated Beneficiary.
- (B) The following options, under each of which the amount of payment is the Equivalent Actuarial Value of the payment under the normal form, are available:
- (1) <u>Joint and Survivor Life Annuity</u>: An annuity, under which payments are made to the Retiree until death, then continue to a specified Joint Annuitant (in same or reduced amount) until death of such Joint Annuitant. Should the Retiree and the Joint Annuitant both die before receiving benefits in the aggregate equal to the Retiree's accumulated Employee contributions plus Interest, such remaining amount shall be paid in a lump sum to the Retiree's Designated Beneficiary.
- (2) Period Certain (5, 10 or 20 years) and Life Annuity: Payments are guaranteed for the lifetime of the Retiree with the provision that not less than 60, 120, or 240 equal monthly payments shall be made as designated by the Retiree, to him or a Designated Beneficiary. Should both parties die before receiving the guaranteed number of payments, the commuted value of the balance of the guaranteed payments shall be paid in a lump sum amount to the estate of the last survivor.

5.2 ELECTION AND REVOCATION OF OPTIONAL FORMS.

At least 3 months prior to his Retirement Date, a Participant should notify the Pension Committee of his anticipated Retirement Date. Subsequently, the Participant shall be notified of the benefit available under the various optional forms of payment as described in Section 5.1 and shall make an election. A Participant may revoke an election made by him at any time prior to his Retirement Date and make a new election subject to the limitations herein.

5.3 DEATH OF JOINT ANNUITANT.

If a Participant elects a joint and survivor annuity and the Joint Annuitant dies prior to the commencement of benefits, then the Participant shall receive his pension in the form of a life annuity.

5.4 UNCLAIMED BENEFITS.

The Pension Committee having made reasonable attempts to locate any Participant or beneficiary during the 7-year period following the date of first notice, may thereafter direct any benefit that may be due to be canceled.

5.5 DIRECT ROLLOVERS OF ELIGIBLE ROLLOVER DISTRIBUTIONS.

Notwithstanding any other provision of the Plan to the contrary which would otherwise limit a Distributee's election under this Section 5.5, a Distributee may elect, at the time and in the manner permitted by the Plan Administrator, to have any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan specified by the Distributee in a Direct Rollover. For purposes of this Section 5.5:

- (A) "Direct Rollover" means a payment by the Plan to the Eligible Retirement Plan specified by a Distributee.
- (B) "Distributee" means a Participant who is an Employee or former Employee. In addition, (1) such a Participant's spouse or former spouse who is the alternate payee under a "qualified domestic relations order," as defined in Section 414(p) of the Code, and (2) the surviving spouse of a deceased Participant who was an Employee or former Employee, are Distributees with regard to the interest of such spouse or former spouse in the Plan. A distributee includes the Participant's or former Participant's nonspouse designated beneficiary, in which case, the distribution can only be transferred to a traditional or Roth IRA established on behalf of the nonspouse designated beneficiary for the purpose of receiving the distribution.
- (C) "Eligible Retirement Plan" means an individual retirement account described in Section 408(a) of the Code, an individual retirement annuity described in Section 408(b) of the Code, an annuity plan described in Section 403(a) of the Code, or a qualified trust described in Section 401(a) of the Code, which accepts a Distributee's Eligible Rollover Distribution. However, in the case of an Eligible Rollover Distribution to a Distributee who is surviving spouse, an "Eligible Retirement Plan" means an individual retirement account or individual retirement annuity. The term "Eligible Retirement Plan" shall also include (i) an annuity contract described in Section 403(b) of the Code, (ii) an eligible plan which is maintained under Section 457(b) of the Code and which is maintained by a state or political subdivision of a state or instrumentality of a state and which agrees to separately account for amounts transferred to such plan from this Plan and (iii) a Roth IRA described in Code Section 408A. The definition of "Eligible Retirement Plan" shall apply in the case of a distribution to a surviving spouse of a Participant or to a spouse or former spouse of a Participant who is an alternate payee under a "qualified domestic relations order," as defined in Section 414(p) of the Code.
- (D) "Eligible Rollover Distribution" means any distribution of all or any portion of the balance to the credit of the Distributee under the Plan, except that an Eligible Rollover Distribution shall not include: (1) any distribution which is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Distributee or the joint lives (or joint life expectancies) of the Distributee and the Distributee's Designated Beneficiary, or for a specified period of 10 years or more; (2) any distribution to the extent such distribution is required under Section 401(a)(9) of the Code; and (3) the portion of any distribution which is not includible in gross income. The enumeration in the preceding sentence of any form of payment shall not imply that any person has the right to receive benefits under the Plan in such form unless otherwise specifically provided under the Plan. The term "Eligible Rollover Distribution" shall also include a direct trustee-to-trustee

transfer of all or any portion of a distribution from the Plan from the benefit of a deceased Participant to an individual retirement account described in Code Section 408(a), or an individual retirement annuity described in Code Section 408(b) (other than an endowment contract), established for the purposes of receiving such distribution or portion thereof on behalf of an individual who is a Designated Beneficiary, within the meaning of Section 8.7(B), who is not the surviving spouse of such Participant.

If, pursuant to Section 7.1, a distribution of more than \$1,000 is payable to a Participant whose employment has terminated, and such Participant does not elect to receive such distribution directly or, pursuant to this Section 5.5, to have such distribution paid to an Eligible Retirement Plan in a direct rollover, the Pension Committee shall cause such distribution to be paid in a direct rollover to an individual retirement account described in Section 408(a) of the Code or to an individual retirement annuity described in Section 408(b) of the Code.

5.6 QUALIFIED MILITARY SERVICE.

- (A) Notwithstanding any provision of this Plan to the contrary, contributions, benefits and service credit as to Qualified Military Service shall be provided in accordance with Section 414(u) of the Code.
- (B) Effective as to deaths occurring on or after January 1, 2007, in the case of a Participant who dies while performing Qualified Military Service, his survivors shall be entitled to any additional benefits (other than benefit accruals related to the period of Qualified Military Service) that may be provided under the Plan had he then resumed and terminated employment on account of death.

ARTICLE 6 DISABILITY BENEFITS

6.1 ELIGIBILITY AND DURATION.

A Participant who becomes Disabled and remains continuously Disabled for a period of at least 6 months shall be eligible for a Disability benefit. The Disability benefit shall commence after 6 months of Disability and be payable monthly during the time the Participant is Disabled and until the latest of the following shall occur:

- (A) The Participant's attainment of age 65;
- (B) The Participant's attainment of his Normal Retirement Date; or
- (C) 5 years from the commencement of Disability benefits.

During the period of Disability, Credited Service for the Disabled Participant shall continue to accrue and no contributions under Article 9 shall be required from him; provided, however, that during such period the benefit payable to him pursuant to Section 6.2 shall not increase on account of such Credited Service, and instead such Credited Service during the period of Disability shall be taken into account in determining the amount of his benefit payable under the Plan following the termination of his Disability benefit.

6.2 AMOUNT OF BENEFIT.

The amount of the monthly Disability benefit shall be the greater of:

- (A) 50% of the Participant's monthly Compensation immediately prior to the onset of Disability; or
- (B) The amount of the pension computed under Section 4.1 based upon such Participant's Credited Service and Average Final Compensation, as if such pension were payable beginning at the onset of Disability,

reduced by any amounts paid or payable to the Participant from the following sources:

- (C) Disability benefits under any plan sponsored by the Employer;
- (D) Any wages or salary paid to the Participant attributable to periods of time when the Participant is receiving benefits hereunder;
- (E) Retirement benefits under this or any other plan to which the Employer contributes; and
 - (F) Benefits under the Social Security or Railroad Retirement Acts.

6.3 DETERMINATION OF DISABILITY BENEFITS BY PENSION COMMITTEE.

The Pension Committee shall utilize such outside professional counsel as it deems necessary to assist it in determining a Participant's initial and continuing eligibility for Disability benefits. The decision of the Pension Committee shall be final.

6.4 ELECTION AS TO FORM OF PAYMENT.

Prior to the commencement of Disability benefits to a Participant, he shall elect a form of payment option under Section 5.1 as if he were retiring and receiving a pension pursuant to Section 4.1. The election which he shall make shall also apply to any normal retirement benefits payable under the Plan after the cessation of his Disability benefits if he does not return to active employment with the Employer following such cessation of his Disability benefits; accordingly, the form of payment option elected by such a Participant under Section 5.1 may not thereafter be changed for any reason.

ARTICLE 7 VESTED BENEFITS AND DEATH BENEFITS

7.1 VESTED BENEFITS.

The following provisions shall apply to a Participant who terminates employment for any reason other than death or Disability prior to his Normal Retirement Date:

- (A) Such Participant shall be vested in his accrued pension determined in accordance with Section 4.1 after the completion of 10 Years of Service. Credited Service bought back pursuant to Section 1.7(B) shall not be taken into account in determining whether a Participant has completed 10 Years of Service for vesting purposes.
- (B) A Vested Participant shall commence receiving his accrued pension determined in accordance with Section 4.1 at his Normal Retirement Date. Alternatively, a Vested Participant may elect to receive, upon termination of employment, his contributions with Interest in full satisfaction of his claims against the Plan.
- (C) A Participant who terminates employment for any reason other than death or Disability prior to his Normal Retirement Date and is not vested in any portion of his accrued pension benefit pursuant to Section 7.1(A), shall receive his contributions with Interest in full satisfaction of his claims against the Plan.

7.2 DEATH BENEFITS PRIOR TO RETIREMENT.

- (A) The death benefit of a Participant, whether or not actively employed by the Town, who dies prior to his Normal Retirement Date but on or after the date he had completed 10 or more years of Credited Service, excluding Credited Service bought back pursuant to Section 1.7(B) and who is survived by a spouse, or eligible Dependent or Dependents, shall be as follows:
- (1) If the Participant is survived by a spouse, the survivor shall be entitled to receive an annual benefit, payable in equal monthly installments, equal to the greater of Clauses (i) and (ii) below:
- (a) 50% of the Participant's Average Final Compensation as of the date of his death. In the event the Participant has not completed 25 years of Credited Service to the date of his death, such benefit shall be reduced by 1-1/2% for each year of Credited Service less than 25 years (adjusted for completed months in excess of completed years).
- (b) The monthly pension computed under Section 4.1 based on the Participant's years of Credited Service and monthly Average Final Compensation as of the date of his death.

In the event the surviving spouse is more than 5 years younger than the deceased Participant, the benefits determined under this Section 7.2(A)(1) shall be reduced by 1/6 of one 1% for each month his age is more than 5 years younger than the Participant's age.

This death benefit shall be payable to the surviving spouse for life.

(2) If the Participant is survived by one or more eligible Dependents but no spouse, the death benefit determined under Section 7.2(A)(1) (one benefit per family) shall be payable monthly pro rata to those of such eligible Dependents who are under 21 years of age until the youngest eligible Dependent reaches age 21.

Notwithstanding the preceding provisions of this Section 7.2(A), no such death benefit shall be less than the Actuarial Equivalent of such Participant's contributions, with Interest thereon to the date of such Participant's death.

- (B) The death benefit of a Participant, whether or not actively employed by the Town, who dies prior to his Normal Retirement Date and
- (1) On or after the date he had completed 10 or more years of Credited Service, excluding Credited Service bought back pursuant to Section 1.7(B), but who is not survived by a spouse, or eligible Dependent or Dependents; or
- (2) Before the date he had completed 10 or more years of Credited Service, excluding Credited Service bought back pursuant to Section 1.7(B),

shall be an amount equal to such Participant's contributions, with Interest thereon to the date of such Participant's death. Such death benefit shall be paid to such Participant's Designated Beneficiary.

7.3 DEATH BENEFITS AFTER RETIREMENT.

No further benefits shall be payable on behalf of a Retiree after his death, except as may be provided under an optional benefit form elected in accordance with Article 5.

ARTICLE 8 REQUIRED DISTRIBUTIONS

8.1 GENERAL RULES.

The provisions of this Article 8 shall take precedence over any inconsistent provisions of the Plan.

All distributions required under this Article 8 shall be determined and made in accordance with the Treasury regulations under Code Section 401(a)(9).

8.2 TIME AND MANNER OF DISTRIBUTION.

- (A) A Participant's entire vested interest in the Plan, as determined pursuant to Section 7.1(A), shall be distributed, or begin to be distributed, to him no later than his Required Beginning Date.
- (B) If a Participant dies before distributions begin, his entire vested interest in the Plan shall be distributed, or begin to be distributed, no later than as follows:
- (1) If such Participant's surviving spouse is his sole Designated Beneficiary, then distributions to such surviving spouse shall begin by December 31 of the calendar year immediately following the calendar year in which such Participant died, or by December 31 of the calendar year in which the Participant would have attained age 70-1/2, if later.
- (2) If such Participant's surviving spouse is not his sole Designated Beneficiary, then distributions to his Designated Beneficiary shall begin by December 31 of the calendar year immediately following the calendar year in which such Participant died.
- (3) If there is no Designated Beneficiary as of September 30 of the year following the year of such Participant's death, such Participant's entire vested interest in the Plan shall be distributed by December 31 of the calendar year containing the fifth anniversary of such Participant's death.
- (4) If such Participant's surviving spouse is his sole Designated Beneficiary and such surviving spouse dies after such Participant but before distributions to such surviving spouse begin, this Section 8.2(B), other than Section 8.2(B)(1), shall apply as if the surviving spouse were the Participant.

For purposes of this Section 8.2(B) and Section 8.5, distributions shall be considered to begin on a Participant's Required Beginning Date, or if Section 8.2(B)(4) applies, the date distributions are required to begin to the surviving spouse under Section 8.2(B)(1). If annuity payments irrevocably commence to a Participant before his Required Beginning Date, or to his surviving spouse before the date distributions are required to begin to such surviving spouse under Section 8.2(B)(1), the date distributions shall be considered to begin is the date distributions actually commence.

Unless a Participant's vested interest in the Plan is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before his Required Beginning Date, as of the first Distribution Calendar Year distributions shall be made in accordance with Sections 8.3, 8.4 and 8.5. If a Participant's interest is distributed in the form of an annuity purchased from an insurance company, distributions thereunder shall be made in accordance with the requirements of Code Section 401(a)(9) and the Treasury regulations.

8.3 DETERMINATION OF AMOUNT TO BE DISTRIBUTED EACH YEAR.

- (A) If a Participant's interest is paid in the form of annuity distributions under the Plan, such distributions shall satisfy the following requirements:
- (1) The annuity distributions shall be paid in periodic payments made at intervals not longer than one year;
- (2) The distribution period shall be over a life (or lives) or over a period certain not longer than the period described in Sections 8.4 or 8.5;
- . (3) Once payments have begun over a period certain, the period certain shall not be changed even if the period certain is shorter than the maximum permitted; and
- (4) Payments shall be non-increasing, or shall increase only to pay increased benefits that result from a Plan amendment.
- (B) The amount which must be distributed on or before a Participant's Required Beginning Date, or if such Participant dies before distributions begin, the date distributions are required to begin under Sections 8.2(B)(1) or 8.2(B)(2), is the payment which is required for one Payment Interval. The second payment need not be made until the end of the next Payment Interval even if that Payment Interval ends in the next calendar year. All of a Participant's benefit accruals as of the last day of the first Distribution Calendar Year shall be included in the calculation of the amount of the annuity payments for Payment Intervals ending on or after his Required Beginning Date.
- (C) Any additional benefits accruing to a Participant in a calendar year after the first Distribution Calendar Year shall be distributed beginning with the first Payment Interval ending in the calendar year immediately following the calendar year in which such amount accrues.
- 8.4 REQUIREMENTS FOR ANNUITY DISTRIBUTIONS COMMENCING DURING A PARTICIPANT'S LIFETIME.

If a Participant's vested interest in the Plan is being distributed in the form of a joint and survivor life annuity for the joint lives of the Participant and a non-spouse Joint Annuitant, annuity payments to be made on or after the Participant's Required Beginning Date to the Designated Beneficiary after the Participant's death must not at any time exceed the applicable percentage of the annuity payment for such period that would have been payable to the Participant using the table set forth in Q&A-2 of Treasury Regulation Section 1.401(a)(9)-6T.

- 8.5 REQUIREMENTS FOR MINIMUM DISTRIBUTIONS WHERE PARTICIPANT DIES BEFORE DATE DISTRIBUTIONS BEGIN.
- (A) If a Participant dies before the date distribution of his vested interest in the Plan begins and there is a Designated Beneficiary, such Participant's entire vested interest shall be distributed, beginning no later than the time described in Sections 8.2(b)(1) or 8.2(b)(2), over the life of the Designated Beneficiary or over a period certain not exceeding:
- (1) Unless the Annuity Starting Date is before the first distribution calendar year, the life expectancy of such Designated Beneficiary determined using his age as of his birthday in the calendar year immediately following the calendar year of such Participant's death; or
- (2) If the Annuity Starting Date is before the first Distribution Calendar Year, the life expectancy of such Designated Beneficiary determined using his age as of his birthday in the calendar year which contains the Annuity Starting Date.
- (B) If a Participant dies before the date distribution of his entire vested interest in the Plan begins and there is no Designated Beneficiary as of September 30 of the year following the year of such Participant's death, distribution of such Participant's entire vested interest in the Plan shall be completed by December 31 of the calendar year containing the fifth anniversary of such Participant's death.
- (C) If a Participant dies before the date distribution of his entire vested interest in the Plan begins and his surviving spouse is the Participant's sole Designated Beneficiary, and such surviving spouse dies before distributions to such surviving spouse begin, this Section 8.5 shall apply as if such surviving spouse were the Participant, except that the time by which distributions must begin shall be determined without regard to Section 8.2(B)(1).

8.6 LIFE EXPECTANCY.

For purposes of this Article 8, life expectancy shall be computed by use of the Single Life Table in Treasury Regulation Section 1.401(a)(9)-9.

8.7 DEFINITIONS.

For purposes of Sections 8.1 to 8.6:

- (A) "Annuity Starting Date" means: (1) The first day of the first period for which an amount is payable as an annuity or other distribution under the Plan; or (2) In the case of a benefit not payable in the form of an annuity, the first day on which all events have occurred which entitle the recipient to such benefit.
- (B) "Designated Beneficiary" means the individual who is designated as the beneficiary under Section 1.6 of the Plan and is the "designated beneficiary" under Code Section 401(a)(9) and Treasury Regulation Section 1.401(a)(9)-4, Q&A-1.

- (C) "Distribution Calendar Year" means a calendar year for which a minimum distribution is required under this Article 8. For distributions beginning before a Participant's death, the first Distribution Calendar Year shall be the calendar year immediately preceding the calendar year which contains his Required Beginning Date. For distributions beginning after a Participant's death, the first Distribution Calendar Year is the calendar year in which distributions are required to begin pursuant to Section 8.2(b).
- (D) "Payment Intervals" means the periods for which payments are received, such as bi-monthly, monthly, semi-annually or annually.
- (E) "Required Beginning Date" means, with respect to a Participant, April 1 of the calendar year following the later of (i) the calendar year in which such Participant attains age 70-1/2 or (ii) the calendar year in which such Participant retires. The benefit of a Participant which does not commence by April 1 of the calendar year following the calendar year in which he attains age 70-1/2 shall be actuarially increased for the period beginning April 1 of the calendar year following the calendar year in which he attains age 70-1/2 and ending on the date after retirement that his benefits commence in an amount sufficient to satisfy the requirements of Section 401(a)(9) of the Code. The amount of actuarial increase payable as of the end of such period shall equal the Actuarial Equivalent of such Participant's pension benefit which would have been payable as of the date such actuarial increase must commence plus the Actuarial Equivalent of additional benefits accrued after such date, and reduced by the Actuarial Equivalent of any benefits paid after such date.

ARTICLE 9 CONTRIBUTIONS AND PENSION FUND

9.1 PARTICIPANT CONTRIBUTIONS.

- (A) Each Employee eligible to participate shall be required as a condition of employment to make contributions to the Plan. Contributions so required are 4% of Compensation. Notwithstanding the preceding sentence, no contributions shall be required with respect to the time a Participant was a probationary Employee.
- (B) In accordance with Section 414(h)(2) of the Code, the Employer shall "pick up" the contributions required of Participants hereunder, such that the contributions so picked up shall be considered contributions of the Employer rather than Participant contributions under the Code. Although such contributions are designated as Participant contributions, they shall be treated as being paid by the Employer in lieu of contributions by Participants. No Participant shall have the option of choosing to receive the contributed amounts directly instead of having them paid by the Employer to the Plan.

9.2 EMPLOYER CONTRIBUTIONS.

The Employer shall make such contributions to the Fund for each fiscal year to insure sufficient funds in the Plan to pay all benefits required to be paid by the Plan in that fiscal year.

9.3 PENSION FUND.

Contributions of Participants and the Employer together with any transferred from any other plan or paid into the Plan in accordance with Section 1.7 shall be placed in a trust, referred to hereunder as the "Pension Fund," to be administered and invested by the Pension Committee or its designated agents in accordance with the terms and provisions of the Plan.

9.4 RESTORATION OF FORFEITURE.

The accrued pension attributable to a terminated Participant shall be restored if such Participant:

- (A) Resumes employment with the Employer before the expiration of:
 - (1) In the case of layoff, 5 years; or
 - (2) In the case of Approved Absence, one year; or
- (3) In the case of termination due to other non-voluntary causes, or to voluntary causes, 5 years from the date of his termination; and
- (B) Repays to the Plan, within the first 12 months following reemployment, or within the first 12 months of a revision of this Section 9.4 which would permit, for the first time, restoration of accrued pension to a current employee, the full amount of the total distributions

received pursuant to Section 7.1(B), together with Interest from the date of the distribution to such Participant, to the date of repayment to the Plan by him.

ARTICLE 10 ALIENATION, FACILITY OF PAYMENT AND REEMPLOYMENT

10.1 ALIENATION.

Except as may be contrary to the laws of the United States or any state having jurisdiction in the premises:

- (A) No pension or benefit payable at any time under the Plan shall be subject in any manner to alienation, sale, transfer, assignment, pledge, attachment or encumbrance of any kind. Any attempt to alienate, sell, transfer, assign, pledge or otherwise encumber any such pension or benefit, whether presently or thereafter payable, shall be void. No pension or benefit, nor the Plan nor Fund shall, in any manner, be liable for or subject to the debts or liability of any Participant or Retiree or of any Joint Annuitant, surviving spouse or Designated Beneficiary.
- (B) If any Participant, Retiree, Joint Annuitant, surviving spouse or Designated Beneficiary shall attempt to, or shall, alienate, sell, transfer, assign, pledge, or otherwise encumber his benefits under the Plan or any part thereof, or if by reason of his bankruptcy or other event happening at any time, such benefit would devolve upon anyone else or would not be enjoyed by such Participant, Retiree or Joint Annuitant, surviving spouse or Designated Beneficiary, then the Pension Committee, in its sole discretion may terminate his interest in any such benefit and hold or apply it to or for the benefit of such person, his spouse or children or any of them, in such manner as the Pension Committee, in its sole discretion, may deem proper.

10.2 FACILITY OF PAYMENT.

If the Pension Committee finds that any person to whom a pension or benefit is payable under the Plan is adjudged incompetent or is a minor, then any payment due shall be made payable to the duly appointed guardian, committee or other court appointed representative upon receipt of a valid release. Any such payment shall be a valid and complete discharge of any liability under the Plan in respect of the amount of pension or benefit so paid.

10.3 SUSPENSION OF BENEFITS.

No Participant shall be eligible for any pension benefit payment under this Plan while in the employ of the Employer.

ARTICLE 11 ADMINISTRATION OF PLAN

11.1 PENSION COMMITTEE.

- (A) The general management and administration of the Plan shall be the responsibility of the Pension Committee.
- (B) No person shall act as a member of the Pension Committee unless notice of his appointment has been given in writing by the appointing party to the Pension Committee and the other party.
- (C) The Employer may at any time remove for cause a member of the Pension Committee appointed by it, by giving written notice of such action to the Pension Committee and the other party.
- (D) Any member of the Pension Committee may resign by written notification of his resignation to the Pension Committee.

11.2 MEETINGS OF PENSION COMMITTEE.

- (A) The Pension Committee shall meet at such time and for such periods for the transaction of necessary business as they may decide. All such meetings shall be at the determination of the Pension Committee.
- (B) To constitute a quorum for the transaction of business, the attendance of a majority of the members of the Pension Committee shall be required. Decisions of the Pension Committee shall be made only by the concurring vote of a majority of all the members present.

11.3 COMPENSATION OF PENSION COMMITTEE

The members of the Pension Committee shall receive no compensation from the Pension Fund for any services performed hereunder, but shall be entitled to reimbursement for any reasonable expenses incurred in the performance of their duties.

11.4 AUTHORITY, POWERS AND DUTIES OF PENSION COMMITTEE.

- (A) The Pension Committee shall have such powers as are necessary and proper for the administration of the Plan including, but not limited to the following:
- (1) To promulgate and establish the rules, regulations and procedures dealing with applications for benefits, determination of eligibility, calculation of benefit amounts and authorization of benefit payments;
- (2) To interpret and apply the various provisions of the Plan, which shall be final and binding on Participants, Retirees, Joint Annuitants, surviving spouses and Designated Beneficiaries;

- (3) To request of the Town to retain (i) an Actuary for purposes of examining into the actuarial soundness of the Plan, assisting the parties in the establishment of pension benefits, and aid in determining specific questions of individual benefits; (ii) legal counsel to advise the Pension Committee concerning legal issues arising under the Plan; and (iii) investment advisors in connection with the investment of the assets of the Pension Fund;
- (4) To request direction from the First Selectman or First Selectman, and on receipt of such direction in writing, the Pension Committee shall be entitled to rely and act thereon;
- (5) To pay or authorize payment from the Pension Fund all reasonable expenses of administering the Plan, including, but not limited to, all expenses which may be incurred in connection with the establishment of the Plan, the purchase or lease of such office space, materials, supplies and equipment, and the employment of such administrative, legal, expert and clerical assistance as the Pension Committee, in its discretion, finds necessary or appropriate in the performance of its duties;
- (6) To delegate any ministerial powers or duties to any agent or employee engaged by the Pension Committee, or to any employee of the Town, or to any one or more of the members of the Pension Committee;
- (7) To issue to any Participant upon his request a statement of his standing showing the amount of his contributions, Credited Service and such other pertinent data as the Pension Committee deems proper;
- (8) To hear and determine any claims or complaints relating to the administration of the Plan pursuant to Section 11.7;
- (9) To advise and inform the Employer as to the details of the administration of the Plan:
 - (10) To monitor the performance of the Pension Fund; and
- (11) To issue directions to its agents as appointed by the Pension Committee pursuant to Section 9.3 as to the making of retirement benefit payments, refunds of contributions, and payment of the expense of administration of the Plan.
- (B) The Pension Committee shall maintain accounts showing the fiscal transactions of the Plan, and shall keep in convenient form such data as may be necessary for determination of benefits and actuarial valuations of the assets and liabilities of the Plan;
- (C) After considering the recommendations of the Actuary, the Pension Committee shall, from time to time, adopt actuarial tables to be used as the basis for all actuarial calculations. The Actuary shall, as an aid to the Pension Committee in adopting tables and in recommending the contributions payable by the Town to the Plan, make periodic actuarial valuations of the assets and liabilities of the Plan and shall certify to the Pension Committee the recommended tables and rates of contributions.

11.5 MODIFICATION OF PLAN.

The Pension Committee shall have no power to change or modify any provisions of the Plan, except as provided in Section 11.4(A)(4).

11.6 PROTECTION OF PENSION COMMITTEE.

- The Pension Committee and each individual member thereof shall be fully protected in acting upon any instrument, certificate, or paper believed by them to be genuine and to be signed or presented by the proper person or persons, and shall be under no duty to make any investigation or inquiry as to any statement contained in any such writing, but may accept the same as conclusive evidence of the truth and accuracy of the statements therein contained. The Pension Committee shall be entitled to rely upon all tables, valuations, certificates and reports furnished by the Actuary; upon all certificates and reports made by any duly appointed advisors acting in the areas of their respective professional expertise and upon all opinions, advises, and certifications given by any duly appointed legal counsel. The Pension Committee and each member thereof shall be fully protected, to the extent permitted by law, against any action taken in good faith in reliance upon any such tables, valuations, certificates, reports, opinions or advises. All actions so taken shall be conclusive upon each of them and upon all persons having any interest under the Plan. No member of the Pension Committee shall be personally liable by virtue of any instrument executed by him or on his behalf as a member of the Pension Committee, or for any mistake or judgment made by himself or any other member thereof or for any neglect, omission or wrongdoing of any other member to the extent permitted by law. Each member of the Pension Committee shall be indemnified by the Plan against expenses reasonably incurred by him in connection with any action to which he may be a party by reason of his membership on the Pension Committee except in relation to matters as to which he shall be adjudged in such action to be liable for gross negligence or willful misconduct in the performance of his duty as a member of the Pension Committee. The foregoing right of indemnification shall be in addition to any other rights to which the Pension Committee may be entitled as a matter of law.
- (B) The Pension Committee may from time to time consult with legal counsel and shall be fully protected in acting and relying upon advice of such counsel to the extent permitted by law.
- (C) The cost and expenses of any action, suit or proceeding brought by or against the Pension Committee or any of the members thereof (including counsel fees) may be paid from the Pension Fund, except in relation to matters as to which it shall be adjudged in such action, suit or proceeding, that such member of the Pension Committee was acting in bad faith in the performance of his duties hereunder.
- (D) The Pension Committee shall not be bound by any notice, direction, requisition, advice or request, unless and until it shall have been received by the Pension Committee.
- (E) The Pension Committee may seek judicial protection by any action or proceeding it may deem necessary to settle its accounts, or to obtain a judicial determination or

declaratory judgment as to any question of construction of the Plan or instructions as to any action thereunder.

- (F) No person, partnership, corporation or association dealing with the Pension Committee shall be obliged to see to the application of any funds, securities or other property paid or delivered to the Pension Committee as a purchase price or otherwise, or to see that the terms of the Plan have been complied with, or be obliged to inquire into the authority of the Pension Committee, or the necessity or expediency of any act of the Pension Committee, and every instrument effected by the Pension Committee shall be conclusive in favor of any person, partnership, corporation or association relying thereon that:
- (1) At the time of the delivery of said instrument the Plan was in full force and effect; and
- (2) Said instrument was effected in accordance with the terms and conditions of the Plan; and
- (3) The Pension Committee was duly authorized and empowered to execute such instrument.

11.7 REVIEW PROCEDURES OF PENSION COMMITTEE.

- (A) Upon the Pension Committee's receipt of a written request for benefits under the Plan as filed by a Participant (or claimant who is a surviving spouse, Joint Annuitant or Designated Beneficiary) on a form furnished for that purpose, the Pension Committee shall act thereon in accordance with the following:
- (1) In the event the Pension Committee approves the request, the Participant (or other claimant) shall be so notified within a period not to exceed 90 days from the date of the Participant's (or claimant's) filing of the request; or
- (2) In the event the Pension Committee denies the request, in whole or in part, the Participant (or other claimant) shall be so notified by the Pension Committee within a period not to exceed 90 days from the date of the filing of the request by the Participant (or other claimant).
- (B) The notice of denial by the Pension Committee shall set forth the reasons for the denial of the request for benefits, citing:
 - (1) The pertinent provisions of the Plan;
- (2) Any interpretation or rule relied upon by the Pension Committee in making its decision;
- (3) A description of any additional material or reason required in connection with the request and the reasons therefor;

- (4) The right of the Participant (or other claimant) to inspect Plan documents pertinent to the request; and/or
- (5) A description of the steps which may be taken by the Participant (or other claimant) to obtain a review of the denial.

11.8 FINAL AND BINDING DECISIONS OF PENSION COMMITTEE.

All decisions of the Pension Committee, including all those made in the interpretation and administration of the Plan, shall be conclusive, final and binding upon all Participants, Retirees, Joint Annuitants, surviving spouses and Designated Beneficiaries. No Participant, Retiree or other person shall have or acquire any right, title or interest in or to the Pension Fund or any portion of the Pension Fund, except by the actual payment or distribution of a portion of the Pension Fund to him under the provisions of the Plan. The determination of the Pension Committee as to the identity of the proper payee of any benefit under the Plan and the amount of such benefit properly payable shall be conclusive, and payment in accordance with such determination shall constitute a complete discharge of all obligations on account of such benefit. Any action to stay, confirm, set aside or modify a decision hereunder shall be brought in the courts of the State of Connecticut.

ARTICLE 12 RETIREE MEDICAL EXPENSES

12.1 EFFECTIVE DATE.

This Article 12 shall apply to Medical Expenses incurred on or after the date specified by resolution of the Pension Committee for implementation of this Article 12.

12.2 PAYMENT OF RETIREE MEDICAL EXPENSES.

The Plan may provide for the payment of any or all Medical Expenses of Retirees and their spouses and eligible Dependents in accordance with Code Section 401(h). Payments will occur within a reasonable time after a Claimant has submitted a claim for payment under the Medical Plan.

12.3 SEPARATE ACCOUNT.

A separate account shall be established and maintained under the Plan with respect to contributions to fund Medical Expenses under the arrangement set forth in this Article 12. Such contributions shall come from the Employer and from Retirees and spouses and eligible Dependents. Such separate account shall be for recordkeeping purposes only, and the Trust Funds allocated to such account need not be separately invested. The Employer's contributions to such separate account shall be reasonable and ascertainable and at the time a contribution to the Plan is made by the Employer, the Employer shall designate that portion of such contribution which is allocable to the funding of Medical Expenses. The aggregate actual contributions to the Plan for Medical Expenses shall not exceed 25% of the total actual contributions to the Plan (other than contributions to fund past service credits) after the date this Article 12 is implemented. It shall be impossible, at any time prior to the satisfaction of all liabilities under the Plan to provide Medical Expenses, for any part of the principal or income of such separate account to be (within the taxable year or thereafter) used for, or diverted to, any purpose other than the providing of Medical Expenses. Notwithstanding the provisions of Section 401(h)(2) of the Code, upon the satisfaction of all liabilities under the Plan to provide Medical Expenses, any amount remaining in such separate account shall be returned to the Employer.

12.4 KEY EMPLOYEE.

In the case of a Retiree who is a "key employee," within the meaning of Section 416(i) of the Code, at any time during the Plan Year or any preceding Plan Year during which contributions were made by the Employer on behalf of him or his spouse and eligible Dependents, if any, a separate account shall be established and maintained for Medical Expenses payable to him (and his spouse and eligible Dependents, if any) and such benefits (to the extent attributable to Plan Years beginning after March 31, 1984, for which such Retiree is a "key employee") shall only be payable to or on behalf of such Retiree (and his spouse and eligible Dependents, if any) from such separate account.

12.5 CO-ORDINATION WITH OTHER COVERAGE.

If Medical Expenses are paid from other sources as well as from the Plan, the benefits payable from the Plan shall be paid before any other sources are used.

12.6 USE OF FORFEITURES.

In the event an individual's interest in such separate account shall be forfeited prior to the termination of the Plan, an amount equal to the amount of such forfeiture shall be applied as soon as possible to reduce the Employer's contributions to the Plan to fund Medical Expenses.

12.7 EXCESS PENSION ASSETS.

There may be transferred to the separate account referred in Section 12.3 "excess pension assets" of the Plan, within the meaning of Section 420(e)(2) of the Code, subject to the following provisions:

- (A) Only one transfer may be made in a taxable year of the Employer.
- (B) The amount transferred shall not exceed the amount which is reasonably estimated to be the amount the Employer will pay out (whether directly or through reimbursement) of such separate account during the taxable year of the transfer for "qualified current retiree health liabilities," within the meaning of Code Section 420(e)(1).
 - (C) No such transfer shall be made after December 31, 2013.
- (D) Any assets transferred, and any income allocable to such assets, shall be used only to pay "qualified current retiree health liabilities" for the taxable year of transfer.
- (E) Any amounts transferred to such separate account (and income attributable to such amounts) which are not used to pay "qualified current retiree health liabilities" shall be transferred back to the defined benefit portion of the Plan.
- (F) Amounts paid out of such separate account shall be treated as paid first out of transferred assets and income attributable to such assets.
- (G) The accrued pension benefits for Participants and Beneficiaries of the Plan shall become nonforfeitable as if the Plan had terminated immediately prior to the transfer (or in the case of a Participant who separated during the one-year period ending on the date of transfer immediately before such separation).
- (H) A transfer will be permitted only if the Medical Plan provides that the "applicable employer cost" for each taxable year during the "cost maintenance period" shall not be less than the higher of the "applicable employer costs" for each of the two taxable years immediately preceding the taxable year of the "qualified transfer" within the meaning of Code Section 420(b)(1). For purposes of the preceding sentence:

- (1) The term "applicable employer cost" means, with respect to any taxable year, the amount determined by dividing
- (a) The "qualified current retiree health liabilities," within the meaning of Code Section 420(e)(1)(A), of the Employer for such taxable year determined (I) without regard to any reduction under Code Section 420 (e)(1)(B), and (II) in the case of a taxable year in which there was no "qualified transfer," in the same manner as if there had been such a transfer at the end of the taxable year,

by

- (b) The number of individuals to whom coverage for "applicable health benefits," within the meaning of Code Section 420(e)(1)(C), was provided during such taxable year.
- (2) The term "cost maintenance period" means the period of five taxable years beginning with the taxable year in which the "qualified transfer" occurs. If a taxable year is in two or more overlapping "cost maintenance periods," the preceding sentence shall be applied by taking into account the highest "applicable employer cost" required to be taken into account for purposes of the first sentence of Section 12.7(h) for such taxable year.
- (I) The requirements of Code Section 420(c)(3) shall be satisfied separately with respect to individuals eligible for benefits under Title XVIII of the Social Security Act at any time during the taxable year and with respect to individuals not so eligible.
- (J) Transferred assets may not be used to provide Medical Expenses for "key employees" and their spouses and eligible Dependents, if any.

12.8 DEFINITIONS.

As used in this Article 12, the following terms shall have the meanings indicated:

- (A) "Claimant" means a Retiree, or his spouse or eligible Dependent child, who has submitted a claim for benefits under the Medical Plan.
- (B) "Medical Expense" means an expense which is payable under the Medical Plan and which is an expense for "medical care" under Code Section 213(d)(1).
- (C) "Medical Plan" means the plan or plans maintained by the Town, pursuant to which Retirees and their spouses and eligible Dependents receive medical, prescription drug and dental benefits.

12.9 ANNUAL LIMITATIONS.

For any Limitation Year, the Annual Additions on behalf of any Participant shall not exceed, in the aggregate, the lesser of (i) 100% of such Participant's Compensation for such Limitation Year; or (ii) \$40,000, subject to cost-of-living adjustments under Section

- 415(d)(1)(C) of the Code. The term "Annual Addition" shall mean, for purposes of this Section 12.9, the sum of the following:
- (A) Employer contributions allocable to such Participant for such Limitation Year under any qualified defined contribution plan maintained by the Employer;
- (B) Forfeitures, if any, allocable to such person for such Limitation Year under any qualified defined contribution plan maintained by the Employer;
- (C) Such person's voluntary non-deductible contributions under any other qualified plan of the Employer for such Limitation Year;
- (D) Amounts allocated, after March 31, 1984, to an individual medical account, as defined in Section 415(1)(2) of the Code, which is part of a pension or annuity plan maintained by the Employer; and
- (E) Amounts derived from contributions paid or accrued after December 31, 1985, in taxable years ending after said date, which are attributable to post-retirement medical benefits allocated to the separate account of such Participant, under a welfare benefit fund, as defined in Section 419(e) of the Code, maintained by the Employer.

The term "Compensation," for purposes of this Section 12.9, shall mean "Compensation" within the meaning of Section 1.6(B) of the Plan.

ARTICLE 13 AMENDMENT, TERMINATION, BENEFIT LIMITATIONS, MERGER AND APPROVAL OF PLAN

13.1 AMENDMENT OF PLAN.

Only upon the previously written direction from the Town, or to the extent required by any applicable statute, may the Pension Committee revoke, modify, alter or amend the Plan, provided, however, that the Pension Committee shall not be required to adopt any amendment which would conflict with any applicable state or federal legislation or regulation. No amendment shall increase the duties or responsibilities of the Pension Committee without their consent thereto in writing. No amendment shall have the effect of revesting the Employer, in whole or in part, in any of the assets of the Plan or of diverting any part of the assets to purposes other than for the exclusive benefit of the Participants, Joint Annuitants, surviving spouses, and Designated Beneficiaries at any time prior to the satisfaction of all the liabilities under the Plan with respect to such persons. No amendment shall reduce the amount of, or restrict the payment of, vested benefits accrued hereunder prior to the effective date of such amendment, and for this purpose only, vested benefits accrued as of such date shall be deemed to be those payable from the Pension Fund as then constituted in the event each Participant terminated employment on the day immediately preceding the effective date of the amendment.

13.2 TERMINATION OR PARTIAL TERMINATION OF PLAN.

The Plan is voluntary on the part of the Employer. The Employer reserves the right to terminate the Plan, in whole or in part, or the Pension Fund, or both, and to suspend, reduce or discontinue contributions at any time. Upon termination of the Plan, or upon the complete discontinuance of contributions, the accrued benefits of Participants to the date of such termination or discontinuance shall be nonforfeitable to the extent then funded.

13.3 ALLOCATION OF ASSETS UPON PLAN TERMINATION.

- (A) In the event of the termination of the Plan, the Pension Committee, after reserving an amount sufficient to pay all expenses of the Plan and Agreement and Declaration of Trust, shall allocate all assets of the Plan or their proceeds in order of preference as hereinafter set forth (but only to the extent that an individual's pension benefit is not fully funded):
- (1) The portion derived from a Participant's own contributions, with Interest (if any).
- (2) In the case of the pension benefit of a Participant, Joint Annuitant, surviving spouse or Designated Beneficiary, the benefit which was in pay status as of the date of such termination.
- (3) To provide benefits to Participants who were eligible to retire in accordance with Article 3 as of the date of such termination.
 - (4) To all other vested pension benefits (if any) under the Plan.

- (5) To all other nonforfeitable pension benefits under the Plan.
- (B) If the assets in the Pension Fund applicable to any of the categories listed in Section 13.3(A) are insufficient to provide for all persons listed in such categories, then the assets shall be allocated among those persons in the last category to which the assets are available in the same proportion which the present value, as determined by the Actuary, of each person's benefit bears to the present value of all benefits attributable to that category.

13.4 INVALID PROVISIONS.

If any provision of the Plan is held invalid or unenforceable, such invalidity or unenforceability shall not affect any other provisions hereof, and the Plan shall be construed and enforced as if such invalid or unenforceable provisions had not been included.

13.5 NO GUARANTY OF EMPLOYMENT; EFFECT OF DENIAL OF BENEFITS.

The Employer, in establishing and maintaining this Plan, assumes no responsibility or liability for continued employment of the Participants hereunder. In the event of any denial of benefits hereunder, the Employer shall be held blameless.

13.6 USAGE.

Wherever any words are used herein in the masculine gender they shall be construed as though they were also in the feminine gender, in all cases where they would so apply. Wherever any words are used herein in the singular form, they shall be construed as though they were used in the plural form, and vice versa, in all cases where they would so apply.

Dated this day of	, 2019.
	THE TOWN OF WESTPORT, CONNECTICUT
	By
	Jim Marpe
	First Selectman
	That belocation
UNION AND NON-SUPE	a true and attested copy of the RETIREMENT PLAN FOR NON-RVISORY EMPLOYEES OF THE TOWN OF WESTPORT, d and restated effective as of July 1, 2019.
Attes	
	Patricia Strauss
	Town Clerk